UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): December 30, 2024

VERU INC.

(Exact name of registrant as specified in its charter)

Wisconsin (State or other jurisdiction of incorporation)

financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

1-13602 (Commission File Number) 39-1144397 (IRS Employer Identification No.)

Emerging growth company □

2916 N. Miami Avenue, Suite 1000, Miami, Florida 33127 Address of principal executive offices) (Zip Code)

Registrant's telephone number, including area code: (305) 509-6897

Not Applicable

(Former name or former address, if changed since last report.)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see

Ger	General Instruction A.2. below):			
	☐ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)			
	□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)			
	□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))			
	□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))			
Securities registered pursuant to Section 12(b) of the Act:				
	Title of each class	Trading Symbol(s)	Name of each exchange on which registered	
Co	mmon Stock, \$0.01 par value per share	VERU	NASDAQ Capital Market	
	Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).			

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised

Item 1.01 Entry into a Material Definitive Agreement

On December 30, 2024, Veru Inc. (the "Company") and The Female Health Company Limited, a wholly owned subsidiary of the Company ("FHC Limited" and, collectively with the Company, the "Sellers") entered into a Stock and Asset Purchase Agreement (the "Purchase Agreement") with Clear Future, Inc. (the "Purchaser"). Pursuant to, and subject to the terms and conditions of, the Purchase Agreement, the Purchased substantially all of the assets (the "FC2 Business Sale") related to the Company's FC2 female condom business ® (internal condom), including the stock of the Company's U.K. and Malaysian operating subsidiaries. The Purchaser will only assume certain liabilities relating to the FC2 business that are specified in the Purchase Agreement. The transaction closed on December 30, 2024.

The purchase price for the FC2 Business Sale is \$18.0 million in cash, subject to adjustment as set forth in the Purchase Agreement. The adjustments to the purchase price in the Purchase Agreement include a customary working capital adjustment based on the amount by which certain working capital items at closing are greater or less than a target set forth in the Purchase Agreement. Estimated proceeds to the Company after deducting a change of control payment due to SWK Funding LLC ("SWK") pursuant to the Residual Royalty Agreement, dated as of March 5, 2018 (the "Royalty Agreement"), between the Company and SWK, together with other customary transaction fees for a transaction of this type, is approximately \$12.5 million, subject to adjustment as set forth in the Purchase Agreement.

The Company and the Purchaser made customary representations and warranties, and agreed to certain customary covenants, in the Purchase Agreement. Subject to certain exceptions and limitations, each party has agreed to indemnify the other for breaches of representations, warranties and covenants and for certain other matters.

The foregoing summary is qualified in its entirety by reference to the Purchase Agreement, which is attached as Exhibit 10.1 to this report and incorporated herein by reference in its entirety. The Purchase Agreement has been included to provide investors with information regarding its terms. It is not intended to provide any other factual information about the Company or any of its subsidiaries. The representations, warranties and covenants contained in the Purchase Agreement were made only for purposes of such agreement and as of specific dates, were solely for the benefit of the parties to such agreement and may be subject to limitations agreed upon by the contracting parties, including being qualified by confidential disclosures exchanged between the parties in connection with the execution of the Purchase Agreement. The representations, warranties and covenants may have been made for the purposes of allocating risk between the Sellers and the Purchaser instead of establishing these matters as facts and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors. Investors are not third-party beneficiaries under the Purchase Agreement and should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts or condition of the Company or any of its subsidiaries or affiliates. Moreover, information concerning the subject matter of the representations and warranties may change after the date of the Purchase Agreement, which subsequent information may or may not be fully reflected in the Company's public disclosures.

Item 1.02 Termination of a Material Definitive Agreement

In connection with the closing of the FC2 Business Sale, on December 30, 2024, the Company made a change of control payment of \$4.2 million to SWK pursuant to the Royalty Agreement, and upon such payment the Royalty Agreement terminated in accordance with its terms.

Item 2.01 Completion of Acquisition or Disposition of Assets

The information contained in Item 1.01 above is incorporated herein by reference.

Item 2.04 Triggering Events That Accelerate or Increase a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement

The information contained in Item 1.02 above is incorporated herein by reference.

Item 8.01 Other Events

On December 31, 2024, the Company issued a press release announcing the closing of the Transaction. A copy of the press release is attached as Exhibit 99.1 to this report and is incorporated by reference herein.

Item 9.01 Financial Statements and Exhibits.

(b) Pro Forma Financial Information.

The following unaudited pro forma consolidated financial statements reflecting the transaction described under Item 2.01 above are filed with this report as Exhibit 99.2 and are incorporated herein by reference:

- the Company's unaudited pro forma consolidated balance sheet as of September 30, 2024; and
- the Company's unaudited pro forma consolidated statement of operations for the years ended September 30, 2024 and 2023.

The unaudited pro forma consolidated financial statements are not intended to represent or be indicative of the Company's consolidated results of operations or financial position that would have been reported had the disposition been completed as of the dates presented, and should not be taken as representation of the Company's future consolidated results of operations or financial condition. The pro forma adjustments are based on available information and certain assumptions that management believes are reasonable under the circumstances.

(d) Exhibits.

Exhibit No.	Document
10.1	Stock and Asset Purchase Agreement, dated as of December 30, 2024, among Veru Inc., The Female Health Company Limited and Clear Future, Inc.
99.1	Press Release issued December 31, 2024.
99.2	Unaudited Pro Forma Financial Information and accompanying notes.
104	Cover Page Interactive Data File (embedded within the Inline XBRL document).
3	

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: January 3, 2025 VERU INC.

By: /s/ Michele Greco Michele Greco Chief Financial Officer and Chief Administrative Officer

STOCK AND ASSET PURCHASE AGREEMENT

 $by\ and\ among$

VERU INC.,

THE FEMALE HEALTH COMPANY LIMITED

and

CLEAR FUTURE, INC.

Dated effective as of December 30, 2024

TABLE OF CONTENTS

			<u>Page</u>
1.	Define	ed Terms; Construction	1
	1.01.	Defined Terms	1
	1.02.	Construction	16
2.	Purcha	se and Sale; Closing	17
	2.01.	Purchase and Sale of Stock	17
	2.02.		17
	2.03.		21
	2.04.		22
	2.05.	Closing	23
	2.06.	Certain Payments and Deliverables at Closing	23
	2.07.	Closing Estimates	25
	2.08.	Purchase Price Adjustment	25
	2.09.	Purchase Price Allocation	27
	2.10.	Non-Assignable Assets	27
	2.11.	Withholding Taxes.	28
_	D.	1W 4 60 W	20
3.		sentations and Warranties of Sellers	29
	3.01.	Organization (A. W. W. L.)	29
	3.02.	Authorization of Agreement; No Violation	29
	3.03.	Capitalization of Business Subsidiaries	30
	3.04. 3.05.	Consents and Approvals Financial Statements	31
			31
	3.06. 3.07.	Absence of Changes	32
	3.07.	Proceedings; Orders Taxes	33 34
	3.08.	Employee Benefit Plans	34
	3.10.	Real Property	37
	3.10.	Transferred Tangible Assets	38
	3.11.	Labor and Employment Matters	38
	3.12.	Material Contracts; Assumed Business Contracts	39
	3.14.	Intellectual Property	41
	3.15.		44
	3.16.	Brokers	44
	3.17.	Insurance	44
	3.18.	Legal Compliance	45
	3.19.	Product Matters	49
	3.20.	Title and Sufficiency of Assets	49
	3.21.	Certain Transactions	50
	3.22.	Customers and Suppliers	50
	3.23.	Compensation of and Contracts with Employees and Independent Contractors.	51

	3.24.	No Additional Representations	52
4.	Repres	sentations and Warranties of Buyer	52
	4.01.	Organization	52
	4.02.	Authorization of Agreement; No Conflict	52
	4.03.	Litigation	53
	4.04.	Brokerage	53
	4.05.	Financial Capability; Solvency	53
	4.06.	Investment Intent	53
	4.07.	No Additional Representations	53
5.	Certair	n Agreements	54
٥.	5.01.	Restrictive Covenants	54
	5.02.	Publicity	56
	5.03.	Conflicts and Privilege	56
	5.04.	Employee Benefit Arrangements and Other Employee Matters	57
	5.05.	Post-Closing Access to Record and Personnel	58
	5.06.	Expenses	58
	5.07.	Wrong Pockets Clause	58
	5.08.	Further Assurances	59
	5.09.	Tax Matters	59
	5.10.	Name Change	63
	5.11.	Use of Inventories of Products	63
	5.12.	Sufficient Capitalization of Business Subsidiaries	63
	5.12.	Covenant Not To Sue	63
	3.13.	Covenant Not 10 Suc	0.
6.		nification	64
	6.01.	Indemnification by Veru	64
	6.02.	Indemnification by Buyer	65
	6.03.	Indemnification Procedure	65
	6.04.	Limitations on the Indemnification Obligations of Sellers	67
	6.05.	Multiple Representations and Warranties	70
	6.06.	Tax Treatment	71
	6.07.	Exclusive Remedy	71
	6.08.	Retention Escrow Amount	71
7.	Genera	al Provisions	71
	7.01.	Notices	71
	7.02.	Entire Agreement	72
	7.03.	Incorporation of Terms	72
	7.04.	No Other Representations, Warranties, Covenants and Agreements	72
	7.05.	Waiver	73
	7.06.	Amendment	73
	7.07.	Assignment; Binding Effect	73
	7.08.	Benefit	73
	7.09.	Disclosure Schedules	73
	7.10.	No Agreement Until Executed	74

7.11.	Governing Law	74
7.12.	Specific Enforcement	74
7.13.	Negotiation of Agreement	74
7.14.	Facsimile Signature; Counterparts	74
7.15.	Severability	75
7.16.	No Recourse	75

STOCK AND ASSET PURCHASE AGREEMENT

This STOCK AND ASSET PURCHASE AGREEMENT (this "<u>Agreement</u>") is dated as of December 30, 2024 among VERU INC., a Wisconsin corporation ("<u>Veru</u>"), THE FEMALE HEALTH COMPANY LIMITED, a company organized under the laws of England and Wales ("<u>Limited</u>" and together with Veru, "<u>Sellers</u>" and each, a "<u>Seller</u>"), and Clear Future, Inc., a Delaware corporation ("<u>Buyer</u>"). Certain capitalized terms used herein but not otherwise defined shall have the meanings set forth in <u>Section 1</u>.

RECITALS

- A. Sellers and the Business Subsidiaries are engaged in, among other matters, the Business.
- B. Sellers are the owners of all of the issued shares (the <u>'Stock</u>") in the capital of The Female Health Company (UK) Plc., a company organized under the laws of England and Wales (<u>'FHC UK</u>"), which is the owner of all of the issued and paid-up share capital or other equity interests (the <u>'Malaysia Stock</u>") of The Female Health Company (M) SDN.BHD, a company incorporated under the laws of Malaysia (<u>'FHC Malaysia</u>").
 - C. Veru owns certain assets used in the Business.
 - D. Sellers desire to sell to Buyer, and Buyer desires to purchase from Sellers, the Stock on the terms and conditions set forth herein.
- E. Veru desires to sell to Buyer, and Buyer desires to purchase from Veru, certain of the assets, rights and claims of Veru which relate to the Business, on the terms and conditions set forth herein.

AGREEMENTS

In consideration of the recitals and the mutual agreements and covenants herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- 1. <u>Defined Terms; Construction</u>.
 - 1.01. <u>Defined Terms</u>. Certain terms used in this Agreement are defined below:
- "Acquired Business Intellectual Property" means the Transferred Business Intellectual Property and all Intellectual Property Rights owned or purportedly owned by any of the Business Subsidiaries.
- "Acquired Business Technology" means the Transferred Business Technology, and all Technology owned or purportedly owned by any of the Business Subsidiaries.

- "Accounting Principles" shall mean the same accounting principles, practices, procedures, policies and methods, classifications, judgments, inclusions, exclusions and valuation and estimation methodologies that were employed in the preparation of the Target Working Capital.
- "Adjustment Escrow Account" means the account established by the Escrow Agent pursuant to the terms of the Escrow Agreement for purposes of holding the Adjustment Escrow Amount.
 - "Adjustment Escrow Amount" means an amount equal to \$300,000.
- "Affiliate" means, with respect to any Person, any other Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such first Person. For purposes of this definition, "control" (including the correlative terms "controlling," "controlled by" and "under common control with"), with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities or by contract or otherwise.
- "Anti-Corruption and Anti-Bribery Laws" shall mean the Foreign Corrupt Practices Act of 1977, any rules or regulations thereunder, the International Travel Act of 1961, and, as applicable United States or foreign anti-corruption or anti-bribery Laws or regulations, including the United Kingdom Bribery Act of 2010 and laws passed pursuant to the OECD Convention on Combating Bribery of Foreign Public Officials in International Business Transactions.
- "Anti-Money Laundering Laws" shall mean the USA PATRIOT Act of 2001 and the Bank Secrecy Act and all regulations thereunder and any other applicable anti-money laundering Law or any similar Legal Requirement of any other jurisdiction where any Business Entity operates with respect to the Business.
- "Assigned Other IPR" means all Intellectual Property Rights owned or purportedly owned by Veru (other than Patents and Marks) that are (a) Related To the Business, including the items listed or described in Schedule 1-1(a), or (b) in, to or embodied by the Transferred Tangible Assets.
- "Assigned Trademarks" means each of the Marks set forth in Schedule 1-2, in each case whether pending, issued, expired, abandoned or closed, and all foreign counterparts of any such Mark.
- "Business" means the business of the manufacture, marketing, sale and distribution of the Product and/or all related services relating to the marketing, sale and distribution of the Product (including in connection with the Portal), in each case as conducted during the twelve (12)-month period immediately preceding the Closing Date.
- "Business Day" means any day, other than a Saturday or Sunday, on which the principal commercial banks in New York City are open for commercial banking business during normal banking hours.

"Business Employees" means (i) all individuals who are employed by any of the Business Subsidiaries immediately prior to the Closing Date and (ii) the other individuals who are employed by a Seller immediately prior to the Closing Date and identified on Schedule 5.04 hereto.

"Business Entities" means the Business Subsidiaries and Sellers.

"Business Material Adverse Effect" means any event, circumstance, development, change, effect or condition that, individually or in the aggregate, (a) has had a material adverse effect on the business, condition (financial or otherwise), assets, Liabilities or results of operations of the Business, taken as a whole, or (b) has materially and adversely impaired or delayed the ability of the Sellers (or the Business Subsidiaries) to perform their obligations hereunder or consummate the transactions contemplated hereby; provided, however, that any event, circumstance, development, change, effect or condition arising from or relating to the following will not be taken into account in determining whether there has been or will be a Business Material Adverse Effect: (i) general economic conditions affecting the industry in which the Business participates, or the United States economy as a whole; (ii) changes in U.S. generally accepted accounting principles after the date hereof; (iii) changes in Law after the date hereof; (iv) national or international political conditions, including armed hostilities or other international or national calamity or act of terrorism involving the United States of America; (v) general changes in financial, banking, or securities markets; (vi) wildfires, floods, epidemics, pandemics, earthquakes, hurricanes, tornados or other natural disasters; (vii) any failure to meet any financial projections, forecasts or estimates of revenue, earnings, cash flow or cash position (it being understood that any underlying cause(s) of any such failure may be taken into consideration when determining whether a Business Material Adverse Effect has occurred); (viii) compliance with the terms of, or taking any action required by, this Agreement; provided further, that, in the case of each of clauses (i), (ii), (iv), (v), and (vi), such change, event, occurrence, development or state of circumstances or facts does not have a materially disproportionate effect on the Business, taken as a whole, relative to ot

"Business Subsidiaries" means FHC UK and FHC Malaysia.

"Cash and Cash Equivalents" means all cash (including deposit accounts, money market accounts, checking accounts, bank accounts, certificates of deposit, time deposits, mutual funds and other investment securities) and cash equivalents of the Business Subsidiaries that are immediately convertible into cash, as determined in accordance with GAAP.

"Change of Control" means, with respect to a Person, any transaction or series of related transactions that constitute: (a) the sale of all or substantially all of an entity's business or assets; (b) any merger, consolidation, share exchange, recapitalization, business combination or other transaction resulting in the exchange of the outstanding shares of an entity for securities or consideration issued, or caused to be issued, by the acquiring corporation or its subsidiary, unless the stockholders of such entity as of the date prior to the closing date of such transaction (or series of related transactions) hold more than fifty percent (50%) of the voting securities in the surviving corporation in such transaction computed on a fully diluted basis; or (c) any person having acquired beneficial ownership or the right to acquire beneficial ownership of, or any "group" (as such term is defined under Section 13(d) of the Securities Exchange Act of 1934, as amended) having been formed that beneficially owns or has the right to acquire beneficial ownership of, fifty percent (50%) or more of the outstanding voting securities of an entity.

"Closing Working Capital Deficit" shall mean the amount, if any, by which the Target Working Capital exceeds the Closing Working Capital, but only to the extent the difference exceeds the Working Capital Collar, and only such amount by which the difference between the Target Working Capital and the Closing Working Capital exceeds the Working Capital Collar; in the event such difference does not exceed the Working Capital Collar, the Closing Working Capital Deficit shall be \$0.

"Closing Working Capital Excess" shall mean the amount, if any, by which the Closing Working Capital exceeds the Target Working Capital, but only to the extent the difference exceeds the Working Capital Collar, and only such amount by which the difference between the Closing Working Capital and the Target Working Capital exceeds the Working Capital Collar; in the event such difference does not exceed the Working Capital Collar, the Closing Working Capital Excess shall be \$0.

"Code" means the Internal Revenue Code of 1986, as amended.

"Competitive Activity" means the development, manufacture, marketing, distribution or sale of any female condoms anywhere in the world.

"Confidential Information" means any confidential or proprietary information, including any formula, pattern, device, compilation or information, proprietary technical, economic, environmental, operational, financial, technology, operating, financial and/or other business information, methods of operation, financial statements, trade secrets, market studies and forecasts, competitive analyses, target markets, advertising techniques, pricing policies and information, specifications for products, equipment and processes, manufacturing and performance specifications and procedures, engineering drawings and graphs, technical, research and engineering data, manufacturing know-how, the substance of agreements with customers and others, marketing and similar arrangements, servicing and training programs and arrangements, customer profiles, customer preferences, other trade secrets and any other documents or materials embodying such information, employee census information and, as required by applicable Law, other employee information.

"Contract" means contracts, agreements, purchase orders, leases, licenses and instruments or other binding arrangement, whether oral or written.

"Data Protection Laws" means the data protection and privacy laws of each country where any Business Entity is established and those of each country where any Personal Information is collected, transmitted, secured, stored, shared or otherwise processed by or on behalf of such Business Entity and that are applicable to such Business Entity, including, as applicable, the Health Insurance Portability and Accountability Act of 1996 (42 U.S.C. § 1320d et seq.), as amended by the Health Information Technology for Economic and Clinical Health Act 42 U.S.C. §§ 17921 et seq. the General Data Protection Regulation 2016/680 of the European Parliament, and shall also mean the Massachusetts data protection law, 201 CMR 17.00, the California Consumer Privacy Act, and Section 5 of the (U.S.) Federal Trade Commission Act and any and all laws and regulations governing privacy, cybercrime, data protection or privacy, or unfair or deceptive trade practices with respect to privacy, data protection, or information security.

"Effective Time" means 11:59 p.m. New York time on the Closing Date.

"Encumbrance" means, with respect to any property or asset, any encumbrance, lien, mortgage, pledge, hypothecation, encroachment, easement, license, lease, deed restriction, zoning, unpaid Taxes, defect or burden on title (including, without limitation, all matters of record), use restriction, right-of-way, title defect, charge, attachment, levy, judgment, option or other right to acquire an interest, right of first refusal or security interest thereupon or in respect thereof.

"Environmental Laws" means all Laws applicable to the Business as in effect as of the Closing Date relating to air quality, soil quality, water quality, sub-slab and indoor contaminant air vapors, wetlands, natural resources, solid waste, hazardous waste, hazardous or toxic substances, pollution, public health, human health or the environment, including, but not limited to, the Comprehensive Environmental Response, Compensation, and Liability Act, 42 U.S.C. § 9601 et seq., the Hazardous Materials Transportation Act (49 U.S.C. § 1801), the Federal Water Pollution Control Act (33 U.S.C. § 1251, et seq.), the Safe Drinking Water Act (42 U.S.C. § 300f, et seq.), the Resource Conservation and Recovery Act (42 U.S.C. § 6901, et seq.), the Clean Air Act (42 U.S.C. § 7401, et seq.), the Toxic Substances Control Act (15 U.S.C. § 2601, et seq.) and the Federal Insecticide, Fungicide, and Rodenticide Act (7 U.S.C. § 136, et seq.).

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations promulgated thereunder.

"ERISA Affiliate" means any entity which is a member of a "controlled group of corporations" with, under "common control" with, or otherwise required to be aggregated with, any Seller, as set forth in Sections 414(b), (c), (m) and (o) of the Code.

"Escrow Agent" means Wilmington Trust, National Association, a national association.

"Escrow Agreement" means an escrow agreement, substantially in the form of Exhibit A attached hereto, to be entered into at the Closing by Veru, Buyer and the Escrow Agent.

"Estimated Closing Statement" that certain statement prepared by Seller and delivered to Buyer at least three (3) Days prior to the Closing Date, which such statement is attached hereto as Exhibit B and sets forth in reasonable detail Sellers' good faith calculations of the Estimated Closing Consideration, including (a) the estimated Net Working Capital immediately prior to the Closing (the "Estimated Working Capital"), (b) the estimated aggregate amount of Indebtedness of the Business Subsidiaries unpaid as of immediately prior to the Closing (the "Estimated Indebtedness Amount"), (c) the estimated aggregate amount of Cash and Cash Equivalents of the Business Subsidiaries as of immediately prior to the Closing (the "Estimated Cash"), provided that the amount of Estimated Cash shall not exceed \$1,274,056, and (d) the estimated amount of Pre-Closing Taxes (the "Estimated Pre-Closing Taxes Amount").

"Estimated Working Capital Deficit" shall mean the amount, if any, by which the Target Working Capital exceeds the Estimated Working Capital, but only to the extent the difference exceeds the Working Capital Collar, and only such amount by which the difference between the Target Working Capital and the Estimated Working Capital exceeds the Working Capital Collar; in the event such difference does not exceed the Working Capital Collar, the Estimated Working Capital Deficit shall be \$0.

"Estimated Working Capital Excess" shall mean the amount, if any, by which the Estimated Working Capital exceeds the Target Working Capital, but only to the extent the difference exceeds the Working Capital Collar, and only such amount by which the difference between the Estimated Working Capital and the Target Working Capital exceeds the Working Capital Collar; in the event such difference does not exceed the Working Capital Collar, the Estimated Working Capital Excess shall be \$0.

"Excluded Scheduled Receivables" means the accounts receivable set forth on Schedule 1-3.

"Export and Import Approvals" shall mean all export licenses, license exceptions, consents, notices, waivers, approvals, orders, authorizations, registrations, declarations and filings, from or with any Governmental Authority, that are required for compliance with Export and Import Control Laws.

"Export and Import Control Laws" shall mean any U.S. or applicable non-U.S. Law, regulation, or order governing (a) imports, exports, reexports, or transfers of products, services, software, or technologies from or to the United States or another country; (b) any release of technology or software in any foreign country or to any foreign person (anyone other than a citizen or lawful permanent resident of the United States, or a protected individual as defined by 8 U.S.C. § 1324b(a)(3)) located in the United States or abroad; (c) economic sanctions or embargoes; or (d) compliance with unsanctioned foreign boycotts.

"FDA" means the United States Food and Drug Administration and any successor agency thereto.

"Fraud" means Delaware common law fraud committed with the intent to defraud, provided that Fraud does not include constructive fraud or other claims based on constructive knowledge, negligent misrepresentation, recklessness or similar theories.

"GAAP" means generally accepted accounting principles in the United States of America, as in effect on the date hereof or, with respect to any financial statements, the date such financial statements were prepared.

"Governmental Authority" means any foreign or United States federal, provincial, state, municipal or local government agency, division, or subdivision thereof or any regulatory body, agency, authority or commission entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, or any court or tribunal (or any department, bureau or division thereof).

"Hazardous Materials" means all wastes, substances or materials defined as "hazardous substances," "toxic substances" or "hazardous wastes," or any other term of similar import under, or otherwise regulated pursuant to, any Environmental Law, including petroleum (including crude oil or any fraction thereof) and polychlorinated biphenyls.

"Health Care Laws" means (a) the Food, Drug, and Cosmetic Act, as amended, and all rules and regulations promulgated thereunder, and similar laws in foreign jurisdictions related to any product or service that is commercialized or marketed, (b) all applicable Laws administered by the FDA and other regulatory authorities in foreign jurisdictions, United States federal, state, and local regulatory authorities governing or relating to good laboratory practices, good clinical practices, good manufacturing practices, recordkeeping, procurement, design, research, studying, manufacturing, quality, registration, listing, certification, licensing, production, handling, sterilization, advertising, importing, exporting, testing, development, approval, processing, reporting, packaging, labeling, storage, marketing, sale, distribution and use of medical devices or components thereof or other medical products manufactured by or on behalf of or health care services provided by or on behalf of the Business or any Business Entity, including, without limitation, 21 CFR 820; (c) the federal Anti-Kickback Statute (42 U.S.C. § 1320a-7b(b)), the Physician Payments Sunshine Act (42 U.S.C. § 1320a-7h), the civil False Claims Act (31 U.S.C. §§ 3729 et seq.), the criminal False Claims Law (42 U.S.C. § 1320a-7b(a)), the exclusion laws (42 U.S.C. § 1320a-7), the civil monetary penalties law (42 U.S.C. § 1320a-7a), the Program Fraud Civil Remedies Act, 31 U.S.C. §§ 3801-3812, EU Member States anti-gift and transparency laws; the EU Medical Devices Directive 93/42/EEC, the EU Medical Devices Regulation 2017/745; all similar or comparable foreign, state, local and international laws, in each case, as amended and the regulations promulgated thereunder; or any legal requirements promulgated by other regulatory authority with jurisdiction over the Business or any Business Entity and each of its products or services; (d) the Health Insurance Portability and Accountability Act of 1996, as amended, and foreign, federal, state, and local privacy and security Laws; (e) Federal Trade Commission Act and implementing regulations and rules; (f) Laws promulgated and enforced by the Consumer Product Safety Commission and similar regulator authorities; and (g) any and all other applicable foreign, federal, state, and local Laws, rules and regulations, ordinances, judgments, decrees, orders, writs, and injunctions applicable to the Business, any health care or medical product or service marketed or commercialized by or for any Business Entity, or any business operations of an Business Entity, each of (a) through (g) as may be amended from time to time.

"Income Tax Returns" means any Tax Return with respect to income Taxes.

"Indebtedness" means, with respect to any Person, (a) all obligations for borrowed money, whether current or funded, secured or unsecured or indebtedness issued in substitution or exchange for borrowed money, (b) all obligations for the deferred purchase price of any property or services, including earn outs, payments under non-compete agreements and seller notes (other than trade accounts payable reflected on the balance sheet of such Person as current liabilities arising in the ordinary course of business consistent with past practice of such Person from the purchase of inventory and supplies), (c) all obligations created or arising under any conditional sale or other title retention agreement regarding property acquired by such Person, (d) all obligations secured by a purchase money mortgage or other lien to secure all or part of the purchase price of property subject to such mortgage or lien, (e) all obligations under leases required to be, in accordance with GAAP, recorded as capital leases under which such Person is liable as lessee, (f) any obligation regarding bankers' acceptances or letters of credit, whether drawn or not, (g) any obligations secured by liens on property acquired by such Person, whether or not such obligations were assumed by such Person at the time of acquisition of such property, (h) all amounts payable by a Business Subsidiary to a Seller, (i) all accrued but unpaid severance obligations payable to former employees relating to periods prior to the Closing or any unpaid bonuses that were payable or accrued, including the bonuses set forth on Schedule 1-6, (j) all obligations of a type referred to in clauses (a) through (h) above which are directly or indirectly guaranteed by such Person or which it has agreed (contingently or otherwise) to purchase or otherwise acquire or in respect of which it has otherwise assured a credit against loss, (k) interest, principal, prepayment penalty, taxes, fees, or expenses, to the extent due or owing in respect of those items in clauses (a) through

"Indemnified Taxes" means (a) all Taxes of Sellers; (b) all Taxes imposed on the Purchased Assets or that otherwise arise with respect to the Business, in each case, for Pre-Closing Taxable Periods; (c) all Taxes of the Business Subsidiaries for Pre-Closing Taxable Periods (including any Taxes that are not yet due and payable), determined (i) as if the Business Subsidiaries used the accrual method of Tax accounting throughout all Pre-Closing Tax Periods, (ii) by treating any advance payments, deferred revenues, or other prepaid amounts received or arising in any Pre-Closing Taxable Period as subject to Tax in such period regardless of when actually recognized for income tax purposes, (iii) by including, for any Pre-Closing Taxable Period, any adjustment in Taxable income or similar amounts arising as a result of transactions, events, or accounting methods employed prior to the Closing pursuant to Section 481(a) of the Code (or any similar provision of U.S. state, local, or other non-U.S. Tax Laws), (iv) by including in the Pre-Closing Tax Period any Taxes attributable to any amount required to be included in the income of Buyer or any of its owners or affiliates under Section 951 or Section 951A of the Code (or, in each case, any similar provision of state, local or other Tax Law) with respect to the Business Subsidiaries, to the extent such amount would be allocable to the Pre-Closing Tax Period if the Taxable year of each of Buyer, the Business Subsidiaries, and each other Person ended on the Closing Date and Buyer's holding period with respect to the Business Subsidiaries included the Closing Date, (d) all Taxes of any other Person for which liability arises as a result of the Business Subsidiaries being or having been a member of an affiliated, consolidated, combined, unitary or similar group (including pursuant to Treasury Regulations Section 1.1502-6 or any similar provision of applicable Law) on or prior to the Closing Date, (e) all Taxes for which the Business Subsidiaries are liable by Contract, as a transferee or successor, by operation of Law or otherwise, which relate to a transaction or event occurring or action taken on or prior to Closing Date, (f) all Taxes arising from the transactions contemplated by this Agreement (limited in the case of Transfer Taxes to Taxes for which Seller is liable pursuant to Section 5.06), including all Taxes required to be withheld or deducted under the Code or any other Tax Law in connection with the transactions contemplated by this Agreement, and (g) any amounts payable after the Closing Date as a result of an election under Section 965(h) of the Code made prior to the Closing Date by or with respect to the Business Subsidiaries, in each case, that are not reflected in the calculation of the Closing Working Capital, the Closing Indebtedness Amount or the Final Pre-Closing Taxes Amount.

"Intellectual Property" means Intellectual Property Rights and Technology.

"Intellectual Property Rights" means all of the following in any jurisdiction throughout the world and all rights therein: (i) patents and applications therefor (including patents issuing on such applications), together with all continuations, continuations-in-part, reissues, renewals, reexaminations, provisionals, divisionals, substitutions, extensions or revisions thereof, any foreign counterparts or equivalents of any of the foregoing and any other patents, applications or extensions that claim priority to or through any of the foregoing ("Patents"); (ii) trade-secret rights and all other rights in confidential business or technical information; (iii) copyrights, copyrights registrations and applications therefor, moral rights, and all other rights corresponding thereto (including mask works); (iv) domain names, uniform resource locators, other names and locators associated with the Internet, and applications therefor; (v) trade names, logos, common law trademarks and service marks and trademark and service mark registrations, and related goodwill and applications therefor ("Marks"); (vi) all rights in databases and data collections; and (vii) any similar or equivalent rights to any of the foregoing, including but not limited to moral rights (as applicable).

"IP Contract" means any Contract (a) to which any Business Subsidiary is a party or is bound or (b) to which either Seller is a party or is bound and that is Related To the Business, the Products or the Portal, in each case by or through which (i) any Intellectual Property is created or developed, or (ii) other Persons grant a Business Entity or a Business Entity grants any other Persons any exclusive or non-exclusive rights or interests in or to any Intellectual Property.

"IRS" means the United States Internal Revenue Service.

"Knowledge," "known to Sellers," "Sellers' Knowledge," of which Sellers "knows," or the like, means the actual knowledge of Harry Fisch, Mitchell Steiner, Kevin Gilbert, Michele Greco, Michael Purvis, Phil Greenberg and Martin Tayler and the knowledge such aforementioned individuals would have after a reasonable inquiry of his or her direct reports.

"Legal Requirement" or "Law" means any foreign government or any United States federal, provincial, state, municipal or local (a) law, common law, statute, directive, ordinance, code, rule or regulation, (b) judicial, administrative or governmental order determination, writ, injunction, restrictions, judgments or decrees, or (c) any similar provision having the force or effect of law, in each case of clauses (a), (b) or (c), as promulgated or applied by a Governmental Authority.

"Liability" means any liability, debt, obligation, deficiency, interest, Tax, penalty, fine, claim, demand, judgment, cause of action or other damages (including loss of benefit or relief), cost or expenses or commitment of any nature whatsoever (whether direct or indirect, known or unknown, accrued or unaccrued, absolute or contingent, or matured or unmatured, liquidated or unliquidated, whether due or to become due, and regardless of when asserted), including any of the foregoing arising under any Legal Requirement, Permit, Proceeding, Order or Contract.

"Loss" means any claim, obligation, loss, damage, Liability, demand, action, cause of action, assessment, judgment, deficiency, cost, penalty, fine, Tax, or other cost or expense, (including reasonable attorney's and accountant's fees, costs, and expenses incurred in investigating and defending against any Claim).

"McKesson Dispute" means that certain dispute as summarized on Schedule 3.05(a).

"Net Working Capital" means the current assets of the Business Subsidiaries and the current assets included within the Purchased Assets (consisting solely of the types of current assets on the balance sheet line item accounts specified on Schedule 1-4, which shall be agreed upon by the parties) less the current liabilities of the Business Subsidiaries and the current liabilities included within the Assumed Liabilities (consisting solely of the types of current liabilities on the balance sheet line item accounts specified on Schedule 1-4, which shall be agreed upon by the parties). Net Working Capital shall not include (i) Cash and Cash Equivalents, (ii) Tax refunds or credits, (iii) deferred income Taxes, (iv) any Liabilities relating to any severance obligations, (v) any Liabilities for any income Taxes, or (vi) any Liabilities relating to any Indebtedness, in each case except to the extent such amounts are otherwise expressly specified on Schedule 1-4.

"Open Source Materials" means all software or other material that is distributed as "free software," "open source software" or under a similar licensing or distribution terms (including the GNU General Public License (GPL), GNU Lesser General Public License (LGPL), Mozilla Public License (MPL), BSD licenses, the Artistic License, the Netscape Public License, the Sun Community Source License (SCSL) the Sun Industry Standards License (SISL) and the Apache License) ("Open Source Materials").

"Order" means any decree, ruling, order, judgment, writ, award, injunction, stipulation or consent of or by, or settlement agreement with, a Governmental Authority.

"Organizational Documents" means (a) articles or certificate of incorporation and the bylaws of a corporation, (b) the partnership agreement and any statement of partnership of a general partnership, (c) the limited partnership agreement and the certificate or articles of limited partnership of a limited partnership of a limited partnership, (e) the operating agreement or limited liability company agreement of a limited liability company and the articles of organization or certificate of formation of a limited liability company, (f) any charter or similar document adopted or filed in connection with the creation, formation or organization of a Person and (g) any amendment to any of the foregoing.

"Party" or "Parties" means each and all of Buyer and Sellers.

"Permits" means all Regulatory Approvals, licenses, approvals, permits, product registrations, listings, certifications and other authorizations (and any applications for the foregoing) issued by a Governmental Authority, notified body, accrediting authority, institutional review boards, ethics committees, or other recognized standard setting body.

"Permitted Encumbrances" means: (a) Encumbrances for Taxes not yet due and payable, or being contested in good faith and for which appropriate reserves have been established in accordance with GAAP; (b) Encumbrances in respect of property or assets (other than Intellectual Property) imposed by Law that were incurred in the ordinary course of business, such as carriers', warehousemen's, materialmen's and mechanics' liens and other similar liens; (c) pledges or deposits made in the ordinary course of business to secure obligations under workers' compensation Laws or similar legislation or to secure public or statutory obligations; (d) Encumbrances that shall be released and, as appropriate, removed of record, at or prior to the Closing in accordance with the terms of this Agreement; (e) restrictions on the transfer of securities arising under securities Laws and (f) in addition, with respect to the Leased Real Property, (i) reciprocal easement agreements, utility easements and other customary encumbrances on title, (ii) Encumbrances which would be disclosed on a current title report or similar report or listing relating to the Leased Real Property, (iii) zoning, ordinances, building codes, regulations and enactments of any Governmental Authority having jurisdiction over the Leased Real Property, and (iv) any conditions that would be shown by a current (as of the date of this Agreement) and accurate survey or personal inspection of the Leased Real Property, provided that such matters described in clauses (i) through (iv) do not, individually or in the aggregate, materially impair the present use of the Leased Real Property in the operation of the Business affected thereby.

"Person" means any individual or any general partnership, limited partnership, limited liability partnership, limited liability company, corporation, joint venture, trust, business trust, cooperative or association, Governmental Authority or any foreign trust or foreign business organization, and the heirs, executors, administrators, legal representatives, successors, and assigns of such "Person" where the context so permits.

"Personal Information" means any information that identifies or can be used to identify a natural person, including any information defined as "personal data," "personally identifiable information," "personal information," "protected health information," or "nonpublic personal information" under applicable Data Protection Laws.

"Portal" means any and all versions of the website located at fc2condoms.com and fc2.us.com related to marketing, selling, and distributing the Product (including enabling users to obtain telehealth or other medical services and access to the Product).

"Pre-Closing Taxable Period" means any taxable period (or portion thereof) ending on or before the Closing Date, including, in the case of a Straddle Period the portion of such Straddle Period ending at the end of the day on the day before the Closing Date.

"Pre-Closing Taxes" means any unpaid income Taxes of the Business Subsidiaries for any Pre-Closing Taxable Periods that are not reflected in the calculation of the Closing Working Capital or the Closing Indebtedness Amount and solely to the extent such income Taxes relate to Tax Returns of the Business Subsidiaries that are first due (with extension) after the Closing Date and, shall be calculated using the Business Subsidiaries' historic method of accounting for income Tax purposes to the extent permitted by applicable Law and taking into account any transaction Tax deductions, and applicable prepayments or estimated payments of Tax, and available net operating losses and similar tax assets actually available to be utilized against Taxes during such Pre-Closing Taxable Periods but excluding for the avoidance of doubt, any such net operating losses and similar tax assets that are carried forward to or utilized in any Tax period or portion thereof beginning after the Closing Date.

"Proceeding" means (a) any Claim or other proceeding before, commenced by or filed with a Governmental Authority, or (b) any investigation, inquiry, charge, audit or examination involving a Governmental Authority.

"Product" means any female or internal condom product of any Business Entity or female or internal condom product candidate of any Business Entity under development, being tested, being commercialized, distributed, approved and/or otherwise authorized for marketing by the FDA or a comparable foreign Government Authority, or planned to be marketed, including but not limited to the Class III medical device approved under Premarket Approval (PMA) Number P080002, supplemental approval (S001) and/or(S002), or similar marketing authorization by a comparable foreign Governmental Authority (subsequently down-classed to a Class II with special controls), any health care product candidate or accessory developed by any Business Entity, in each case relating to contraception or the prevention of sexually transmitted infections, that is subject to a granted or pending marketing application, clearance, approval, certification, or other regulatory authorization, and/or any supplements, modifications, derivatives, variant, or product or labeling extensions, or services related thereto, whether occurring prior to or after the Closing, including without limitation, any versions with scents or colors or any other new female condom.

"Regulatory Approvals" means, with respect to the Product, any and all approvals, licenses, certifications, clearances, listings, permissions, consents, registrations and/or other authorizations of any Governmental Authority necessary to manufacture, process, store, handle, test, import, export, distribute, sell or market the Product, whether held by a Seller or an Affiliate of a Seller, including, where applicable, pre- and post- marketing authorizations and labeling approvals.

"Regulatory Documentation" means, with respect to the Product, all (i) documentation and records concerning and supporting any Regulatory Approvals and necessary for maintaining Regulatory Approvals, including all documentation and records related to regulatory compliance and quality controls, and all source data and reports from nonclinical and clinical trials of the Product, to the extent that such source data is in the possession or control of a Seller or any of its Subsidiaries, regulatory applications, submissions, reports, and filings in connection therewith, (ii) correspondence and reports related to the Product or otherwise relevant to the ability to commercially distribute, sell or market the Product submitted to, received from, or required to be retained by any Governmental Authorities (including minutes and official contact reports relating to any communications with any Governmental Authority) and relevant supporting documents with respect to such correspondence and reports, including, but not limited to, documents demonstrating compliance with manufacturing, formulation, labeling, and advertising regulatory requirements, and adverse events, complaint handling, and medical device reports (periodic and expedited) and related records, and (iii) data (including clinical trial, nonclinical, and post-market data) referenced in any of the foregoing, all documentation related to any study or testing conducted by or on behalf of the Seller or any Affiliate of Seller with respect to any Product, including but not limited to the study protocol, informed consent documentation, material correspondence with institutional review boards or ethics committees, study data, statistical analyses, data reports, and publications, in each case to the extent in the possession or control of a Seller or any of its Subsidiaries; provided that the Regulatory Documentation shall not be deemed to include any correspondence, reports, documents or data to the extent applicable Law prohibits their transfer, or where transfer thereo

"Related To" means related to, used or held for use in, or necessary for.

"Representatives" means, as to any Person, any of such Person's officers, directors, employees, consultants, investment bankers, lenders, attorneys, accountants, agents and other advisors or representatives.

"Retention Escrow Account" means the account established by the Escrow Agent pursuant to the terms of the Escrow Agreement for purposes of holding the Retention Escrow Amount.

"Retention Escrow Amount" means \$54,000.

"R&W Policy" means the buyer-side representations and warranties insurance policy issued by ANV Global Services, Inc. (the <u>R&W Insurer</u>") to Buyer, the premium and costs of which shall be borne by the Buyer other than the Seller R&W Contribution.

"R&W Exclusion" shall mean the following matters listed in Section III (Exclusions), Item (B) of the R&W Policy: romanette (ii), (iv), (vii), (viii), (viii), and (ix).

"Seller R&W Contribution" means \$125,750.

"Straddle Period" means any taxable period that begins before and ends after the Closing Date.

"Subsidiary" means with respect to any Person, (a) any corporation at least a majority of the outstanding voting stock of which is owned, directly or indirectly, by such Person or by one or more of its subsidiaries, or by such Person and one or more of its subsidiaries, (b) any general partnership, joint venture, limited liability company, statutory trust, or other entity (other than a limited partnership), at least a majority of the outstanding partnership, membership, or other similar interests of which shall at the time be owned by such Person, or by one or more of its subsidiaries, or by such Person and one or more of its subsidiaries, and (c) any limited partnership of which such Person or any of its subsidiaries is a general partner. For the purposes of this definition, "voting stock" means shares, interests, participations, or other equivalents in the equity interest (however designated) in such Person having ordinary voting power for the election of a majority of the directors (or the equivalent) of such Person.

"SWK" means SWK Funding LLC, a Delaware limited liability company.

"SWK Royalty Agreement" means that certain Residual Royalty Agreement dated as of March 5, 2018 between Veru and SWK, as amended.

"Target Working Capital" means an amount equal to \$5,700,000, the calculation of which is set forth on Schedule 1-4 for illustrative purposes only.

"Tax" means (i) any income, alternative or add-on minimum tax, gross income, estimated, gross receipts, sales, use, ad valorem, value added, transfer, franchise, capital stock, profits, license, registration, withholding, payroll, social security (or equivalent, including employer and employees' contributions), employment, unemployment, disability, excise, severance, stamp, occupation, premium, property (real, tangible or intangible), environmental or windfall profit tax, custom duty or other tax, governmental fee or other like assessment or charge of a similar kind in the nature of a tax, together with any interest or any penalty, addition to tax or additional amount (whether disputed or not) imposed by any Governmental Authority responsible for the imposition of any such tax (domestic or foreign), (ii) any liability for the payment of any amounts of the type described in clause (i) of this sentence as a result of being a transferee of or successor to any Person or as a result of any express or implied obligation to assume such Taxes or to indemnify any other Person, including by operation of Law.

"Tax Returns" means any return (including any information return), report, statement, declaration, estimate, schedule, notice, notification, form, election, certificate or other document filed with or submitted to, or required to be filed with or submitted to, any Governmental Authority (foreign or domestic) in connection with the determination, assessment, collection or payment of any Tax or in connection with the administration, implementation or enforcement of or compliance with any Law relating to any Tax, including any schedule or amendment thereof or attachment thereto.

"Technology" means all know-how, techniques, methods, processes, designs, schematics, trade secrets, inventions (whether or not patentable) and invention disclosures, discoveries, tools, improvements, software (including computer programs, computer files, algorithms, models, routines, modules, source code, object code, design tools, user interfaces and other software and all related documentation (including developer notes, comments, annotations, error logs, bug reports and training materials), in each case, in any form or format, however fixed), data, compilations of data and databases, and works of authorship (including documentation), any media on which any of the foregoing is recorded, and any other tangible embodiments of any of the foregoing.

"Transferred Business Intellectual Property" means (a) the Assigned Trademarks, and (b) the Assigned Other IPR.

"Transferred Business Technology" means all Technology Related To the Business that is owned or purportedly owned by Veru, including (a) any Technology constituting, included in, or embodied by any part of any and all versions of a Product or the Portal, (b) know-how and other Technology Related To the Business and known by any Business Employees, or (c) the Technology otherwise listed or described on Schedule 1-5(c).

"Transferred Tangible Assets" is defined in Section 2.02(a)(vi).

"Transaction Documents" means the agreements, instruments, certificates and documents being or to be executed and delivered under this Agreement or in connection with the transactions contemplated hereby.

"UK Stamp Tax" means the stamp tax arising as a result of the transfer of FHC UK at Closing.

"UK Transfer Documentation" means the transfer paperwork substantially in the form attached hereto as Exhibit F, and any other documentation necessary to effectuate the transfer of the Stock.

"Working Capital Collar" means \$250,000.

Certain other defined terms used in this Agreement shall have the meanings given to them in the sections set forth below:

Defined Term	Section
Actual Pre-Closing Taxes	Section 5.09(f)
Agreement	Preamble
Arbitrating Accountant	Section 2.08(b)
Assignment and Assumption Agreement	Section 2.06(a)(iii)
Assumed Business Contracts	Section 2.02(a)(iii)
Assumed Liabilities	Section 2.03(a)
Bankruptcy and Equity Exception	Section 3.02(a)
Basket	Section 6.04(c)
Bill of Sale	Section 2.06(a)(ii)
Books and Records	Section 2.02(a)(vii)
Business Benefit Plans	Section 3.09(b)
Business Subsidiaries Cash Target	Section 5.12
Buyer	Preamble
Buyer Benefit Programs	Section 5.04(a)
Buyer Indemnified Parties	Section 6.01
Buyer's Damages	Section 6.01
Cap	Section 6.04(d)
Claim	Section 3.07
Closing	Section 2.05
Closing Cash	Section 2.08(a)
Closing Date	Section 2.05
Closing Statement	Section 2.08(a)
Closing Working Capital	Section 2.08(a)
Consent Schedule	Section 3.02(b)
Counsel	Section 5.03
De Minimis Basket	Section 6.04(b)
Dispute Notice	Section 2.08(b)
Estimated Closing Consideration	Section 2.04
Excluded Assets	Section 2.02(b)

Excluded Business Contract	Section 2.02(b)(viii)
Excluded Lease	Section 2.02(b)(xviii)
Excluded Liabilities	Section 2.03(b)
FHC Malaysia	Recital B
FHC UK	Recital B
Financial Statements	Section 3.05(a)
Fundamental Representations	Section 6.04(a)
Indemnified Party	Section 6.03(a)
Indemnifying Party	Section 6.03(a)
Insurance Policies	Section 3.17
Interim Balance Sheet	Section 3.05(a)
International Benefit Plans	Section 3.09(b)
Inventory	Section 2.02(a)(ii)
Lease	Section 3.10
Leased Real Property	Section 3.10
Limited	Preamble
Malaysia Stock	Recital B
Material Customers	Section 3.22(a)
Material Suppliers	Section 3.22(b)
Notice of Claim	Section 6.03(a)
Purchase Price	Section 0
Purchased Assets	Section 2.02(a)
Receivables	Section 2.02(a)(i)
Schedules	Section 3
Seller(s)	Preamble
Seller Indemnified Parties	Section 6.02
Sellers' Damages	Section 6.02
Stock	Recital B
Straddle Returns	Section 5.09(b)
SWK Payoff Letter	Section 2.06(a)(viii)
Tangible Personal Property	Section 2.02(a)(vi)
Tax Representations	Section 6.04(a)(i)
Third Party Claim	Section 6.03(b)
Transfer Taxes	Section 5.06
Transition Services Agreement	Section 2.06(a)(iv)
U.S. Benefit Plans	Section 3.09(a)
Veru	Preamble

1.02. <u>Construction.</u> In this Agreement (except where the context otherwise requires):

(a) any reference to a Recital, Section, Annex, Exhibit or Schedule is to the relevant Recital, Section, Annex, Exhibit or Schedule of or to this Agreement and any reference to a subsection or paragraph is to the relevant subsection or paragraph of the Section, Exhibit or Schedule in which it appears;

- (b) any reference to legislation or any provision of any legislation shall include any modification, amendment, re-enactment thereof, any legislative provision substituted therefore and all rules, regulations and statutory instruments issued or related to such legislation;
 - (c) the Section headings are included for convenience only and shall not modify or affect the construction or interpretation of this Agreement;
 - (d) all references to accounting terms, including "GAAP" shall mean the then-current common usage of such terms in the United States;
 - (e) references to "Dollars" or "\$" shall mean United States Dollars unless otherwise specifically noted;
 - (f) the words "include," "includes" and "including" shall be deemed to be immediately followed by the term "but not limited to";
- (g) the words "hereof," "herein" and "hereunder" and words of similar import shall be deemed to refer to this Agreement as a whole and not any particular provision of this Agreement;
 - (h) the use of the word "or" shall not be exclusive;
 - (i) the word "will" shall be construed to have the same meaning and effect as the word "shall";
 - (j) terms defined in the singular shall have a comparable meaning when used in the plural and vice versa; and
- (k) the language used in this Agreement shall be deemed to be the language chosen by the Parties hereto to express their mutual intent, and if an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the Parties hereto and no presumption or burden of proof will arise favoring or disfavoring any Person by virtue of the authorship of any of the provisions of this Agreement.

2. Purchase and Sale; Closing.

2.01. <u>Purchase and Sale of Stock.</u> Subject to the terms and conditions set forth in this Agreement, at the Closing, Sellers shall sell, assign, transfer, convey and deliver to Buyer, and Buyer shall purchase from Sellers, the Stock free and clear of all Encumbrances other than Permitted Encumbrances.

2.02. Purchase and Sale of Assets.

(a) <u>Purchased Assets</u>. Subject to the terms and conditions set forth in this Agreement, at the Closing, Veru shall sell, assign, transfer, convey and deliver to Buyer, and Buyer shall purchase from Veru, all of Veru's right, title and interest in, to and under the Purchased Assets free and clear of all Encumbrances other than Permitted Encumbrances. The "<u>Purchased Assets</u>" shall mean all of the assets, properties, and rights of every kind and nature, whether real, personal, mixed, tangible or intangible (including goodwill) wherever located as they exist on the Closing Date which Relate To the Business (or such other standard applicable to such type of asset set forth in clauses (i) – (xiii) below), including the following assets, properties and rights of Veru as they exist on the Closing Date and in all cases expressly excluding the Excluded Assets:

- (i) all accounts receivable, notes receivable, and other receivables (whether or not current and including any amounts received by Veru in connection therewith after the Closing but excluding the Excluded Scheduled Receivables) Related To the Business (collectively, the "Receivables");
- (ii) all raw materials, work-in-process, finished goods, other inventory, and related parts and supplies, in each case including all packaging, labels, and other similar items, Related To the Products or Business (collectively, "Inventory");
- (iii) the rights under the Contracts set forth in Schedule 2.02(a)(iii) ("Assumed Business Contracts"), including all claims or causes of action with respect to the Assumed Business Contracts;
- (iv) all Transferred Business Intellectual Property (including the right to register, prosecute, maintain or record any of such Intellectual Property Rights with any Governmental Authority and the right to all past and future income, royalties, damages and payments due with respect to such Intellectual Property Rights, including rights to damages and payments for past, present or future infringements or misappropriations thereof, as well as all goodwill associated with such Intellectual Property Rights) and Transferred Business Technology;
- (v) all of Veru's rights in all licenses, permits, approvals, authorizations, registrations and certifications of any Governmental Authority issued to or held by Veru Related To the Business;
- (vi) all tangible personal property of any type or nature (other than Inventory and Transferred Business Technology), including all plant, machinery, equipment, supplies, spare parts, tools, leasehold improvements, furniture, furnishings, computer hardware and vehicles, in each case Related To the Business ("Other Transferred Tangible Property") (together with the Transferred Business Technology and Inventory, the "Transferred Tangible Assets");
- (vii) all books and records, including the Regulatory Documentation and any other information, files, invoices, correspondence, records (including billing records), sales and promotional literature, manuals, customer and suppler correspondence, data, plans and reports, including customer, supplier, distribution, price and mailing lists, copies of Tax Returns, and all accounting or other books and records in whatever media retained or stored by Veru that exclusively relate to the Business, the Products, the Portal or the Purchased Assets and those described on Schedule 2.02(a)(vii) (collectively, "Books and Records");
 - (viii) all guaranties, warranties, indemnities and similar rights in favor of Veru related to any item listed in (i) through (vii) above;

- (ix) all contractual rights of Veru to enforce any confidentiality, non-disclosure, non-competition, non-solicitation, or other restrictive covenants in favor of, or similar obligations owed to, Veru Related To the Business;
- (x) all insurance and warranty benefits and proceeds received after the Closing Date with respect to damage, non-conformance of, or loss to the Purchased Assets or with respect to the Purchased Assets or the Assumed Liabilities;
- (xi) all prepaid expenses, credits, advance payments, security deposits, charges, sums, and fees relating to the Purchased Assets, including those set forth on Schedule 2.02(a)(xi):
- (xii) all rights to the claims, counterclaims, credits, causes of action, rights of recovery, rights of setoff or similar rights related to any item listed in (i) through (xi) above, whether arising out of actions or conditions occurring prior to, on or after the Effective Time; and
 - (xiii) all goodwill and the going concern value of the Business.

Notwithstanding anything in this Agreement to the contrary and for clarity, this Section 2.02(a) shall be deemed to cover and include any assets of any type covered by a breach of the final sentence in Section 3.20(a).

- (b) Excluded Assets. Buyer understands and agrees that all assets of Veru that are not covered under<u>Section 2.02(a)</u> (collectively, the "Excluded Assets") shall be excluded from the Purchased Assets and shall be retained by Veru. For the avoidance of doubt, the Excluded Assets shall include the following:
- (i) except as otherwise provided in this Agreement or any other Transaction Document, all rights to receive any administrative and corporate (overhead, shared and other) services and benefits of any kind provided to the Business by Veru, either directly or indirectly through third party service providers;
- (ii) all Intellectual Property of Veru other than the Transferred Business Intellectual Property and Transferred Business Technology (including all corporate and trade names of Veru (including, but not limited to the name, label, logo and mark of "Veru" and any variation or derivation thereof), and all Uniform Resource Locators (or URLs), websites, social media accounts and internet domain names in each case consisting of such names);
- (iii) all rights of Veru under this Agreement, the Transaction Documents and any other documents, instruments or certificates executed in connection with this Agreement and the transactions contemplated hereby;
 - (iv) all assets of Veru Related To Veru's biopharmaceutical, pharmaceutical and drug businesses other than the Purchased Assets;

- (v) all tangible personal property of Veru other than the (a) Inventory, (b) Transferred Business Technology, and (c) all Other Transferred Tangible Property that is not listed specifically in Schedule 2.02(b)(v)(c);
- (vi) information technology system tangible and related tangible assets (including computers, servers and related hardware and software) of Veru other than the (a) Inventory, (b) Transferred Business Technology, and (c) all Other Transferred Tangible Property that is not listed specifically in Schedule 2.02(b)(vi)(c);
- (vii) all cash or cash equivalents, government securities, or investment securities of either Seller (including any related accounts with banks, brokerages, or other similar Persons);
 - (viii) each Contract of Veru (and rights thereunder) that is not an Assumed Business Contract (any such Contract, an <u>Excluded</u>
- (ix) all insurance policies of Veru and all rights, claims or causes of action of Veru of every nature arising under such insurance policies (other than any items covered under Section 2.02(a)(x));
 - (x) the Excluded Scheduled Receivables;
- (xi) all licenses, permits, approvals, authorizations, registrations and certifications of any Governmental Authority that are not included in the Purchased Assets;
 - (xii) all Business Benefit Plans of Veru (including any Contracts related thereto) and all assets held with respect to the Business

Benefit Plans of Veru;

Business Contract");

- (xiii) other than Books and Records in the Purchased Assets, all Books and Records and other protected business information of Veru, including Veru's Organizational Documents, qualifications to conduct business as a foreign corporation, arrangements with registered agents relating to foreign qualifications, taxpayer and other identification numbers, seals, minute books, stock transfer books, blank stock certificates, Tax Returns and Books and Records relating to Veru's Tax Returns or otherwise relating to Tax matters of Veru, for all periods and other documents relating to the organization, maintenance, and existence of Veru as a corporation or other form of legal entity;
 - (xiv) all confidentiality or other agreements of Veru relating to the potential sale of the Business;
 - (xv) all Tax assets (including net operating loss carryforwards and duty and Tax refunds and prepayments) of Veru;
- (xvi) all rights, privileges and claims under any insurance policy, program or arrangement (including, for the avoidance of doubt, under any worker's compensation policy, program or arrangement);

(xvii) all rights, claims and credits to the extent arising out of or relating to any Excluded Asset or any Excluded Liability, including claims in bankruptcy, and any such items arising under guarantees, warranties, offsets, indemnities and all other intangible property rights or claims and similar rights in favor of Veru to the extent arising out of or relating to any Excluded Asset or any Excluded Liability;

- (xviii) the lease agreement for Veru's corporate headquarters in Miami, Florida (the 'Excluded Lease');
- (xix) any attorney-client privilege or other legally recognized privilege, or any product of or documentation relating to due diligence conducted by or for the Business Entities, to the extent relating to Veru, the Purchased Assets, the Assumed Liabilities or the operation of the Business prior to the Closing; and
 - (xx) the assets specifically identified in Schedule 2.02(b)(xx).

2.03. <u>Liabilities</u>.

- (a) <u>Assumed Liabilities</u>. At the Closing, subject to the terms and conditions of this Agreement and as partial consideration for the purchase of the Purchased Assets, Buyer shall assume and agree to pay, perform and discharge as and when due, the following, and only the following, Liabilities of Veru: (a) all Liabilities included in the Closing Working Capital (as finally determined pursuant to <u>Section 2.08(a)</u> of this Agreement); and (b) all Liabilities arising after the Closing under the Assumed Business Contracts, to the extent (and only to the extent) that such Liabilities (i) first arise or accrue after the Closing and relate to the period after the Closing, and (ii) do not arise as a result of any action, inaction, error, omission, breach, or default by Veru or any of its Affiliates (collectively, the "<u>Assumed Liabilities</u>").
- (b) Excluded Liabilities. Except for the Assumed Liabilities, Buyer shall not assume and shall not be responsible to pay, perform or discharge any of, and Veru shall timely pay, perform, satisfy and discharge in accordance with their respective terms, the Liabilities of Veru (including any arising out of, relating to or otherwise in respect of the Business or the Purchased Assets prior to the Closing), including the following (collectively, the "Excluded Liabilities"):
 - (i) any Liabilities relating to or arising out of the Excluded Assets;
 - (ii) all Indebtedness of Veru and its Affiliates;
 - (iii) any Liability for Indemnified Taxes;
- (iv) all Liabilities arising out of or relating to the Assumed Business Contracts that are not Assumed Liabilities, including any Liabilities associated with the McKesson Dispute;
- (v) all Liabilities arising out of or relating to any Proceeding (i) pending at the Effective Time or (ii) initiated after the Effective Time to the extent relating to Veru's ownership of the Purchased Assets or the operation or conduct of the Business prior to the Effective Time, and all Liabilities arising out of or relating to Claims relating to the bankruptcy of The Pill Club Holdings, Inc.;

- (vi) all Liabilities resulting from any products manufactured or sold with respect to the Business, including the Products, prior to Closing, including any product liability claim or recall;
 - (vii) any Liabilities of Veru relating to the conduct or operation of any business of Veru other than the Business;
- (viii) any Liabilities of Veru arising or incurred in connection with the negotiation, preparation, investigation and performance of this Agreement, the Transaction Documents and the transactions contemplated hereby and thereby, including, without limitation, fees and expenses of counsel, accountants, consultants, advisers, brokers, investment bankers and others; and
- (ix) any Liabilities of Veru with respect to (A) any employees or former employees of Veru (including, for the avoidance of doubt, any change of control bonus or severance obligations of Veru with respect to employees or former employees of Veru) and (B) the Business Benefit Plans.

2.04. Consideration.

- (a) At the Closing, Buyer shall (i) pay to Sellers, by wire transfer of immediately available funds, an aggregate amount (the 'Estimated Closing Consideration') equal to (A) \$18,000,000 (the 'Purchase Price'), plus (B) the Estimated Cash, minus (C) the Estimated Working Capital Deficit, if any, minus (D) the Estimated Indebtedness Amount, minus (E) the Seller R&W Contribution, minus (F) the Retention Escrow Amount, minus (G) the Adjustment Escrow, minus, (H) the Estimated Pre-Closing Taxes Amount, minus (I) the payoff amount specified in the SWK Payoff Letter (which shall be paid by Buyer to SWK on behalf of Sellers the payoff amount specified in the SWK Payoff Letter), minus (J) \$22,438, which the parties agree is the portion of the UK Stamp Tax for which Seller is responsible at Closing pursuant to Section 5.06, minus (K) \$11,182 in respect of the McKesson Dispute; and (ii) assume the Assumed Liabilities. The Estimated Closing Consideration shall be adjusted as set forth in Section 2.08.
- (b) Following the Closing until the determination of the Final Closing Consideration pursuant to Section 2.08, upon receipt by the Buyer or any of its Affiliates of payment for any outstanding Receivables, Buyer shall immediately cause payment to the Sellers, by wire transfer of immediately available funds, of the amount of such payment of outstanding Receivables until an aggregate amount equal to the Estimated Working Capital Excess has been paid to Sellers from such payments of outstanding Receivables. Upon the determination of the Final Closing Consideration pursuant to Section 2.08, any part of the Estimated Working Capital Excess which has not been paid to Sellers pursuant to this Section shall be taken into account as provided in Section 2.08.

2.05. Closing. The closing of the transactions contemplated by this Agreement ("Closing") shall take place via the remote exchange of documents and signature pages on the date hereof (the "Closing Date"). The Closing shall be deemed to be effective as of the Effective Time.

2.06. Certain Payments and Deliverables at Closing.

- (a) Sellers shall deliver (or caused to be delivered) to Buyer the following items at or prior to the Closing:
- (i) certificates evidencing the Stock, duly endorsed in blank or accompanied by stock powers or other written instruments of transfer reasonably acceptable to Buyer, or if such certificates are not then available, affidavits of loss in lieu thereof;
 - (ii) a bill of sale, in the form attached as Exhibit C hereto (the "Bill of Sale"), duly executed by Veru;
- (iii) an assignment and assumption agreement, in the form attached as $Exhibit\ D$ hereto (the "Assignment and Assumption Agreement"), duly executed by Veru;
 - $(iv) \quad \text{a transition services agreement, in the form attached as} \underline{Exhibit} \, \underline{E} \, \text{hereto (the ``} \underline{Transition} \, \underline{Services} \, \underline{Agreement'} \text{'}), \, \text{duly executed by}$

Veru;

- (v) assignments, in recordable form satisfactory to Buyer, of all Acquired Registered Business IPR, duly executed by Veru;
- (vi) a certified copy of the resolutions adopted by the Board of Directors of Veru authorizing: (A) the sale by Sellers of the Stock, (B) the sale by Veru of the Purchased Assets, (C) the execution, delivery and performance of this Agreement by Sellers and consummation of the transactions contemplated hereunder, and (D) the execution, delivery and performance by Sellers of all Transaction Documents that are to be executed, delivered and performed by Sellers;
- (vii) a certified copy of the resolutions adopted by the directors of Limited authorizing: (A) the sale by Limited of the Stock to be sold by Limited, (B) the execution, delivery and performance of this Agreement by Limited and consummation of the transactions contemplated hereunder, and (C) the execution, delivery and performance by Limited of all Transaction Documents that are to be executed, delivered and performed by Limited;
- (viii) a payoff letter in form and substance mutually agreeable to the Parties with respect to all amounts owed by Veru under the SWK Royalty Agreement (the "SWK Payoff Letter");
 - (ix) the Escrow Agreement, duly executed by Veru;

- (x) a properly executed IRS Form W-9 or appropriate series of IRS Form W-8 (together with any attachments) of each Seller;
- (xi) releases of all Encumbrances on any of the Purchased Assets (or a commitment by the applicable lender to release upon receipt of amounts due, as stated in any applicable payoff statement or instruction); and
 - (xii) the Transferred Tangible Assets.
 - (b) Buyer shall deliver (or caused to be delivered) to Sellers the following items at or prior to the Closing:
 - (i) the Bill of Sale, duly executed by Buyer;
 - (ii) the Assignment and Assumption Agreement, duly executed by Buyer;
 - (iii) the Transition Services Agreement, duly executed by Buyer;
 - (iv) the Escrow Agreement, duly executed by Buyer and the Escrow Agent;
 - (v) the UK Transfer Documentation, in a form reasonably acceptable to Buyer; and
- (vi) a certified copy of the resolutions adopted by the Board of Directors of Buyer authorizing: (A) the purchase from Sellers of the Stock, (B) the purchase of the Purchased Assets from Veru, (C) the execution, delivery and performance of this Agreement by Buyer and consummation of the transactions contemplated hereunder, and (D) the execution, delivery and performance by Buyer of all Transaction Documents that are to be executed, delivered and performed by Buyer.
- (c) Buyer shall pay or cause to be paid by wire transfer of immediately available funds, on behalf of Seller, the Estimated Indebtedness Amount, to each of payees thereof for whom Seller has provided a payoff letter to Buyer prior to the Closing Date.
 - (d) Buyer shall deposit with the Escrow Agent the Adjustment Escrow Amount to be held in the Adjustment Escrow Account.
 - (e) Buyer shall deposit with the Escrow Agent the Retention Escrow Amount to be held in the Retention Escrow Account.
- (f) Buyer will deliver or cause to be delivered to Sellers by wire transfer of immediately available funds at the Closing, to one or more accounts previously specified in writing by Sellers to Buyer, cash equal to the Estimated Closing Consideration as provided in Section 2.04.

2.07. <u>Closing Estimates</u>. No later than three (3) Business Days prior to the Closing Date, Sellers shall cause to be prepared and delivered to Buyer the Estimated Closing Statement. The Estimated Closing Statement will be prepared in the form and manner, and on a basis consistent with the Accounting Principles. Seller shall reasonably consider any comments or suggestions made by Buyer on the Estimated Closing Statement.

2.08. Purchase Price Adjustment.

- (a) Post-Closing Statement. Within 90 days following the Closing Date, Buyer shall prepare and deliver to Seller a statement (the 'Closing Statement') setting forth Buyer's calculation of (i) Net Working Capital as of immediately prior to the Closing (the 'Closing Working Capital'), (ii) Cash and Cash Equivalents as immediately prior to the Closing (the "Closing Cash"), provided that the amount of Closing Cash shall not exceed \$1,274,056, (iii) aggregate amount of Indebtedness of the Business Subsidiaries unpaid as of immediately prior to the Closing (the "Closing Indebtedness Amount"), and (iv) an updated estimate of the amount of Pre-Closing Taxes (the "Final Pre-Closing Taxes Amount"). It is the intent of the parties that in calculating and determining Closing Working Capital (and the components thereof) under this Section 2.08, none of the parties nor the Arbitrating Accountant shall be permitted to (x) include balance sheet line items or accounts that are different from those included on Schedule 1-4 or (y) make any changes or modifications to the accounting principles, policies, practices, assumptions, procedures, elections, categorizations and methods (including those relating to the nature of accounts and inclusion of balance sheet line items and the level of reserves and/or accruals, any calculations or estimations thereof or any adjustments thereto, and/or any determinations of applicable foreign exchange conversion rates) that were used in the calculation of the Target Working Capital.
- (b) <u>Disputes</u>. Seller may dispute the amounts reflected on the Closing Statement. If Seller so disputes all or any portion of the Closing Statement, Seller shall notify Buyer of such disputes in writing, setting forth in reasonable detail the particulars of such disputes and including a revised version of the Closing Statement reflecting all such disputed amounts (the "Dispute Notice"), within 30 days of its receipt of the Closing Statement. In the event that Seller does not provide a Dispute Notice within such 30-day period, Seller shall be deemed to have accepted the Closing Statement in its entirety, which shall be final, binding and conclusive for all purposes hereunder. In the event any such Dispute Notice is provided within such 30-day period, Buyer and Sellers shall use commercially reasonable efforts for a period of 15 days (or such longer period as they may mutually agree in writing) to negotiate and resolve any disputes by Seller set forth in the Dispute Notice. If, at the end of the 15-day period, Buyer and Sellers do not resolve any such disputes, then Buyer and Sellers shall engage a nationally recognized firm of independent certified public accountants as to which the Sellers and Buyer mutually agree (such consent or agreement not to be unreasonably withheld), to resolve such disputes (the "Arbitrating Accountant"). The Arbitrating Accountant shall be provided with (i) a copy of this Agreement, (ii) the Closing Statement and related supporting detail prepared by Buyers and delivered to Seller, (iii) the Dispute Notice and any supporting detail accompanying such Dispute Notice prepared by Seller, and (iv) any information requested by the Arbitrating Accountant as necessary or appropriate in resolving such dispute. When acting pursuant to this Section 2.08(b), the Arbitrating Accountant shall determine, using the Accounting Principles and following the requirements of Section 2.08(a) whether and to what extent, if any, Buyers' calculations on the Closing Statement requires adjustment. The Arbitrating Accountant shall address only those issues in dispute pursuant to this Section, may not assign a value to any item greater than the greatest value for such item claimed by a Party or less than the smallest value for such item claimed by a Party, and may not apply any accounting methods, treatments, principles or procedures other than the Accounting Principles. Within 30 days following appointment, the Arbitrating Accountant shall deliver its determination of the Closing Statement calculated in accordance with the terms of this Agreement and setting forth its resolution of the disputes. The Arbitrating Accountant shall also apportion its fees and expenses between Sellers, on the one hand, and Buyer, on the other hand, in proportion to the difference between the relative position of each Party and the Arbitrating Accountant's ultimate determination with respect to the amounts in the Closing Statement. The decision and award of the Arbitrating Accountant, including the apportionment of its fees, shall be final and binding on the parties and shall be subject to confirmation and entry of judgment in accordance with applicable Law. In no event shall the Arbitrating Accountant award either Party consequential, incidental or punitive damages.

(c) Determinations; Adjustments.

- (i) Upon the determination of the amount of Closing Cash, Closing Indebtedness Amount, Closing Working Capital (including the amount of the Closing Working Capital Excess or Closing Working Capital Deficit (if any)) and the Final Pre-Closing Taxes Amount, pursuant to this Section 2.08, the parties shall recalculate the Estimated Closing Consideration based on such amounts (such amount, the "Final Closing Consideration").
- (ii) If the Final Closing Consideration is greater than the Estimated Closing Consideration taking into account any part of the Estimated Working Capital Excess which has not previously been paid by Buyer (such excess, if any, the "Positive Adjustment Amount"), then within two (2) Business Days after such determination (A) the Positive Adjustment Amount shall be paid by or on behalf of the Buyer to Veru by wire transfer of immediately available funds to an account designated by Veru and (B) the Buyer and the Seller shall jointly instruct the Escrow Agent to pay the balance of the Adjustment Escrow Account to the Seller by wire transfer of immediately available funds to an account designated by the Seller.

(iii) If the Final Closing Consideration is less than the Estimated Closing Consideration taking into account any part of the Estimated Working Capital Excess which has not previously been paid by Buyer (such deficit, if any, the "Negative Adjustment Amount"), then within two (2) Business Days after such determination (A) the Buyer and the Seller shall jointly instruct the Escrow Agent to pay the Negative Adjustment Amount to the Buyer by wire transfer of immediately available funds to an account designated by the Buyer, and (B) any amounts remaining in the Adjustment Escrow Account following the payment specified in clause (A) shall be paid by the Escrow Agent to the Seller by wire transfer of immediately available funds to an account designated by the Seller. To the extent the Negative Adjustment Amount exceeds the Adjustment Escrow Amount, Seller shall pay such amounts by wire transfer of immediately available funds to an account designated by Buyer.

(d) <u>Payments</u>. Any payment made under this <u>Section 2.08</u>, to the maximum extent permitted by applicable Law, shall be treated for all Tax purposes as an adjustment to the Final Closing Consideration.

2.09. Purchase Price Allocation.

- (a) The Purchase Price shall be allocated between the Sellers in the percentages and amounts as provided on Schedule 2.09.
- (b) The parties shall allocate the consideration paid hereunder with respect to the purchase and sale of the Purchased Assets by Buyer from Veru (including taking into account the Assumed Liabilities to the extent required by applicable Law) among the Purchased Assets using the methodology set forth on Schedule 2.09.
- (c) Unless otherwise required by applicable Law, Buyer and Sellers agree to utilize such values set forth in this Section 2.09 and on Schedule 2.09 for all Tax purposes, including for purposes of filing applicable Tax Returns with applicable Governmental Authorities. None of the parties will voluntarily take any position inconsistent therewith upon examination of any such Tax Return, in any action or proceeding or otherwise with respect to such Tax Returns. The parties each agree to provide the other promptly with any other information required to complete such Tax Returns.

2.10. Non-Assignable Assets.

(a) Notwithstanding anything to the contrary in this Agreement, but subject to the remaining provisions of this Section 2.10. to the extent that the sale, assignment, transfer, conveyance or delivery, or attempted sale, assignment, transfer, conveyance or delivery to Buyer of any asset, property, or right included in the Purchased Asset (including those on Schedule 2.10) would require the consent or authorization of a Person who is not a party to this Agreement (including any Governmental Authority) and such consent or authorization shall not have been obtained prior to the Closing (each such asset, property or right, a "Non-Assignable Asset"), this Agreement shall not constitute a sale, assignment, transfer, conveyance or delivery, or an attempted sale, assignment, transfer, conveyance or delivery of such Non-Assignable Asset, unless and until such consent or authorization has been obtained or Buyer has notified Sellers expressly in writing that any such Non-Assignable Asset should be transferred or assigned hereunder notwithstanding the absence of such consent; provided that this Section 2.10(a) shall not affect whether any asset, property or right shall be deemed to be a Purchased Asset for any other purpose under this Agreement.

- (b) Following Closing, Sellers and Buyer shall use commercially reasonable efforts, and shall cooperate with each other, to obtain any consents or authorizations required to transfer and assign to Buyer all Non-Assignable Assets. Prior to such time that any such authorization or consent is obtained for a Non-Assignable Asset, Sellers and Buyer shall use commercially reasonable efforts to enter into such lawful and reasonable arrangements (such as subleasing, sublicensing or subcontracting) to provide the economic and, to the extent permitted under applicable Law, operational equivalent of the transfer of such Non-Assignable Asset to Buyer as of the Effective Time and the performance by Buyer of its obligations with respect thereto. Without limiting the generality of the foregoing, prior to such time that any authorization or consent required to transfer and assign to Buyer a Non-Assignable Asset is obtained, (i) Sellers shall (a) promptly pay or cause to be paid to Buyer all monies due to or received by Sellers with respect to any such Non-Assignable Asset and (b) enforce, at the written request and at the sole expense of Buyer, any rights of Sellers arising with respect thereto, and (ii) to the extent Buyer receives the benefit of any such Non-Assignable Asset, Buyer shall perform and discharge on behalf of Sellers' covenants and obligations incurred after the Closing with respect to any such Non-Assignable Asset, if any (other than any Liabilities to the extent arising in connection with the breach, nonperformance or defective performance by Sellers of any Contract, which are deemed Excluded Liabilities), thereunder relating to the period following the Effective Time, in accordance with the provisions thereof.
- (c) Once any consent or authorization required to assign and transfer a Non-Assignable Asset is obtained, Sellers shall and hereby do sell, assign, transfer, convey and deliver to Buyer the applicable Non-Assignable Asset to which such consent or authorization relates for no additional consideration.
- (d) From time to time following the Closing, Sellers and Buyer shall, and shall cause their respective Affiliates to, execute, acknowledge and deliver all such further conveyances, notices, assumptions, releases, acquittances and other instruments, and shall take such further actions, as may be necessary or appropriate to transfer fully to, and vest in, Buyer and each of its successors or assigns, all of the properties, rights, titles, interests, estates, remedies, powers and privileges intended to be conveyed to Buyer under this Agreement and to assure fully to Sellers and its Affiliates and each of their respective successors and assigns, the assumption of the liabilities and obligations intended to be assumed by Buyer under this Agreement, and to otherwise make effective the transactions contemplated hereby (including transferring to Buyer any asset or liability contemplated by this Agreement to be a Purchased Asset or an Assumed Liability, respectively, which was erroneously or inadvertently not transferred to Buyer at the Closing).
- (e) Transfer and delivery of the Transferred Business Technology shall include physical or electronic delivery of all Transferred Business Technology, including delivery or production of appropriate documentation thereof as reasonably requested by Buyer to facilitate the transfer and operation of the Business. To the extent reasonably practicable, all Transferred Business Technology to be delivered hereunder shall be delivered by electronic means in a manner specified by Buyer. Neither Veru nor any of its Affiliates shall retain in its possession or control any Transferred Business Technology or any copy thereof or any tangible Purchased Assets (except as may be necessary in connection with the performance of its obligations under this Agreement or any of the other Transaction Documents).
- 2.11. Withholding Taxes. Buyer and its Affiliates and any other withholding agent shall be entitled to deduct and withhold from any amounts payable or otherwise deliverable pursuant to this Agreement such amounts as may be required to be deducted or withheld therefrom under any provision of federal, state, local or foreign Tax Law. Buyer and its Affiliates shall use commercially reasonable efforts to provide Sellers with advance notice of any anticipated withholding (which notice shall provide in reasonable detail the authority, basis and method of calculation for the proposed deduction or withholding) and to cooperate with Sellers to obtain any reduction of or relief from any deduction or withholding from any consideration payable to Sellers hereunder. To the extent such amounts are so deducted or withheld, such amounts shall be timely and properly remitted to the applicable Governmental Authority, shall be treated for all purposes under this Agreement as having been paid to the Person to whom such amounts would otherwise have been paid. To the extent that such amounts are not so deducted and withheld, such Person shall indemnify the applicable withholding agent for any amounts imposed by a Governmental Authority, together with any related Losses.

- 3. <u>Representations and Warranties of Sellers.</u> Sellers hereby jointly and severally make to Buyer the representations and warranties contained in this <u>Section 3</u> as of the date hereof (unless another date is so indicated, then only as of such date). Such representations and warranties shall be subject to the exceptions set forth in the disclosure schedules delivered by Sellers to Buyer, dated as of the date of this Agreement (the "<u>Schedules</u>").
- 3.01. Organization. Veru is duly incorporated, validly existing and in current status under the Laws of the State of Wisconsin. Limited has been duly incorporated and is validly existing under the Laws of England and Wales. Each Seller has all necessary corporate or other organizational power to carry on the Business as now conducted and to own or lease the properties and assets it now owns or leases. FHC UK has been duly incorporated and is validly existing under the Laws of Malaysia. Each Business Subsidiary has all necessary corporate or other organizational power to carry on the Business as now conducted and to own or lease the properties and assets it now owns or leases. Each Seller is duly licensed and qualified to do business in and (to the extent such concept is applicable in the relevant jurisdiction) is in good standing under the Laws of each state or other jurisdiction where the failure to do so would not have, individually or in the aggregate, a Business Material Adverse Effect. Each Business Subsidiary is duly licensed and qualified to do business in and (to the extent such concept is applicable in the relevant jurisdiction) is in good standing under the Laws of each country or other jurisdiction where the failure to do so would not have, individually or in the aggregate, a Business Material Adverse Effect.

3.02. Authorization of Agreement; No Violation.

(a) Each Seller has all requisite corporate power and authority to execute and deliver this Agreement. Each Business Entity has all requisite corporate power and authority to execute and deliver the other Transaction Documents to which it is a party and to consummate the transactions provided for herein and therein. The execution and delivery of this Agreement by each Seller and the other Transaction Documents to be executed and delivered by each Seller and the performance by each Seller of the obligations to be performed hereunder and thereunder have been duly authorized by all requisite corporate action. The execution and delivery of the other Transaction Documents to be executed and delivered by each Business Subsidiary and the performance by each Business Subsidiary of the obligations to be performed hereunder have been duly authorized by all requisite corporate action. This Agreement is, and each other Transaction Documents to be executed by each Seller or Business Subsidiary signatory will be when so executed, a valid and binding obligation of each such Seller or Business Subsidiary, as applicable, enforceable in accordance with its respective terms, except that enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors rights generally and by general equitable principles (the "Bankruptcy and Equity Exception").

(b) Except as set forth on Schedule 3.02 (the "Consent Schedule"), the execution and delivery of this Agreement and the other Transaction Documents to be executed and delivered by each Seller or Business Subsidiary, as applicable, and the consummation of the transactions contemplated hereby and thereby do not and will not, with or without the giving of notice or the passage of time, violate, conflict with, result in or constitute a breach, default, right to accelerate or loss of rights under, or result in the creation of any Encumbrance pursuant to, the terms or conditions of such Seller's Organizational Documents or Business Subsidiary's Organizational Documents, as applicable, any Legal Requirement or any mortgage, lease, franchise, license, Contract, agreement and/or instrument to which such Seller or any of its Subsidiaries is a party or by which such Seller or any of its Subsidiaries is bound.

3.03. Capitalization of Business Subsidiaries.

- (a) The authorized, issued and outstanding capital stock or other equity interests of each Business Subsidiary is as set forth on Schedule 3.03(a). All of the issued and outstanding capital stock or other equity interests of each Business Subsidiary (including the Stock) is duly authorized, validly issued, fully paid and non-assessable. There are no outstanding subscriptions, options, warrants, commitments, preemptive rights, deferred compensation rights, agreements, arrangements or commitments of any kind to which either Seller or any Business Subsidiary is a party relating to the issuance of, or outstanding securities convertible into or exercisable or exchangeable for, any equity interests of any Business Subsidiary, nor is either Seller or any Business Subsidiary committed to issue any such option, warrant, right of any type or security or to enter into any such agreement, arrangement or commitment. There are no agreements or understandings to which either Seller or any Business Subsidiary is a party with respect to the voting of any equity interests of any Business Subsidiary or which restrict the transfer of any such equity interests. There are no outstanding contractual obligations of either Seller or any Business Subsidiary to repurchase, redeem or otherwise acquire any equity interests of any Business Subsidiary.
- (b) Except as set forth on Schedule 3.03(b) and except for any Business Subsidiary that is a Subsidiary of another Business Subsidiary, none of the Business Subsidiaries controls, directly or indirectly, or has any direct or indirect equity participation or investment in, any corporation, limited liability company, unlimited liability company, partnership, trust or other business association (other than in a money market or other mutual fund or other short-term investment).
- (c) Sellers legally and beneficially own all of the Stock. FHC UK owns of record and beneficially all of the Malaysia Stock. The Stock is, and when delivered by Limited to Buyer at the Closing pursuant to this Agreement, will be, free and clear of any and all Encumbrances, other than applicable restrictions on the transfer of securities under securities Laws and Encumbrances arising from acts of Buyer from and after the Closing Date. The Malaysia Stock is, and at the Closing will be, free and clear of any and all Encumbrances, other than applicable restrictions on the transfer of securities under securities Laws and Encumbrances arising from acts of Buyer from and after the Closing Date.

3.04. Consents and Approvals. Except as set forth on the Consent Schedule, no consent, approval or authorization of, or declaration, filing or registration with, any Governmental Authority, and no consent, notice, approval or authorization of, or declaration, filing or registration with, any other Person is required in connection with either Seller's and its Subsidiaries' sale, transfer and assignment of the Purchased Assets to Buyer, or continued conduct of the Business, and the execution and delivery of this Agreement and the other Transaction Documents to be executed and delivered by such Seller and/or the consummation or performance of any of the transactions contemplated hereby or thereby.

3.05. Financial Statements.

- (a) True, correct and complete copies of (i) the unaudited consolidated balance sheet of the Business as of September 30, 2024 and 2023 and the related unaudited consolidated statements of operations and cash flows of the Business for the years then ended (the balance sheet as of September 30, 2024 is referred to herein as the "Balance Sheet") and the related unaudited consolidated statements of operations and cash flows of the Business for the period then ended (the Financial Statements") are attached hereto as Schedule 3.05(a). The Financial Statements have been prepared from the books and records of Sellers and in all material respects in accordance with GAAP applied on a consistent basis for all periods represented by the Financial Statements, except that the Financial Statements do not include footnotes that may be required under GAAP. Subject to Section 3.05(b), (i) the statements of operations included in the Financial Statements present fairly in all material respects the consolidated results of operations of the Business on a pro forma stand-alone basis for the periods covered thereby, and (ii) the balance sheets included in the Financial Statements present fairly in all material respects the consolidated financial condition of the Business as of its date, less the Excluded Assets and the Excluded Liabilities. The Business Entities' books and records are complete and correct and accurately reflect all of the assets, Liabilities, transactions, and results of operations of the Business in all material respects.
- (b) Section 3.05(a) is qualified by the fact that the Business Subsidiaries and the Purchased Assets comprising the Business have not operated as a separate "stand-alone" entity within Veru. As a result, the Financial Statements have been prepared (i) as pro-forma adjusted consolidated financial statements to provide Sellers' commercially reasonable estimate of the consolidated results of operations of the Business on a stand-alone basis and (ii) by the elimination of common expenses or the normalization of items, other than the Purchased Assets and the Assumed Liabilities. Such allocations of charges and credits do not necessarily reflect the amounts that would have resulted from arms-length transactions or the actual costs that would be incurred if the Business operated as an independent enterprise.
- (c) All Receivables whether or not reflected on the Balance Sheet, or that have arisen since the date of the Balance Sheet, (i) represent, valid claims for bona fide, arms-length sales of goods and services actually made by the Business Entities in the ordinary course of business and (ii) are the valid and legally binding obligations of the parties obligated to pay such amounts. To Sellers' Knowledge, none of the Receivables is subject to any set off or counterclaim or is in dispute.

- (d) Except as set forth on Schedule 3.05(d), the Inventory is of a quality and quantity usable and saleable in the ordinary course of business consistent with past practices. The value at which the Business carries the Inventory on the Balance Sheet reflects its customary inventory valuation policy of stating inventory at the lower of cost (based on the first-in, first-out (FIFO) method) or market all in accordance with GAAP. The quantity of Inventory set forth on the Balance Sheet is sufficient and adequate for the conduct of the Business as it previously has been conducted.
- (e) Except as set forth on Schedule 3.05(e) or in the Financial Statements, there are no Liabilities of the Business, of either Seller related to the Business, or of Business Subsidiaries of any character or nature (whether or not required to be disclosed in accordance with GAAP), other than those that have been incurred in the ordinary course of business since June 30, 2024 and that are not material, individually or in the aggregate, to the Business, taken as a whole. Neither Seller is insolvent, nor will either Seller be rendered insolvent by any of the transactions contemplated herein. Except as set forth on Schedule 3.05(e), no Liability of the Business, of Seller Related To the Business Subsidiaries arises out of, relates to, is in the nature of or were caused by any breach of Contract, breach of warranty, tort, infringement or violation of Law. Veru does not have any Indebtedness pursuant to the Paycheck Protection Program under Sec. 1102, et seq., of the CARES Act. There are no Liabilities of the Business Subsidiaries of any character or nature (whether or not required to be disclosed in accordance with GAAP) arising or incurred in connection with the negotiation, preparation, investigation and performance of this Agreement, the Transaction Documents and the transactions contemplated hereby and thereby, including, without limitation, fees and expenses of counsel, accountants, consultants, advisers, brokers, investment bankers and others.
- (f) Veru and its Subsidiaries maintain a system of "internal control over financial reporting" (as defined in Rules 13a-15(f) and 15d-15(f) of the Securities Exchange Act of 1934) that is sufficient to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP including policies and procedures that: (i) require the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of the assets of Veru and its Subsidiaries; (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP and that receipts and expenditures of Veru and its Subsidiaries are being made only in accordance with appropriate authorizations of Veru's management and its Board of Directors; and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the assets of Veru and its Subsidiaries.
- 3.06. <u>Absence of Changes</u>. Except as set forth on <u>Schedule 3.06</u>, since October 1, 2023, (i) there has not been any Business Material Adverse Effect, and (ii) the Business has been operated only in the ordinary course of business consistent with past practices, except in connection with this Agreement and the other Transaction Documents and the discussions, negotiations, actions and transactions related hereto and thereto. Except as set forth on <u>Schedule 3.06</u>, since October 1, 2023:
- (a) no Business Entity has discharged or satisfied any Encumbrances or paid any material obligation or Liability, in each case relating to the Business, other than in the ordinary course of business, or cancelled, compromised, waived, or released any material right or claim;

- (b) no Business Entity has (i) sold, assigned, licensed, or transferred any of its assets Relating To the Business, except for sales in the ordinary course of business, (ii) mortgaged, pledged, or subjected any of its assets (including any Acquired Business Intellectual Property) to any Encumbrance that has not been satisfied or reserved, other than a Permitted Encumbrance, or (iii) sold, transferred or otherwise disposed of, leased, licensed, abandoned or allowed to enter the public domain, or otherwise distributed any Acquired Business Intellectual Property;
 - (c) no Business Entity has sold, assigned, transferred, abandoned, or permitted to lapse any Permits relating to the Business;
- (d) no Business Entity has suffered any extraordinary loss, damage, destruction, or casualty loss or waived any rights of material value, whether or not covered by insurance and whether or not in the ordinary course of business;
- (e) no Business Subsidiary has commenced any Proceeding or binding dispute resolution process or settled or compromised any pending or threatened Proceeding;
- (f) no Business Entity changed the compensation payable or benefits provided to, or made any other material change in the employment terms for, any Business Employees, except as required by applicable Law or the terms of any Contract or employee benefit plan in existence on the date hereof and made available to Buyer;
 - (g) no Business Subsidiary incurred, assumed, or guaranteed any Indebtedness of any third party;
 - (h) no Business Entity has made or incurred any capital expenditures relating to the Business in excess of \$25,000 in the aggregate;
- (i) no Business Subsidiary has made, rescinded, revoked or changed any material election with respect to Taxes, changed any Tax accounting period, adopted or changed any accounting method with respect to Taxes, filed any amended Tax Return, entered into an agreement with respect to Taxes with any Governmental Authority (including a "closing agreement" under Section 7121 of the Code), surrendered any right to claim a refund for Taxes, or consented to an extension or waiver of the statute of limitations applicable to any Tax claim or assessment; or
 - (j) no Business Subsidiary has entered into any other material transaction, other than in the ordinary course of business.
- 3.07. Proceedings; Orders. Except as set forth on Schedule 3.07, (a) there is no claim, dispute, demand, complaint, action, cause of action, charge, inquiry, audit, notice of violation, litigation, citation, summons, subpoena, investigation, arbitration, lawsuit, hearing or other proceeding (each, a "Claim"), pending or, to Sellers' Knowledge, threatened against the Business Subsidiaries, (b) there are no Claims pending or, to Sellers' Knowledge, threatened against either Seller (or any of its Affiliates) affecting or in any way relating to the Business or the Purchased Assets, (c) there were no Claims involving the Purchased Assets, the Business Subsidiaries or the Business in the five years immediately prior to the date of this Agreement, and (d) there are no Orders affecting the Purchased Assets, the Business Subsidiaries or the Business.

3.08. <u>Taxes</u>.

- (a) Each Seller (i) has timely filed (or will timely file) with the appropriate Governmental Authorities all Tax Returns required to be filed by it with respect to the Business and the Purchased Assets and (ii) has timely paid (or will timely pay) all Taxes it owes with respect to the Business and the Purchased Assets whether or not shown as payable on such Tax Returns. Each Business Subsidiary (i) has timely filed (or will timely file) with the appropriate Governmental Authorities all Tax Returns required to be filed by it and (ii) has timely paid (or will timely pay) all Taxes it owes whether or not shown as payable on such Tax Returns. All such Tax Returns filed by a Business Entity (in the case of a Seller, with respect to the Business or the Purchased Assets) are true and correct in all material respects. Each Business Entity (in the case of a Seller, with respect to the Business or the Purchased Assets) has withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, shareholder or other Person, and each Business Entity (in the case of a Seller, with respect to the Business or the Purchased Assets) has complied in all material respects with all reporting and recordkeeping requirements in connection therewith. There are no Encumbrances (other than Encumbrances for Taxes not yet due and payable) for Taxes on any of the Purchased Assets.
- (b) There is no Proceeding or audit now pending against any Business Entity in respect of any Taxes or assessments (in the case of a Seller, with respect to the Business or the Purchased Assets), and no Business Entity has waived any applicable statute of limitations with respect to such Taxes or agreed to any extension of time with respect to such Taxes (in the case of a Seller, with respect to the Business or the Purchased Assets). To Seller's Knowledge, no claim has been made in writing by a Governmental Authority that a Business Entity may be subject to taxation (in the case of a Seller, with respect to the Business or the Purchased Assets) in a jurisdiction in which such Person is not currently filing Tax Returns.
 - (c) [Intentionally omitted.]
- (d) None of the Business Entities is a party to or bound by any closing or other agreement or ruling with any Governmental Authority with respect to Taxes (in the case of a Seller, with respect to the Business or the Purchased Assets). There are no matters primarily relating to Taxes under discussion between any Governmental Authority and any of the Business Entities (in the case of a Seller, with respect to the Business or the Purchased Assets).
- (e) None of the Assumed Liabilities are or include, and no Business Entity (in the case of a Seller, with respect to the Business or the Purchased Assets) nor any of its Affiliates is a party to, a Tax sharing, Tax allocation or similar arrangement (including indemnity arrangements), pursuant to which Buyer or any of its Affiliates (including, after the Closing, the Business Subsidiaries) may have any obligations to make payments after the Closing (other than a commercial Contract entered into in the ordinary course of business the primary purpose of which is not a Tax matter).

- (f) There are no Tax Liabilities of any Business Entity (in the case of a Seller, with respect to the Business or the Purchased Assets) or its Affiliates that could result in Liability to Buyer as a transferee or successor or otherwise attach to any of the Purchased Assets.
- (g) Each Business Entity has, to the extent applicable, furnished properly completed exemption certificates and maintained all such records and supporting documents in respect of any sales and use Taxes (in the case of a Seller, Related To the Business and the Purchased Assets) in the manner required by applicable Law.
 - (h) Each Business Subsidiary is treated as a foreign corporation for U.S. federal tax purposes.
- (i) None of the Business Subsidiaries has been intended to constitute either a "distributing corporation" or a "controlled corporation" in a distribution of stock intended to qualify under Section 355 of the Code.
- (j) None of the Business Subsidiaries, nor any predecessor thereof has (A) ever been a member of an affiliated, consolidated, combined or unitary group for Tax purposes, (B) ever been a party to, or bound by, any Tax sharing, indemnification, reimbursement or allocation agreement (other than a commercial Contract entered into in the ordinary course of business the primary purpose of which is not a Tax matter), or owes any amount under any such agreement, (C) any potential liability for the Taxes of any person as a transferee or successor, by operation of Law, or by Contract (other than a commercial Contract entered into in the ordinary course of business the primary purpose of which is not a Tax matter), or (D) ever been a party to any joint venture, partnership or other agreement that was treated as a partnership for Tax purposes.
- (k) All transactions and agreements occurring prior to the Closing between the Business Entities and any related parties (in the case of a Seller, with respect to the Business or the Purchased Assets) were made on arm's length terms.
- (l) No Business Subsidiary is a "passive foreign investment company" within the meaning of Section 1297 of the Code or a "controlled foreign corporation" within the meaning of Section 957 of the Code.
- (m) Notwithstanding anything to the contrary in this Agreement, Sellers make no representations or warranties in respect of the existence, amount or usability of any tax attributes, including, without limitation, net operating losses, capital loss carryforwards, foreign tax credit carryforwards, research and development credits, asset bases and depreciation periods for taxable periods or portions thereof beginning after the Closing Date.

3.09. Employee Benefit Plans.

- (a) Schedule 3.09(a) sets forth a true, correct, and complete list of every material employee benefit plan, within the meaning of section 3(3) of ERISA, currently maintained, sponsored or contributed to by either Seller or any ERISA Affiliates for the benefit of any Business Employee who performs services within the United States (the "U.S. Benefit Plans"). Except as set forth on Schedule 3.09(a), (i) each U.S. Benefit Plan intended to be qualified under section 401(a) of the Code has received a favorable determination or opinion letter from the IRS regarding its qualifications thereunder; (ii) each U.S. Benefit Plan has been administered in accordance with its terms and requirements of applicable Law; (iii) no U.S. Benefit Plan is subject to Title IV of ERISA or section 412 of the Code or is a "multimployer plan," as defined in section 3(37) of ERISA or a "multiple employer welfare arrangement" (as defined in Section 3(40) of ERISA); (iv) no U.S. Benefit Plan provides for post-employment life or health insurance benefits for any participant or any beneficiary of a participant, except as may be required by part 6 of Subtitle B of Title 1 of ERISA or similar state, local or foreign Law; (v) since the date of the Balance Sheet, no U.S. Benefit Plan has been amended or modified in a manner that increases in any material amount the benefits payable pursuant to such U.S. Benefit Plan other than in the ordinary course of business or as required by applicable Law; and (vi) no actions or proceedings have been taken or instituted to terminate or wind-up a U.S. Benefit Plan. All of the U.S. Benefit Plans comply in form and in operation in all material respects with their terms, the applicable requirements of ERISA, the applicable requirements of the Code, and all applicable Laws.
- (b) Schedule 3.09(b) sets forth a list of every material benefit plan currently maintained, sponsored or contributed to by each Seller or any Subsidiary of either Seller for the benefit of any Business Employee who performs services outside the United States other than governmental plans or statutorily required plans (the "International Benefit Plans" and, collectively with the U.S. Benefit Plans, the "Business Benefit Plans"). With respect to each International Benefit Plan except for such matters as would not have, individually or in the aggregate, a Business Material Adverse Effect: (i) each International Benefit Plan required to be registered has been registered and has been maintained in good standing (to the extent such concept exists in the relevant jurisdiction) with the appropriate regulatory authorities; (ii) each International Benefit Plan has been administered in accordance with its terms and applicable Law; (iii) since the date of the Balance Sheet, no International Benefit Plan has been amended or modified in a manner that increases in any material amount the benefits payable pursuant to such International Benefit Plan other than in the ordinary course of business or as required by applicable Law; (iv) no actions or proceedings have been taken or instituted to terminate or wind-up an International Benefit Plan; and (v) no International Benefit Plan provides for post-employment life or health insurance benefits for any participant or any beneficiary of a participant, except as may be required by applicable Law.
- (c) No litigation or other Claims (other than routine claims for benefits) is pending or, to Sellers' Knowledge, asserted, in each case against any of the Business Benefit Plans.
- (d) Each U.S. Benefit Plan that is a "non-qualified deferred compensation plan" within the meaning of Section 409A(d)(1) of the Code and any award thereunder, in each case that is subject to Section 409A of the Code, has been administered and drafted or amended, in such a manner so that the additional tax described in Section 409A(1)(B) of the Code will not be assessed against any individual participating in any such non-qualified deferred compensation plan with respect to benefits due or accruing thereunder. There have been no "prohibited transactions" (within the meaning of Section 406 of ERISA or Section 4975 of the Code) with respect to any U.S. Benefit Plan.

- (e) Except as set forth in Schedule 3.09(e), the consummation of the transactions contemplated by this Agreement will not result in any Liability of any Business Entity for any payment to or other compensation or benefits of any current or former employee or director of any Business Entity, or the acceleration of payment, vesting or funding of any compensation or benefits of any Business Employee.
- 3.10. Real Property. None of the Business Entities owns any real property. Except for the Excluded Lease Schedule 3.10 sets forth a list of all real property (i) leased or possessed by any Business Subsidiary or (ii) leased or possessed by a Seller and used primarily in connection with the Business (collectively, the "Leased Real Property"). With respect to the items listed on Schedule 3.10 (each a "Lease"):
- (a) each Business Entity has a valid and enforceable leasehold interest to the leasehold estate in the Leased Real Property granted to such Business Entity pursuant to such Lease, subject to the Bankruptcy and Equity Exception;
- (b) no such Business Entity, nor, to Seller's Knowledge, any other party to any lease relating to Leased Real Property, is in default under such Lease, and no event has occurred or circumstance exists which, with the delivery of notice, the passage of time or both, would constitute such a breach or default under such Lease;
 - (c) there are no Proceedings with respect to any Lease;
- (d) no such Business Entity, nor, to Seller's Knowledge, any other party to a Lease, has subleased, licensed, or otherwise granted any Person the right to use or occupy such Leased Real Property or any portion thereof;
- (e) to Seller's Knowledge, no brokerage or leasing commissions or other compensation is or will be due or payable to any Person pursuant to any Contract entered into by a Business Entity with respect to or on account of any Lease or any extensions or renewals thereof;
- (f) no such Business Entity, nor, to Seller's Knowledge, any other party to a Lease, is entitled to any concession, allowance, rebate, or refund with respect to any rent payable or paid under any Lease;
 - (g) Sellers have made available to Buyer true, correct, and completed copies of each Lease (including all amendments thereto); and
 - (h) no Business Entity has assigned, transferred, conveyed, mortgaged, deeded in trust or encumbered any interest in any Lease.

- (i) To Seller's Knowledge, (i) there are no material structural, physical, or mechanical defects or other material adverse physical conditions affecting any of the Leased Real Property; (ii) all building systems and all improvements to, or which constitute a portion of, the Leased Real Property are structurally sound and in good operating condition and repair, ordinary wear and tear excepted; (iii) neither the whole nor any part of any of the Leased Real Property is subject to any pending or threatened Claim for condemnation or other taking by any Governmental Authority; (iv) the current uses of and existing structures located on the Leased Real Property in compliance in all material respects with all applicable zoning and other land use or occupancy requirements, and all covenants, conditions, and agreements affecting the Leased Real Property; (v) the Business Entities, to the extent required by any applicable Law, are in possession of all certificates of occupancy with respect to the Leased Real Property as is reasonably adequate for the current operation thereof; (vii) no construction, alteration, or other leasehold improvement work with respect to the Leased Real Property remains to be paid for or performed by any party to such Lease except for any such work required by the parties thereunder as part of the maintenance, repair, and replacement obligations, including with respect to casualty damage; (viii) there is no occurrence, condition, change in Law, or issue with any of the Leased Real Property in existence, or anticipated by a Business Entity that would in any way cause any such Leased Real Property (including Buyer's intended use thereof) to be in violation of any Law and/or thwart, prohibit, or otherwise materially interfere with Buyer's intended use thereof; and (ix) all material mechanical and other systems located on any of the Leased Real Property are in an operating condition good for the use to which the same are put by a Business Entity in the current operation of the Business, ordinary wear
- 3.11. <u>Transferred Tangible Assets</u>. Except as set forth on Schedule 3.11, each Business Entity has good title to, or a valid leasehold interest in or valid rights to use, all of the Transferred Tangible Assets, free and clear of any Encumbrances other than Permitted Encumbrances. All of the Transferred Tangible Assets are, in all material respects, in good operating condition (with due consideration for reasonable wear and tear and the age of each specific Transferred Tangible Assets). Notwithstanding anything to the contrary in this Section 3.11, this Section 3.11 does not cover Transferred Business Technology.

3.12. <u>Labor and Employment Matters.</u>

- (a) Except as set forth on <u>Schedule 3.12(a)</u>, the Business Entities are in compliance in all material respects with all applicable Laws relating to employment and employment practices, terms and conditions of employment and wages and hours, in each case, as they relate to the Business.
- (b) Except as set forth on Schedule 3.12(b), no Business Entity is a party to or otherwise bound by any collective bargaining agreement, contract or other agreement or understanding with a labor union or labor organization. No Business Entity is subject to any charge, demand, petition or representation proceeding seeking to compel, require or demand it to bargain with any labor union or labor organization, nor is there pending or, to Sellers' Knowledge, threatened in writing, any labor strike, dispute, walkout, work stoppage, slow down or lockout related to any Business Employees, in each case, as they are Related To the Business.

3.13. Material Contracts; Assumed Business Contracts.

- (a) Schedule 3.13(a) includes, as of the date of this Agreement, a true, complete and correct list of each of the following Contracts (y) to which any Business Subsidiary is a party or is bound or (z) to which either Seller is a party or is bound and which is Related To the Business or the Products or the Portal ((y) and (z) collectively, as listed or required to be listed below, the "Material Contracts"):
- (i) (y) any Contract with a Material Supplier or (z) any purchase and supply Contract with gross purchase volume relating to the Business in excess of \$100,000 in fiscal year 2023 or 2024;
- (ii) (y) any Contract with a Material Customer or (z) any customer Contracts with gross purchase volume relating to the Business in excess of \$100,000 in fiscal year 2023 or 2024;
 - (iii) any distributor, original equipment manufacturer, contract manufacturer, reseller, value added reseller, sales agency or referral
 - (iv) any material IP Contract (as well as any Contract for the development of the Portal);
 - (v) any Contract that relates to a partnership, joint venture, joint marketing, joint development or similar arrangement with any other

Person:

partner Contract;

- (vi) any Contract that restricts any Business Entity from, or following the Closing would or would purport to restrict Buyer or any of its Affiliates from (1) engaging in any aspect of their respective businesses, (2) participating or competing in any line of business, market or geographic area, (3) freely setting prices for its products, services or technologies (including most favored customer pricing provisions), or (4) soliciting potential employees, independent contractors or other suppliers or customers;
- (vii) any Contract under which (i) any Business Entity grants or is bound by any exclusive rights, noncompetition rights, rights of refusal, rights of first negotiation or similar rights, or (ii) following the Closing, any Business Entity would or would purport to have Buyer or any of its Affiliates grant or be bound by, any incremental and new exclusive rights, noncompetition rights, rights of refusal, rights of first negotiation or similar rights;
- (viii) any Contract that following the Closing would or would purport to require Buyer or any of its Affiliates to grant any incremental and new Intellectual Property right or license;
- (ix) any Contract relating to the settlement or other resolution of any dispute or threatened claim (including any agreement under which any employment-related claim is settled);

other Person;

(x) any Contract that involves the sharing of profits or revenue with other Persons or the payment of royalties or referral fees to any other Person;

(xi) any Contract that contains an earn-out, escrow or other similar contingent payment or obligations;

(xii) any Contract between any Business Entity and any Governmental Authority, university, college other educational institution or research center;

(xiii) any Contract for the advertisement or marketing of any Products;

(xiv) any Contract with any provider of telehealth or online medical services;

(xv) any Contract with a sole source supplier for any component related to the manufacture, testing, improvement, advertisement or sale of any Products;

(xvii) any Contract between any Business Entity and any pharmacy or other entity that prescribes access to the Product;

(xviii) each Assumed Business Contract;

(xviii) any Contract between or among any of the Business Entities or any of their Affiliates; and

(b) Each Material Contract, as of the date of this Agreement, is valid, binding and in full force and effect, enforceable in accordance with its terms, except as such may be limited by any applicable bankruptcy, reorganization, insolvency or similar Laws affecting the enforcement of creditor's rights generally or by general principles of equity (whether considered at law or in equity). Neither any Business Entity nor, to the Seller's Knowledge, any other party thereto, is in material default under any Material Contract and no written or oral notice of any claim of material default has been given to any Business Entity. No Business Entity has waived any of its material rights under any such Material Contract.

(xix) any Contract that is not covered by the sub-sections (i) through (xviii) above that is otherwise material to the Business.

(c) The Assumed Business Contracts constitute all Contracts to which Veru is a party that Relate To the Business. Except as set forth on Schedule 3.13, neither the Business nor any Assumed Business Contract has or includes an obligation to pay any Person any royalty, milestone payment or other amount based on or arising out of any sale or distribution of the Products.

3.14. Intellectual Property.

- (a) Schedule 3.14(a)(i) contains a list of all of Acquired Business Intellectual Property that are the subject of an application, certificate, filing, registration or other document issued, filed with, or recorded by any Governmental Authority ("Acquired Registered Business IPR"), and, in respect of each such item of Acquired Registered Business IPR, including the current owner, the jurisdiction in which each item has been registered or filed and the applicable registration, application or serial number or similar identifier, the filing date, and applicable issuance, registration or grant date. With respect to each item of Acquired Registered Business IPR: all necessary registration, maintenance and renewal fees and taxes as of the Closing Date have been paid, and all necessary documents and certificates as of the Closing Date have been filed with the relevant authorities in the United States or foreign jurisdictions, as the case may be, for the purposes of registering, maintaining and renewing, as applicable, and maintaining in full force and effect, such Intellectual Property Rights; each such item is currently in compliance with formal legal requirements (including payment of filing fees); and each such item is not subject to any late unpaid registration, maintenance or renewal fees. Each item of Acquired Registered Business IPR is subsisting, valid and enforceable. The Transferred Marks include all Marks owned by, purported to be owned by, or held in the name of Veru or any of its Affiliates (other than the Business Subsidiaries) that as of the Closing Date are Related To the Business other than any Marks set forth in Schedule 3.14(a).
- (b) There are no Patents owned by, purported to be owned by, licensed to, or held in the name of Veru or any of its Affiliates that as of the Closing Date cover, or would be infringed by, the making, having made, use, sale, offer for sale, importation, or other exploitation of the Purchased Assets or the operation of the Business.
- (c) All Acquired Business Intellectual Property is exclusively owned by the Sellers or the Business Subsidiaries, free and clear of all Encumbrances other than Permitted Encumbrances, and no Acquired Business Intellectual Property has been exclusively licensed or granted to any other Person.
- (d) The operation of the Business has not, and is not currently, violating, diluting, infringing or misappropriating any Intellectual Property Rights of any other Person, and, to the Knowledge of Sellers, will not violate, dilute, infringe or misappropriate any Intellectual Property Rights of any other Person when conducted in substantially the same manner after the Closing Date, and has not constituted, and does not constitute, unfair competition or trade practices under the Laws of any jurisdiction. No Business Entity has received written notice from any Person (1) claiming the operation of the Business, or any Acquired Business Technology, the Portal or Product, violates, dilutes, infringes or misappropriates any Intellectual Property Rights of any Person or constitutes unfair competition or trade practices under the Laws of any jurisdiction (nor does any Seller have Knowledge of any basis therefor) or (2) demanding or offering to license to any Business Entity any Intellectual Property Rights in connection with the Business. To the Knowledge of Sellers, no Person is violating, diluting, infringing or misappropriating any Acquired Business Intellectual Property. No Proceeding has been brought or threatened against any Person by any Business Entity alleging that a Person is violating, diluting, infringing, or misappropriating any Acquired Business Intellectual Property.
- (e) The Business Entities have taken commercially reasonable steps to maintain and protect the confidentiality of all Confidential Information (including trade secrets) included within the Acquired Business Intellectual Property. There has been no unauthorized use or disclosure of any such Confidential Information. No Business Entity, nor any other Person acting on behalf of any such entity, has disclosed, delivered or licensed to any escrow agent or other Person, agreed to disclose, deliver or license to any escrow agent or other Person of, any source code for or used in any Product or the Portal, and no Business Entity has any duty or obligation (whether present, contingent or otherwise) to deliver, license or make available such source code to any escrow agent or other Person, including as a result of the consummation of this Agreement.

(f) There are no forbearances to sue, consents, settlement agreements, judgments, orders or other legal or contractual obligations to which any
Business Entity or any of its Affiliates is a party or are otherwise bound that (i) restrict the rights of such Business Entity to use, transfer, license or enforce any of the
Acquired Business Intellectual Property; (ii) restrict the conduct of the Business in order to accommodate a third party's Intellectual Property; or (iii) grant any third party
any right with respect to any Acquired Business Intellectual Property.

- (g) The applicable Business Entity exclusively owns, or has adequate, valid, and enforceable rights to, all Intellectual Property Related To the operation of the Business as conducted immediately prior to Closing.
- (h) Neither the execution, delivery or performance of this Agreement nor the consummation of the Transaction Documents, nor the assignment of the Assumed Business Contracts, will, with or without notice or lapse of time, result in: (i) a forfeiture, termination, restriction, loss of, change to or acceleration of any rights under, or Encumbrance on, any Acquired Business Intellectual Property or any IP Contract; (ii) a breach of or default under any IP Contract; (iii) a new or increased payment or new or increased royalty or an obligation to offer any discount or be bound by any "most favored pricing" terms under any IP Contract; or (iv) the grant, assignment or transfer to any other Person of any license or other right or interest in, under, or with respect to, any Intellectual Property Rights owned or controlled by Buyer or any of its Affiliates as of Closing.
- (i) No government funding, facilities or resources of a Governmental Authority, university, college, or other educational institution or research center was used in the development of any Acquired Business Intellectual Property and no Governmental Authority, university, college, or other educational institution or research center has any claim or right in or to any Acquired Business Intellectual Property. No Person who was involved in, or who contributed to, the creation or development of any Acquired Business Intellectual Property, has performed services for any Governmental Authority, university, college, or other educational institution or research center during a period of time during which such Person was also performing services for any Business Entity.
- (j) All employees and independent contractors (including consultants) that have participated in the development or creation of any of the Acquired Business Intellectual Property have executed valid, enforceable, and appropriate assignment agreements, pursuant to which each such employee or independent contractor has assigned to any Business Entity all of its rights in and to all Intellectual Property that relate to the Business and that were conceived, created, authored, or developed, in whole or in part, by such employee or independent contractor. No past or present employee or independent contractor of any Business Entity has any ownership interest, license, permission, or other right in or to any of the Acquired Business Intellectual Property.

- (k) During the five (5) years immediately preceding the Closing, all Personal Information collected, disclosed, or otherwise processed by any Business Entity in connection with the Business has been, and is being, collected, processed and disclosed in material compliance with applicable laws relating to privacy, data protection, and information security, including, to the extent applicable, the Data Protection Laws. With respect to the Business, no Business Entity has, nor, any third party acting on behalf of any Business Entity has, received any: (A) written or oral notice or complaint alleging non-compliance with any Data Protection Law; (B) written or oral claim for compensation for loss or unauthorized collection, processing or disclosure of Personal Information; or (C) written or oral notification of an application for rectification, erasure or destruction of Personal Information that is still outstanding. Each of the mechanism through which any Business Entity collects Personal Information in connection with the Business, including internet websites, owned or operated by or on behalf of any Business Entity maintains a publicly posted privacy statement or policy that accurately describes the Business Entity's practices with respect to the collection, processing, use and disclosure of Personal Information in connection with the Business as required to comply in all material respects with all applicable Data Protection Laws. Each Business Entity's privacy policies with respect to the Business conform in all material respects and at all times have conformed in all material respects to its own published and internal privacy policies, terms of use and guidelines related to information privacy and security, including with respect to the collection, processing, use, disposal, disclosure, maintenance and transmission of Personal Information at the time such policies, terms of use or guidelines were in effect.
- (l) All Products, the Portal and all Acquired Business Technology are free of any "back door," "time bomb," "Trojan horse," "worm," "drop dead device," "virus" or other software routines or hardware components that permit unauthorized access or the unauthorized disablement, data corruption or erasure.
- (m) All Business Entities are in compliance with the terms and conditions of all licenses for the Open Source Materials used by such Business Entity in any way in connection with the Purchased Assets. No Business Entity has (i) incorporated Open Source Materials into, or combined Open Source Materials with, the Acquired Business Technology, the Portal or the Products or (iii) used Open Source Materials, in such a way that, with respect to (i), (ii) or (iii), creates or purports to create, obligations for a Business Entity with respect to any Acquired Business Technology or grant, or purport to grant, to any Person, any rights or immunities under any Acquired Business Intellectual Property (including using any Open Source Materials that require, as a condition of use, modification and/or distribution of such Open Source Materials that other software incorporated into, derived from or distributed with such Open Source Materials be (A) disclosed or distributed in source code form, (B) licensed for the purpose of making derivative works or (C) redistributable at no charge).

(n) Veru has and will have all rights required to grant the rights set forth in Section 5.11 on the terms set forth in that Section without breaching any Contract or violating or infringing any Intellectual Property of any of its Affiliates.

3.15. Environmental Matters.

- (a) The Business Entities are in compliance with all Environmental Laws applicable to the operation of the Business.
- (b) No Business Entity has transported, treated, stored, or disposed of any Hazardous Material, except in compliance with all applicable Environmental Laws.
- (c) No Business Entity has received any notice, demand, letter, claim or request for information, in each case in writing, alleging that such Business Entity may be in violation of any Environmental Law or alleging that such Business Entity is responsible for the investigation or remediation of a release of Hazardous Material at a property not owned or operated by a Business Entity.
- (d) No Business Entity is a potentially responsible party under any Environmental Law, or any analogous state, local, or foreign Law arising out of or relating to events occurring prior to the Closing Date.
- (e) No Business Entity is subject to any order, decree, or injunction with any Governmental Authority relating to Liability under any Environmental Law or relating to any Hazardous Material. Sellers have delivered to Buyer true and complete copies of all material environmental reports, studies, assessments, sampling data and other environmental information in its possession relating to the Business.
- 3.16. Brokers. Except as set forth on Schedule 3.16, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with this Agreement or the transactions contemplated hereby based upon any agreements, written, oral or otherwise made by or on behalf of Sellers.
- 3.17. Insurance. Schedule 3.17 sets forth a complete and accurate list and description of all policies of insurance presently in effect with respect to the Purchased Assets or the Business (the "Insurance Policies"). All such Insurance Policies are valid, outstanding and enforceable and in full force and effect. No written notice of cancellation or termination has been received by any Business Entity with respect to any of the Insurance Policies. All premiums currently payable or previously due on the Insurance Policies have been paid and no Business Entity is in default under, or breach of, any such Insurance Policy. The Business Entities (in the case of a Seller, with respect to the Business or the Purchased Assets) are currently insured by insurers unaffiliated with the Business Entities with respect to its properties, assets, and operation of the Business or the Purchased Assets) or by which it or any of its assets is bound that relate to the Business. True, correct, and complete copies of all the Insurance Policies have been made available to Buyer. Except as set forth on Schedule 3.17, there are no claims by the Business Entities pending under any Insurance Policy and during the three (3)-year period prior to the date of this Agreement, no claim by the Business Entities under any Insurance Policy has been denied coverage or resulted in the issuance of a reservation of rights letter by the insurer under such policy. Veru has made available to Buyer loss runs for each such Insurance Policy for the three (3)-year period ending on November 30, 2024.

3.18. Legal Compliance.

- (a) Except as set forth on Schedule 3.18, each Business Entity is, and during the five (5) years immediately preceding the Closing has been at all times, in compliance in all material respects with all Legal Requirements and Orders Related To the Business or the Purchased Assets. Except for inquiries set forth on Schedule 3.18, no Proceeding has been filed or commenced during the five (5) years immediately preceding the Closing by any Governmental Authority against any Business Entity alleging a violation of any such applicable Legal Requirements or Orders, nor has there been a reasonable basis for any such Proceeding. With respect to the Business and the Purchased Assets, no Business Entity has any Liability for past or continuing violations of any Legal Requirement or Order.
- (b) Each Business Entity is, and during the five (5) years immediately preceding the Closing have been at all times, in compliance with all Anti-Corruption and Anti-Bribery Laws. No director, officer, employee or, to Sellers' Knowledge, any other Person acting for or on behalf of any Business Entity has violated any Anti-Corruption and Anti-Bribery Laws during the five (5) years immediately preceding the Closing in connection with the Business or the Purchased Assets. No Business Entity has received any notice from any Governmental Authority during the five (5) years immediately preceding the Closing alleging any violations, litigation or investigations by any Business Entity related to any violation of any Anti-Corruption and Anti-Bribery Laws in connection with the Business or the Purchased Assets.
- (i) For the previous five (5) years, none of the Business Entities (in the case of a Seller, with respect to the Business or the Purchased Assets), nor any of their respective officers, directors, employees, or consultants, Affiliates or other Persons associated with or acting on its behalf have in connection with the Business, (1) directly or indirectly, provided, attempted to provide, authorized, or used any corporate funds for unlawful contributions, gifts, services of value, entertainment or other unlawful expenses relating to political activity; (2) made, promised to make, offered, attempted, or authorized any unlawful payment or given, offered, attempted, or promised to give, anything of value to foreign or domestic governmental officials or employees in any manner that is unlawful; or (3) made, promised to make, offered, attempted, authorized any unlawful bribe, rebate, payoff, influence or extortion payment, kickback or other similar unlawful payment or provision to any person.
- (ii) Sellers have no Knowledge of any whistleblower complaint, or other informal or formal allegation related to the potential or actual noncompliance of their respective officers, directors, agents, employees, affiliates or other Person associated with or acting on its behalf with Anti-Corruption and Anti-Bribery Laws, in each case in connection with the Business.

(iii) There are no Claims or conditions or circumstances pertaining to the Business that could reasonably be expected to give rise to
any future claims, charges, investigations (including internal investigation), violations, settlements, civil or criminal actions, lawsuits, or other court actions under any Anti-
Corruption and Anti-Bribery Laws.

- (iv) Each Business Entity (in the case of a Seller, with respect to the Business or the Purchased Assets) has established and maintains a compliance program designed to provide reasonable assurance regarding compliance with the Anti-Corruption and Anti-Bribery Laws.
- (v) The Business Entities do not and have not in the past five (5) years engaged in the business of remitting or transmitting funds or engaged in any act that would make it subject to the regulatory compliance regimes of the Anti-Money Laundering Laws. The Business Entities have not violated or taken any act in furtherance of violating the Anti-Money Laundering Laws.

(c) Without limiting the generality of Section 3.18(a) and Section 3.18(b):

- (i) The Business and each Business Entity (in the case of a Seller, with respect to the Business or the Purchased Assets) are being, and has been conducted in compliance in all material respects with all applicable Health Care Laws. No Business Entity has received any written notice, charge, assertion or other communication from the FDA or any other Governmental Authority alleging any violation of any Health Care Law in respect of the Business. Schedule 3.18(c) sets forth a list of all of the material Permits issued to or held by any Business Entity Related To the Business or any Product or Purchased Assets, including all pending applications submitted to and/or regulatory authorizations from the FDA or any other regulatory authority or Governmental Authority Related To the Business, Product, or any Purchased Assets. Each such Permit is in full force and effect. Each Business Entity is in material compliance with the terms of each such Permit applicable to it, and no suspension, material modification, non-renewal, or cancellation of such Permit is threatened. Each such Permit will continue in full force and effect immediately following the Closing. Sellers have provided to the Buyer all such Permits.
- (ii) The Product is in material compliance with the conditions of its approval by the FDA and, in its current form, may be marketed and sold to consumers in the United States. The operation of each Business Entity (in the case of a Seller, with respect to the Business, any Product, or the Purchased Assets), including any manufacture, import, export, testing, clinical or nonclinical investigation, development, processing, packaging, labeling, storage, marketing, certification, registration, listing, advertising, reporting, record keeping, sale and distribution of the Product, and to the Seller's Knowledge, all such activities or processes relating to the Business that are outsourced to third parties, is and has been in compliance in all material respects with all applicable Laws (including, without limitation, any Health Care Law), in all markets and jurisdictions in which the Business is conducted.
- (iii) As of the date of this Agreement, no Business Entity nor any Person acting on behalf of such Business Entity is conducting any clinical trials on the Product. No Business Entity nor any Person acting on behalf of such Business Entity has received any written notice that the FDA or any other Governmental Authority or institutional review board has initiated, or to the Knowledge of Sellers, threatened to initiate, any clinical hold or other action to suspend any planned clinical trial or otherwise restrict any research or study Related To the Product.

(iv) Sellers have delivered or made available upon request to Buyer all of the following which bear in any material way on the Business' compliance with Health Care Laws: (A) all correspondence and meeting minutes received from or sent to the FDA and any other Governmental Authority, (B) written reports of all phone conversations, visits or other contact with the FDA and any other similar Governmental Authority, including any and all notices of inspectional observations, and (C) any other documents received by such Business Entity from the FDA or any similar Governmental Authority.

(v) No Business Entity nor any officer, employee or agent of any Business Entity has made an untrue statement of a material fact or fraudulent statement to the FDA or any similar Governmental Authority, failed to disclose a material fact required to be disclosed to the FDA or any similar Governmental Authority, or committed any act, made any statement, or failed to make any statement, that would reasonably be expected to provide a basis for the FDA to invoke its policy respecting "Fraud, Untrue Statements of Material Fact, Bribery, and Illegal Gratuities," set forth in 56 Fed. Reg. 46191 (September 10, 1991). No Business Entity nor any officer, employee or agent of any Business Entity has been convicted of any crime or engaged in any conduct that would reasonably be expected to result in or that has resulted in (A) debarment under 21 U.S.C. 335a or any similar applicable Law or (B) exclusion from participating in the federal health care programs under 1128 of the Social Security Act or any similar applicable Law. No Business Entity (in the case of a Seller, with respect to the Business, any Product, or the Purchased Assets) nor any of its Representatives (i) is a party to a corporate integrity agreement with any Governmental Authority, (ii) has entered into a settlement agreement with a Governmental Authority, (iii) is or has been a defendant in any qui tam or federal civil False Claims Act litigation or (iv) has been served with or received any written search warrant, subpoena (other than those related to actions against third parties), civil investigative demand or Contact Letter from a Governmental Authority. For purposes of this Section, a "Contact Letter" means a written communication from a Governmental Authority notifying a Business Entity or any of its Representatives of a potential violation under a Health Care Law in respect of the Business or any Product that allows the Business Entity or any of its Representatives an opportunity to respond prior to the Governmental Authority taking further action.

(vi) Each Business Entity has all governmental authorizations from the FDA and any other comparable Governmental Authority or quasi-governmental body such as a notified body required to conduct the operations of the Business. Each such governmental authorization is valid and subsisting in full force and effect. Neither the FDA nor any comparable Governmental Authority or quasi-governmental body has terminated, limited, suspended or revoked any governmental authorizations or changed the marketing classification or materially changed the labeling of any of the Products, and to the Seller's Knowledge, neither the FDA nor any comparable Governmental Authority or quasi-governmental body is considering any such action. Each Business Entity has fulfilled and performed its obligations under each governmental authorization in all material respects, and to the Seller's Knowledge, no event has occurred or condition or state of facts exists which would constitute a material breach or material default under or would cause revocation, termination, suspension or any limitation of any such governmental authorization. To the Seller's Knowledge, any third party that is a manufacturer or contractor for the Business Entity is in material compliance with all governmental authorizations from the FDA or comparable Governmental Authority insofar as they pertain to the manufacture of the Products or any components, parts or accessories of the Products. All products developed, tested, investigated, manufactured, processed, held, stored, packaged, packed, labeled, imported, exported, distributed, marketed or sold, as applicable, by or on behalf of any Business Entity in respect of the Business or any Product that are subject to the jurisdiction of the FDA or comparable Governmental Authority, have been and are being developed, tested, investigated, manufactured, processed, held, stored, packaged, packed, labeled, imported, exported, distributed, marketed and sold in material compliance with all Health Care Laws, including the U.S. Federa

(vii) Within the last five (5) years prior to the date of this Agreement, (i) there have been no recalls, field corrections, removals, field notifications, safety alerts, FDA letters to health care providers, suspensions, seizures, detentions, withdrawals, or discontinuations of any of the Products or Related To the Business (and none are threatened or pending), and (ii) no report of any problems, malfunctions, serious adverse events, or defects involving any Products or Related To the Business has been filed or is required to have been filed with any Governmental Authority. There are no pending and, within the last five (5) years prior to the date of this Agreement, there have not been any actions, claims or threats thereof related to product liability involving any Product, and no such actions, claims or threats have been settled, adjudicated or otherwise disposed of.

(viii) Except as set forth on Schedule 3.18(c)(viii), as of the date of this Agreement, there are no citations, decisions, adjudications or written statements by any Governmental Authority or consent decrees stating that any Product is defective or unsafe or fails to meet any standards or requirements promulgated by any such Governmental Authority. To the Seller's Knowledge, there is no fact or condition currently in existence related to any Product that would impose upon any Business Entity a duty to recall, withdraw, or replace any Product, change the marketing classification or materially change the labeling of any Product, or terminate or suspend the marketing of any Product, or result in material liability for returns or other product liability claims with respect to the Products.

(ix) All arrangements involving the offer, payment or provision by any Business Entity in respect of the Business or Product, to any health care professional, institution, organization, pharmacy, or other health care provider, or any Representative thereof, are in compliance with Health Care Laws.

(d) The Business Entities (in the case of a Seller, with respect to the Business, any Product, or the Purchased Assets) have, at all times in the past five years, conducted its export and import transactions in accordance with all applicable Export and Import Control Laws in all material respects. Without limiting the foregoing, in the past five years, with respect to the Business or the Purchased Assets: (i) each Business Entity has obtained and is in compliance with the terms of all required Export and Import Approvals; (ii) no Business Entity has engaged in nor conducted business, directly or indirectly in a country subject to an embargo by the United States in violation of Export and Import Control Laws and has not allowed access to its services, technologies, software or products to a person located in a country subject to an embargo by the United States in violation of Export and Import Control Laws; (iii) there are no pending or, to the Knowledge of the Seller, threatened claims, charges, investigations, violations, settlements, civil or criminal enforcement actions, lawsuits, or other actions against any Business Entity with respect to any Export and Import Control Laws; (iv) to the Knowledge of the Seller, there are no actions, conditions or circumstances pertaining to the Business Entities' export or import transactions that may give rise to any future claims, charges, investigations, violations, settlements, civil or criminal actions, lawsuits, or other actions under the Export and Import Control Laws; and (v) no approval from a Governmental Authority is required for the transfer of Export and Import Approvals to Buyer are required, or such approvals can be obtained expeditiously without material cost.

(e) None of the Business Entities (in the case of a Seller, with respect to the Business, any Product or the Purchased Assets) have made a submission under FDA Form 2253.

3.19. Product Matters.

- (a) No Claim has been filed, threatened in writing or commenced during the five (5) years immediately preceding the Closing against any Business Entity alleging that any Product is defective or unsafe or fails to meet any standards promulgated by any Governmental Authority or any Business Entity. During the five (5) years immediately preceding the Closing, none of the Products has been the subject of any replacement, modification or recall campaign by any Business Entity and no facts or conditions related to any Product exist which would reasonably be expected to result in such a campaign.
- (b) Each Product marketed or sold by any Business Entity during the five (5) years immediately preceding the Closing has been in conformity in all material respects with internal specifications, good manufacturing practices, industry standards, contractual agreements, and standard operating procedures.

3.20. Title and Sufficiency of Assets.

(a) Subject to Section 3.20(c), Veru has good and valid title to the Purchased Assets, free and clear of all Encumbrances (other than Permitted Encumbrances). Except as set forth on Schedule 3.20(a), Veru has the complete and unrestricted power and unqualified right to sell, transfer, assign, convey and deliver the Purchased Assets to Buyer pursuant to this Agreement. Subject to Section 3.20(c), and taking into account all Transaction Documents, no Affiliate of Veru, other than the Business Subsidiaries, owns, controls or possesses any assets of any type that are Related To the Business other than (i) the Shares held by Limited prior to the transfer and assignment of the same as set forth in this Agreement and (ii) the Marks covered by Section 5.11.

- (b) Subject to Section 3.20(c), on the Closing Date (assuming the notices, declarations, filings, consents and approvals set forth in Section 3.04 and the Consent Schedule are made or obtained), the Purchased Assets and the assets of the Business Subsidiaries, taking into account all Transaction Documents, constitute all of the assets necessary for Buyer to conduct the Business immediately following the Closing Date in the same manner as the Business is being conducted as of immediately prior to the Closing Date.
- (c) Nothing in this Section 3.20 shall be deemed to constitute a representation or warranty regarding (a) the adequacy of the amounts of Cash and Cash Equivalents or Net Working Capital (or the availability of the same) or (b) any infringement or misappropriation of any Intellectual Property of any other Person beyond those expressly set forth in Section 3.14.
- 3.21. Certain Transactions. No Business Entity nor any of its controlled Affiliates, nor, to Sellers' Knowledge, any officer, director or employee of any Business Entity or any of its controlled Affiliates, nor, to Sellers' Knowledge, any family member of any such officer, director or employee: (a) has any ownership or interest in any of the Purchased Assets, (b) participates in, engages in or owns any business that (i) has an ongoing business relationship with Business, or (ii) competes with or is engaged in the same business as the Business, (c) is a party to any Assumed Business Contract, or (d) owes any amount to, or is owed any amount by, any Business Entity with respect of the Business.

3.22. Customers and Suppliers.

- (a) Schedule 3.22(a) sets forth a list of the names of the ten largest revenue generating Contracts of the Business (measured by dollar volume of sales) (the "Material Customers"), in each case during the 12 months ended September 30, 2024 and the 12 months ended September 30, 2023.
- (b) Schedule 3.22(b) sets forth a list of the names of the ten largest Contracts of the Business based on total amounts spent (measured by dollar amounts spent) (the "Material Suppliers"), in each case during the 12 months ended September 30, 2024 and the 12 months ended September 30, 2023.
- (c) Except as set forth in <u>Schedule 3.22(c)</u>, no Business Entity is engaged in any dispute related to the Business with any Material Customer or Material Supplier, and no Material Customer or Material Supplier has provided written notice that it intends to terminate or materially and adversely alter its relationship with the Business.
- (d) Since October 1, 2023, no Material Customer or Material Supplier has cancelled or otherwise adversely modified in any material respect its relationship with a Business Entity and no fact, condition, or event exists that would reasonably be expected to adversely affect in any material respect the relationship of a Business Entity with any Material Customer or Material Supplier. No Business Entity has received any written notice that any of the Material Customers or Material Suppliers has ceased, or intends to cease after the Closing, to use the goods or services of the Business Entities, or to supply goods and services to the Business Entities, respectively, or has modified, amended, or reduced, or intends to modify, amend, or reduce, its business relationship with the Business Entities in a manner, that is, or is reasonably likely to be, adverse to the Business Entities.

3.23. Compensation of and Contracts with Employees and Independent Contractors.

- (a) Schedule 3.23(a) sets forth a true, correct, and complete list of all Business Employees as of the date hereof, and sets forth for each such Person the following: (i) name; (ii) title or position (including whether full or part-time); (iii) with respect to Business Employees in the United States, classification for overtime pay purposes under federal and applicable state Law (exempt or non-exempt); (iv) hire date; (v) current annual base salary or hourly compensation rate; (vi) commission, bonus, or other incentive-based compensation; (vii) the rate and amount of such compensation paid to each such employee for calendar year 2024; (viii) whether the employee is on active or inactive status and, if inactive, the expected date of return; (ix) firinge benefits, other than benefits generally made available to employees; and (x) whether the employee is authorized to work as a result of a visa and, if so, the type of visa and expiration date of the work authorization. There have been no material changes or changes outside the ordinary course of business in any compensation since October 1, 2023, in each case including bonuses and other compensation and fringe benefits, and each Business Entity has timely paid all such compensation when due and owing, including bonuses and other compensation and fringe benefits do not have any unpaid pension, bonus, commission, or other similar payment obligations payable by any Business Entity to Business Employees, other than base salary. No Business Entity has outstanding loans or advances to Business Employees. All Business Employees classified as exempt. All Business Employees classified as non-exempt under the Fair Labor Standards Act and any applicable state and local wage and hour Laws at all times in the past three (3) years have been properly classified as exempt. All Business Employees classified as non-exempt under the Fair Labor Standards Act and any applicable state and local wage and hour Laws at all times in the past three (3) years have been fully and properly pai
- (b) Each individual independent contractor currently performing services for a Business Entity as an independent contractor who spends a majority of his or her working time providing services for the benefit of a Business Entity is listed on <u>Schedule 3.23(b)</u> (the "<u>Applicable Contractors</u>"). <u>Schedule 3.23(b)</u> sets forth for each such independent contractor the following: (i) name; (ii) the start date and anticipated completion date for the performance of services; (iii) the type of services to be provided; and (iv) the hourly or other rate of pay provided to the independent contractor.
- (c) Each Business Employee and Applicable Contractor is an "at-will" employee or an independent contractor, as applicable, whose employment or engagement, respectively, may be terminated at any time without advance notice by or liability to a Business Entity.
- (d) All salaries, wages, commissions, bonuses, vacation pay, withholdings, remittances, and other Liabilities related to the employment or engagement of Business Employees and Applicable Contractors that are accrued or due to be paid on or before the Closing Date have been fully paid as of the Effective Time and in accordance with applicable Law.

- (e) The Business Entities (in the case of a Seller, with respect to the Business) have not, within the past five (5) years, received any "cease and desist" letter or similar communication alleging that any of their respective employees or independent contractors is, and to Seller's Knowledge, no employee or independent contractor is, performing any job duties or engaging in other activities on behalf of such Business Entity that would violate any employment, non-competition, non-solicitation, non-disclosure, or other similar agreement between such individual and any former employer or any Law.
- 3.24. No Additional Representations. Sellers acknowledge that Buyer has not made any representation or warranty, express or implied, regarding Buyer, except for the representations and warranties of Buyer expressly set forth in Section 4 of this Agreement. Sellers further agree that, except with respect to the representations and warranties of Buyer expressly set forth in Section 4 of this Agreement, Buyer shall not have or be subject to any liability to Sellers resulting from the distribution to a Seller, or a Seller's use of, any information, document or material provided to Sellers or made available to Sellers, as Sellers acknowledge that no representations or warranties are made with respect to any such information, document, material or statement. Notwithstanding the foregoing, the foregoing provisions shall not limit or abrogate (i) the representations and warranties expressly set forth in Section 4 of this Agreement or a Seller's right to rely thereon or (ii) Buyer's liability to Sellers in the event of Fraud by or on behalf of the Buyer.

EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES OF SELLERS EXPRESSLY SET FORTH IN THIS<u>SECTION 3</u>, SELLERS MAKE NO REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, AT LAW OR IN EQUITY, IN RESPECT OF THE SELLERS, THE BUSINESS, THE BUSINESS ENTITIES, THE STOCK, THE PURCHASED ASSETS OR THE ASSUMED LIABILITIES.

- 4. <u>Representations and Warranties of Buyer</u>. Buyer hereby makes to Sellers the representations and warranties contained in this <u>Section 4</u> as of the date hereof (unless another date is so indicated, then only as of such date).
- 4.01. Organization. Buyer is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware. Buyer has all necessary limited liability company power and authority to carry on its business as now conducted and to own or lease the properties and assets it now owns or leases.

4.02. Authorization of Agreement; No Conflict.

(a) Buyer has all requisite limited liability company power and authority to execute and deliver this Agreement and the other Transaction Documents to be executed and delivered by Buyer and to consummate the transactions provided for herein and therein. The execution and delivery by Buyer of this Agreement and the other Transaction Documents to be executed and delivered by Buyer and the performance by Buyer of the obligations to be performed hereunder and thereunder have been duly authorized by Buyer by all requisite corporate action. This Agreement is, and each other Transaction Documents to be executed by Buyer will be when so executed, a valid and binding obligation of Buyer, enforceable in accordance with its terms, subject to the Bankruptcy and Equity Exception.

- (b) The execution and delivery of this Agreement and the other Transaction Documents to be executed and delivered by Buyer and the consummation of the transactions contemplated hereby and thereby do not and will not, with or without the giving of notice or the passage of time, conflict with, result in or constitute a breach, default, right to accelerate or loss of rights under, or result in the creation of any Encumbrance pursuant to, the terms or conditions of Buyer's Organizational Documents or any Legal Requirement or any mortgage, lease, franchise, license, permit, contract, agreement and/or instrument to which Buyer is a party or by which Buyer is bound.
- 4.03. <u>Litigation</u>. There is no Proceeding pending or, to Buyer's knowledge, threatened against Buyer, except as would not reasonably be expected to materially and adversely affect or restrict Buyer's ability to consummate the transactions contemplated by this Agreement.
- 4.04. <u>Brokerage</u>. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with this Agreement or the transactions contemplated hereby based upon any agreements, written, oral or otherwise made by or on behalf of Buyer.
- 4.05. <u>Financial Capability; Solvency</u>. Buyer has sufficient immediately available funds in cash to pay the Estimated Closing Consideration described in this Agreement and to pay its related fees and expenses. Buyer is not insolvent, nor will Buyer be rendered insolvent by any of the transactions contemplated herein. As used in this Section, "insolvent" means that the sum of the debts and other probable Liabilities of Buyer exceeds the present fair saleable value of the assets of Buyer.
- 4.06. <u>Investment Intent</u>. Buyer is acquiring the Stock for its own account and not with a view to distribution of the Stock within the meaning of any applicable securities Laws.
- 4.07. No Additional Representations. Buyer acknowledges that Sellers have not made any representation or warranty, express or implied, regarding Sellers, the Business, the Business Entities, the Stock, the Purchased Assets or the Assumed Liabilities, except for the representations and warranties of the Sellers expressly set forth in Section 3 of this Agreement. Buyer further agrees that, except with respect to the representations and warranties of the Sellers expressly set forth in Section 3 of this Agreement, the Sellers shall not have or be subject to any liability to Buyer or any other Buyer Indemnified Party resulting from the distribution to Buyer, or Buyer's use of, any information, document or material provided to Buyer or made available to Buyer in any "data room", management presentations or any other form in expectation of the transactions contemplated by this Agreement, including any estimates, forecasts, plans, financial projections or statements concerning the backlog or pipeline of the Business, as Buyer acknowledges that no representations or warranties are made with respect to any such information, document, material or statement. Buyer further acknowledges that there are uncertainties inherent in attempting to make any such estimates, forecasts, projections or statements, that Buyer is familiar with such uncertainties, and that Buyer is fully responsible for making its own evaluation of the adequacy and accuracy of any such estimates, forecasts, projections or statements. Buyer further acknowledges and agrees that (a) it has conducted such investigations of the Sellers and the Business as it deems necessary in connection with the execution of this Agreement and the consummation of the transactions contemplated hereby, (b) it and its representatives have been permitted access to the records, facilities, equipment, Tax Returns, Contracts and other properties and assets of the Sellers which they have desired and requested to see and/or review, and (c) it and its representatives have had the opportunity to meet with representatives of the Sellers to discuss the Business, the Business Entities, the Stock, the Purchased Assets and the Assumed Liabilities. Notwithstanding the foregoing, the foregoing provisions shall not limit or abrogate (i) the representations and warranties expressly set forth in Section 3 (as modified by the Schedules) of this Agreement or Buyer's right to rely thereon or (ii) the Sellers' liability to Buyer or any other Buyer Indemnified Party in the event of any Fraud by or on behalf of the Sellers.

EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES OF BUYER EXPRESSLY SET FORTH IN THIS<u>SECTION 4</u>, BUYER MAKES NO REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, AT LAW OR IN EQUITY, IN RESPECT OF BUYER.

Certain Agreements.

5.01. Restrictive Covenants.

(a) Nondisclosure. For a period of five (5) years following the Closing Date, except as otherwise expressly provided in this Agreement, each Seller will hold in confidence and not disclose, release or use (except as may be necessary to enforce its rights as described in clause (iii) below, in connection with the performance of its obligations under this Agreement or any of the other Transaction Documents or the preparation of any Tax Returns required to be filed by it pursuant to this Agreement) without the prior written consent of Buyer, any and all Confidential Information related to the Business; provided, that each Seller may disclose, or may permit disclosure of, such information: (i) to its representatives who have a need to know such information for a purpose not prohibited by this Section 5.01(a) and are informed of their obligation to hold such information confidential to the same extent as is applicable to such Seller and in respect of whose failure to comply with such obligations such Seller will be responsible, (ii) if such Seller or its representatives are required to disclose any such information pursuant to applicable Law or stock exchange rules or (iii) in connection with the enforcement of any right or remedy or the defense of any claim relating to this Agreement or any of the other Transaction Documents or the transactions contemplated hereby and thereby. Notwithstanding anything to the contrary in the foregoing, in the event that any demand or request for disclosure of such Confidential Information is made pursuant to clause (ii) above, Sellers will to the extent permissible promptly notify Buyer of the existence of such request or demand and will provide Buyer a reasonable opportunity to seek an appropriate protective order or other remedy (and cooperate with Buyer with respect thereto, at Buyer's sole cost and expense), and in the event such protective order or other remedy is not obtained, Sellers may disclose such Confidential Information without Liability hereunder, but will furnish only that portion of such Confidential Information that Sellers are advised by legal counsel (which may be in-house counsel) it is legally required to disclose and will, to the extent requested by Buyer, exercise reasonable efforts, at Buyer's sole cost and expense, to preserve the confidentiality of such information; provided that, with respect to Confidential Information that constitutes a trade secret under applicable Law, Sellers' confidentiality obligations with respect to such Confidential Information pursuant to this Section 5.01(a) shall survive indefinitely. Notwithstanding anything to the contrary herein, this Section 5.01(a) will not apply to information: (A) to the extent not relating to the Business or the Purchased Assets, (B) that is or becomes generally available to the public other than as a result of disclosure by a Seller or an Affiliate or representative of a Seller in breach of any confidentiality obligation, (C) that becomes available to a Seller or an Affiliate of a Seller after the Closing Date on a non-confidential basis from a source other than Buyer or an Affiliate or representative of Buyer (provided, that such source is not known by such Seller to be bound by any obligation of confidentiality to Buyer) or (D) that is independently developed by a Seller or any Affiliate of a Seller following the Closing without reference to or reliance upon the Confidential Information and only by Persons who have never accessed such Confidential Information.

(b) Noncompetition; Nonsolicitation. For a period of five (5) years following the Closing Date, each Seller will not, and will cause its controlled Affiliates to not, directly or indirectly, (i) engage in a Competitive Activity anywhere in the world, (ii) control any entity that engages in a Competitive Activity, or (iii) knowingly and willfully cause, induce or encourage any material customer, supplier, contract manufacturer or licensor of the Business as of the Effective Time, or any other Person who has a material business relationship with the Business as of the Effective Time, to terminate or modify such material customer, supplier, licensor or other material business relationship with the Business; provided, however, that neither (x) the ownership of not more than five percent (5%) of the outstanding shares of any publicly held company that engages in an activity otherwise prohibited by this Section 5.01(b) nor (y) any services contemplated by the Transition Services Agreement shall constitute a breach of this Section 5.01(b). Further, if a Seller acquires a business that, while primarily engaged in other activities, also engages in a Competitive Activity (a "Prohibited Business"), and such Seller (x) promptly winds down the Prohibited Business, or (y) divests the Prohibited Business within sixty (60) days following the closing date of the acquisition of the Prohibited Business, then such actions shall not constitute a breach of this Section 5.01(b). Contemporaneously with and in contemplation of the execution and delivery of this Agreement, and as a condition and inducement to Buyer's willingness to enter into this Agreement, Mitchell Steiner and Harry Fisch shall each enter into non-compete agreements with Buyer, to be effective upon the Closing.

(c) Reasonableness; Equitable Relief. Each Seller acknowledges and agrees that the restrictions and covenants contained in this Section 5.01 are reasonably necessary to protect the legitimate business interests of Buyer and constitute a material inducement to Buyer to enter into this Agreement and to consummate the transactions contemplated hereby. Each Seller specifically acknowledges and agrees that Buyer will be entitled to seek injunctive relief for a breach of any restriction contained in this Section 5.01. If at any time a court or arbitrator's award holds that the restrictions in this Section 5.01 are unreasonable under circumstances then existing, the parties agree that the maximum period, scope or geographical area reasonable under such circumstances shall be substituted for the stated period, scope or area. The parties agree that any breach of the provisions contained in this Section 5.01 will result in serious and irreparable injury and therefore money damages would not be an adequate remedy for any such breach. Therefore, in the event of a breach or threatened breach of any provisions of this Section 5.01 that is continuing, Buyer, its successors and assigns and any third-party beneficiary to this Agreement, in addition to other rights and remedies existing in their favor, shall be entitled to specific performance or injunctive or other relief in order to enforce, or prevent any violations of, the provisions hereof (without posting a bond or other security). In addition, in the event of a breach or violation by a Seller or any of their controlled Affiliates of this Section 5.01, the applicable non-compete term shall be tolled until such breach or violation has been duly cured

- 5.02. Publicity. No public announcement related to this Agreement or the transactions contemplated herein will be issued without the joint approval of Sellers and Buyer, which approval shall not be unreasonably withheld, conditioned or delayed, except in any public disclosure which Sellers, in their good faith judgment, believe is required by applicable Law or by any stock exchange on which their securities are listed. If a Seller, in its good faith judgment, believes such disclosure is required, such Seller shall use its commercially reasonable efforts to consult with Buyer, and to consider in good faith any revisions proposed by Buyer, as applicable, prior to making (or prior to any Affiliate making) such disclosure; provided that Buyer shall deemed to have consented to any such disclosure to the extent such Seller has provided Buyer with a draft thereof and Buyer has not suggested any revisions within five (5) Business Days of Buyer's receipt of such draft. Notwithstanding the foregoing, without the approval of the other Party, Buyer and Sellers and their respective Affiliates may, following the Closing Date and subject to the other terms and conditions of this Agreement, (a) make internal announcements to their and their Affiliates' respective employees and Affiliates, (b) communicate with Governmental Authorities regarding this Agreement and the transactions contemplated hereby, and (c) make the public announcements and engage in public communications regarding this Agreement and their affiliates (c), solely to the extent such announcements or communications are consistent with a communications plan agreed upon by Sellers and Buyer or their respective prior public communications made in compliance with this Section 5.02). The press release announcing the execution and delivery of this Agreement shall not be issued prior to the approval of each of Sellers and Buyer. Veru shall file a current report on Form 8-K with the Securities and Exchange Commission attaching the announcement press release and a copy of
- 5.03. Conflicts and Privilege. It is acknowledged by each of the parties hereto that the Business Subsidiaries may retain counsel to them (Counsel') to act as their counsel in connection with the transactions contemplated hereby. Buyer hereby agrees that, in the event that a dispute arises after the Closing between Buyer and Sellers, Counsel may represent Sellers in such dispute even though the interests of Sellers may be directly adverse to Buyer or the Business, and even though Counsel may have represented the Business Subsidiaries in a matter substantially related to such dispute, or may be handling ongoing matters for Buyer or any Business Subsidiary. Buyer further agrees that, as to all communications among Counsel, Sellers and their respective Subsidiaries that relate in any way to the transactions contemplated by this Agreement, the attorney client privilege and the expectation of client confidence belongs to Sellers and may be controlled by Sellers and shall not pass to or be claimed by Buyer or any Business Subsidiary. Notwithstanding the foregoing, in the event that a dispute arises between Buyer or any Business Subsidiary and a third party other than a party to this Agreement after the Closing, the Business Subsidiaries may assert the attorney client privilege to prevent disclosure of confidential communications by Counsel to such third party; provided, however, that none of the Business Subsidiaries may waive such privilege without the prior written consent of Sellers.

5.04. Employee Benefit Arrangements and Other Employee Matters.

(a) General. Buyer shall ensure that all Business Employees who are employed by the Business Subsidiaries immediately preceding the Closing, including those on vacation, family leave, short-term disability or other approved leave of absence, will remain employed in a comparable position on and immediately after the Closing Date, and Buyer shall, or shall cause a Business Subsidiary or other Subsidiary of Buyer to, offer employment in a comparable position on the Closing Date, to all Business Employees identified on Schedule 5.04 hereto, including any such Business Employees who are absent due to vacation, family leave, shortterm disability or other approved leave of absence, in each case at not less than the same base rate of pay and with fringe benefits that are substantially comparable in the aggregate to the fringe benefits paid or provided to each Business Employee immediately prior to the date of this Agreement. Business Employees shall be credited for their length of service with, or credited by, the Business Subsidiaries or the Business for all purposes under any benefit plan or program or fringe benefit made available to Business Employees by Buyer or any of its Affiliates after the Closing ("Buyer Benefit Programs"); provided, however, that such service shall not be so credited (i) for purposes of calculating accrued benefits under a defined benefit pension plan or (ii) to the extent that crediting such service would result in the duplication of either benefits or accruals to a Business Employee. Any preexisting condition clause in any Buyer Benefit Programs (including medical, dental and disability coverage) in which Business Employees are eligible to participate after the Closing shall be waived for the Business Employees. Buyer shall credit the Business Employees with any amounts paid under the Business Benefit Plans on or prior to the Closing toward satisfaction of the applicable deductible amounts and copayment obligations under the corresponding Buyer Benefit Programs for the plan year in which the Business Employees become eligible to participate in Buyer Benefit Programs including, without limitation, by providing credit, as applicable, towards any Buyer Benefit Program that provides long-term disability benefits or any short-term disability benefits provided under the Business Benefit Plans. Each written employment agreement between any Business Employee or any Business Subsidiary that is effective prior to the Closing shall remain effective in accordance with its terms after the Closing.

(b) No <u>Duplicative Benefits</u>. Notwithstanding anything herein to the contrary, nothing in this Agreement shall be construed to require Buyer or any of its Subsidiaries to provide duplicative benefits or accruals to any Business Employee (or anyone else) or to employ or continue to employ any Business Employee or anyone else, and except to the extent expressly provided in this <u>Section 5.04</u>, nothing in this Agreement shall be construed as in any way limiting or restricting Buyer's ability to amend, modify or terminate any benefit plan at any time and from time to time, including any Business Benefit Plan, after the Closing.

5.05. Post-Closing Access to Record and Personnel. For a period of five (5) years after the Closing Date, upon receipt of reasonable prior notice, Buyer and Sellers each agree to provide, or cause to be provided, to each other, as soon as reasonably practicable after written request therefor and at the requesting Party's sole expense (but only for the reasonable out-of-pocket costs and expenses incurred by the Party in providing such access), reasonable access during normal business hours, and in a manner so as not to unreasonably interfere with the conduct of such other Party's business, to the other Party's employees and to any books, records, documents, instruments, accounts, correspondence, writings, evidences of title and other papers relating solely to the conduct of the Business on or before the Closing Date, in the possession or under the control of the other Party that the requesting Party reasonably needs: (a) to comply with reporting, disclosure, filing or other requirements imposed on the requesting Party by a Governmental Authority having jurisdiction over the requesting Party or by applicable stock exchange rules, (b) for use in any Proceeding, or in order to satisfy Tax, audit, accounting, claims, regulatory, or other similar requirements, (c) in connection with the filing of any Tax Return or election or any amended return or claim for refund, determining a Liability for Taxes or a right to a refund of Taxes or any Tax audit or other Proceeding in respect of Taxes or (d) to comply with its obligations under this Agreement or any of the other Transaction Documents; provided, however, that no Party will be required to provide access to or disclose information where such access or disclosure would be reasonably expected to violate any Law or Contract, or waive any attorney-client or other similar privilege (provided, that in the event such provision of information would reasonably be expected to violate any Law or Contract or waive any attorney-client or other similar privilege, the Party with such information will inform the other as to the general nature of what is being withheld as a result of the foregoing and will take all reasonable measures to permit the disclosure in a manner that avoids any such violation or waiver). Any information owned by a Party that is provided to a requesting Party pursuant to this Section 5.05 will be deemed to remain the property of the providing Party. Unless specifically set forth herein, nothing contained in this Agreement will be construed as granting or conferring rights of license or otherwise in any such information. No Party will have any Liability to the other Party in the event that any information exchanged or provided pursuant to this Section 5.05 is found to be inaccurate or is destroyed or lost after commercially reasonable efforts by such Party to comply with the provisions of this section 5.05.

5.06. Expenses. Except as otherwise provided herein, Buyer, on the one hand, and Sellers, on the other hand, agree to each bear their own expenses, fees, and costs in connection with the transactions contemplated by this Agreement; provided, however, that (a) all transfer, documentary, sales, use, stamp, registration, recording and other such similar Taxes, and all other conveyance fees, recording charges and other charges and fees (including any penalties and interest) incurred in connection with consummation of the transactions contemplated by this Agreement ("Transfer Taxes") shall be paid fifty percent (50%) by Sellers and fifty percent (50%) by Buyer, and (b) all costs required to obtain the consents of Persons listed on the Consent Schedule shall be paid by Sellers. Notwithstanding the foregoing sentence, Buyer shall pay 100% of the UK Stamp Tax within 30 days following the Closing Date. The Parties agree to reasonably cooperate to (i) sign and deliver any resale and other certificates or forms as may be necessary or appropriate to establish an exemption from (or otherwise reduce) any Transfer Taxes and (ii) prepare and file (or cause to be prepared and filed) all Tax Returns in respect of any such Transfer Taxes.

5.07. Wrong Pockets Clause.

(a) If, after the Closing, either Seller receives any amounts with respect to Receivables, such Seller will remit such amounts to Buyer within five (5) Business Days after such Seller's receipt thereof.

(b) If, after the Closing, Buyer receives any amounts with respect to Excluded Scheduled Receivables, Buyer will remit such amounts to Sellers within five (5) Business Days after such Buyer's receipt thereof.

(c) If, after the Closing, either Buyer on the one hand, or Sellers or any of its Subsidiaries, on the other hand, becomes aware that any of the Purchased Assets (including any assets of any type deemed in the "Purchased Assets" pursuant to the final sentence in Section 2.02(a)) has not been transferred to Buyer or that any of the Excluded Assets has been transferred to Buyer, it shall promptly notify the other Party in writing and the Parties shall, as soon as reasonably practicable, ensure that such property is transferred and delivered, with any necessary prior third party consent or approval, to (a) Buyer, in the case of any Purchased Asset that was not transferred and delivered to Buyer at the Closing; or (b) Seller, in the case of any Excluded Asset that was transferred or delivered to Buyer at the Closing.

5.08. Further Assurances.

- (a) Upon request, from time to time, each Party agrees that it shall (or direct its employees to, if applicable) execute and deliver all documents, make all rightful oaths, testify in any proceedings and do all other acts which may be necessary or desirable in the reasonable opinion of the other Party to consummate the transactions contemplated hereby.
- (b) After Closing, in each case, in accordance with the terms of the Transition Services Agreement, Veru will, no later than the date that is thirty days following the Closing Date, notify the FDA in writing of the change in ownership of the Product and related establishments, as applicable, in favor of Buyer and to permit Buyer to operate the business as contemplated under this Agreement, and take all other commercially reasonable steps to address any questions the FDA or other foreign, federal and state regulatory authorities may have regarding the transfer of ownership of the Product and related establishments and to support Buyer in any regulatory inquiries related to such transfer of ownership.

5.09. Tax Matters.

(a) Sellers shall, at their own expense, prepare or cause to be prepared and timely file or cause to be timely filed all Income Tax Returns of the Business Subsidiaries, for all periods ending on or prior to the Closing Date that have not yet been filed and are required to be filed after the Closing Date. Such Income Tax Returns shall be prepared in accordance with the past practices employed in the preparation of such Income Tax Returns, except as required by Law. At least thirty (30) days prior to the due date for filing any such Income Tax Returns, Sellers shall provide Buyer with a draft copy of such Income Tax Returns. Buyer shall notify Sellers of any reasonable objections Buyer may have to any items set forth in such draft Income Tax Returns and Sellers and Buyer agree to consult and resolve in good faith any such objections. If the parties cannot resolve any such objections, the item in question shall be resolved by the Arbitrating Accountant. The fees and expenses of the Arbitrating Accountant shall be borne one-half by Sellers and one-half by Buyer. Sellers shall pay or cause to be paid any cash Taxes due in respect of such Income Tax Returns except to the extent that any such Taxes are taken into account in the calculation of Closing Working Capital, the Closing Indebtedness Amount, or the Final Pre-Closing Taxes Amount. If any such Income Tax Return must be signed by Buyer, any Affiliate thereof, or any of the Business Subsidiaries (or any representative of the foregoing) following the Closing Date, Buyer agrees that it will (or will cause such other parties to) cooperate fully and punctually in signing such Tax Return in order to permit the timely filing of such Income Tax Return.

- (b) Buyer shall timely prepare and file, or cause to be timely prepared and filed, all other Tax Returns of the Business Subsidiaries for any Pre-Closing Taxable Period. Income Tax Returns that are required to be filed by or with respect to the Business Subsidiaries for Straddle Periods ("Straddle Returns") shall be prepared consistently with past practice to the extent permitted by applicable Law. Buyer shall provide, or cause to be provided, to Sellers a draft of any Straddle Return at least 30 days prior to the due date, giving effect to extensions thereto, for filing such Income Tax Return, for review by Sellers. Sellers shall notify Buyer of any reasonable objections Sellers may have to any items set forth in such draft Straddle Return and Buyer and Sellers agree to consult and resolve in good faith any such objection. If the parties cannot resolve any such objections, the item in question shall be resolved by the Arbitrating Accountant. The fees and expenses of the Arbitrating Accountant shall be borne one-half by Sellers and one-half by Buyer. Sellers shall be responsible for the cash amount of any income Taxes due for a Pre-Closing Taxable Period except to the extent that any such Taxes are taken into account in the calculation of Closing Working Capital, the Closing Indebtedness Amount, or the Final Pre-Closing Taxes Amount, and Buyer shall notify Sellers of any such cash amounts of income Taxes due from Sellers in respect of any Straddle Return no later than 10 Business Days prior to the date on which such Straddle Return is due, and no later than five Business Days prior to the date on which such Straddle Return is due, Sellers shall pay to Buyer the cash amount of the income Taxes for which Sellers are responsible.
- (c) For the sole purpose of apportioning any Tax relating to a Straddle Period, the portion of such Straddle Period that ends on the Closing Date shall be (i) in the case of a Tax that is not based or measured by income, receipts, profits, or wages or that is not imposed in connection with any sale or other transfer or assignment of property or any other specifically identifiable transaction or event, the total amount of such Tax for the full taxable period that includes the Closing Date (or, in the case of such Taxes determined on an arrears basis, the amount of such Taxes for the immediately preceding period) multiplied by a fraction, the numerator of which is the number of days from the beginning of such taxable period to and including the Closing Date and the denominator of which is the total number of days in such full taxable period, and (ii) in the case of a Tax that is based on or measured by income, receipts, profits or wages of the Business Subsidiaries or imposed in connection with any sale or other transfer or assignment of property or any other specifically identifiable transaction or event, the Tax that would be due based on an interim closing of the books as of the close of business on the day before the Closing Date. All determinations necessary to give effect to the foregoing allocations shall be made in a manner consistent with the prior practices of the Business Entity unless otherwise required by applicable Law.
- (d) To the extent permitted by applicable Law at a "more likely than not" (or greater) level of comfort, any net operating losses, credits or other income Tax attributes of any Business Subsidiary as of the Closing Date shall first be used to reduce the cash amount of any income Taxes relating to a Pre-Closing Taxable Period for which Sellers are responsible pursuant to Section 5.09(a) or 5.09(b) before they can be used for any other purpose.

- (e) Any refund of Taxes of, or any credit in lieu of a refund of Taxes of, any Pre-Closing Taxable Period (including any interest actually received or credited with respect thereto, and including any overpayment of income Taxes credited to estimated income Taxes), but excluding any refund or credit arising from the carryback of any Tax attribute arising after the Closing, if received by, or credited to, Buyer or the Business Subsidiaries or any other affiliated entity of Buyer, net of applicable costs, expenses and Taxes thereon, shall be the property of Sellers and shall be payable promptly to Sellers. None of Buyer nor the Business Subsidiaries shall waive any carryback of any Tax attribute of the Business Subsidiaries generated in or otherwise attributable to a Pre-Closing Taxable Period if such waiver would reduce or delay the amount due to Sellers pursuant to this Section 5.09(e). All payments required to be made pursuant to this Section 5.09(e) shall be made within ten (10) days after receipt of the Tax refund or credit against Taxes (or other benefit) from the applicable Governmental Authority by Buyer or any of its Affiliates (including the Business Subsidiaries following the Closing).
- (f) Without duplication of any refunds or other amounts payable to Sellers under Section 5.09(e), Buyer shall pay to Sellers, an amount of cash equal to the amount by which (i) the Final Pre-Closing Taxes exceeds (ii) the amount of Pre-Closing Taxes actually paid ("Actual Pre-Closing Taxes") (determined as of the date on which the final Income Tax Return of the Business Subsidiaries for the Pre-Closing Taxable Period is filed) within 30 days after the filing of the last Income Tax Return of the Business Subsidiaries for the Pre-Closing Taxable Period. All such Income Tax Returns shall be prepared by Buyer in good faith and in a manner consistent with the agreements and covenants set forth in this Section 5.09.
- (g) Without the prior written consent of Sellers (which consent shall not to be unreasonably withheld, delayed or conditioned), Buyer shall not, and shall cause the Business Subsidiaries not to, (i) amend or cause the amendment of a Tax Return of the Business Subsidiaries with respect to any Pre-Closing Taxable Period, (ii) change an annual accounting period or adopt or change any accounting method of the Business Subsidiaries with respect to any Pre-Closing Taxable Period, (iii) file any ruling request with any Governmental Authority that relates to Taxes or Tax Returns of the Business Subsidiaries for a Pre-Closing Taxable Period, (iv) other than any election pursuant to Section 338(g) of the Code, which shall be expressly permitted, file or amend any Tax election concerning the Business Subsidiaries with respect to any Pre-Closing Taxable Period, if, in any such case, such action could reasonably be expected to reduce the amount of any Tax refund or credit that is the property of Sellers under Section 5.09(e), increase the Taxes of Sellers for a Pre-Closing Taxable Period, increase any Indemnified Taxes, increase the Actual Pre-Closing Taxes or increase any Taxes reflected in the calculation of Closing Working Capital, the Closing Indebtedness Amount or the Final Pre-Closing Taxes Amount.
- (h) Buyer and Sellers further agree, upon request, to provide the other Party with all information in such Party's possession that such other Party may be required to report pursuant to Code § 6043 or Code § 6043A or Treasury Regulations promulgated thereunder.

- (i) Buyer and Sellers shall cooperate fully, as and to the extent reasonably requested by the other Party, in connection with the filing of Income Tax Returns and any audit, litigation or other proceeding with respect to income Taxes. Such cooperation shall include the retention and (upon the other Party's request) the provision of records and information that are reasonably relevant to any such audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of material provided hereunder. Buyer and Sellers agree (i) to retain all books and records with respect to income Tax matters pertinent to the Business Subsidiaries relating to any taxable period beginning before the Closing Date until the expiration of the statute of limitations (including any extensions thereof) applicable to the respective income Tax periods and to abide by all record retention agreements entered into with any Governmental Authority, and (ii) to give the other Party reasonable written notice prior to transferring, destroying or discarding any such books and records and, if the other Party so requests, Buyer or Sellers, as the case may be, shall allow the other Party to take possession of such books and records.
- (j) From and after the Closing Date, Buyer shall give prompt notice to Sellers if any Governmental Authority provides notice of an intent to audit, review or conduct any other proceeding with respect to the Taxes of the Business Subsidiaries for any Pre-Closing Taxable Period. Buyer shall control the conduct of any Tax audit or proceeding involving the Business Subsidiaries that occurs after the Closing Date. Notwithstanding the foregoing, if such audit or proceeding could reasonably be expected to reduce the amount of any Tax refund or credit that is the property of Sellers under Section 5.09(e), increase the Taxes of Sellers for a Pre-Closing Taxable Period, increase any Indemnified Taxes, increase the Actual Pre-Closing Taxes or increase any Taxes reflected in the calculation of Closing Working Capital, the Closing Indebtedness Amount or the Final Pre-Closing Taxes Amount, then (i) Buyer shall keep Sellers reasonably informed of the progress of any such audit or other proceeding involving a Pre-Closing Taxable Period (including providing copies of all written communication with any taxing authority), (ii) Sellers shall be entitled to participate in the defense of any such audit or other proceeding and to employ counsel of their choice for such purpose, the fees and expenses of which separate counsel shall be borne by Sellers, and (iii) if Buyer resolves, settles, compromises or abandons any issue or claim with respect to such an audit or other proceeding, or causes a Business Subsidiary to so resolve an issue or claim, without the prior written consent (which consent shall not be unreasonably withheld, delayed or conditioned) of Sellers, then such resolution shall not be determinative of the Sellers' indemnification obligations with respect to such issue or claim hereunder.
- (k) The Business Subsidiaries shall treat any Tax deductions attributable to any transaction expenses that were economically borne by Sellers on or before the Closing Date as deductible in a Pre-Closing Taxable Period to the extent such amounts are deductible in such Pre-Closing Taxable Period at a "more likely than not" or higher level of confidence under applicable Law.
- (l) Buyer shall not, and shall not allow the Business Subsidiaries to, engage in any transaction after the Closing, but on the Closing Date, that is outside of the ordinary course of business and is not contemplated by this Agreement and could reasonably be expected to reduce the amount of any Tax refund or credit that is the property of the Seller under Section 5.09(e), increase the Taxes of Sellers for a Pre-Closing Taxable Period, increase any Indemnified Taxes, increase the Actual Pre-Closing Taxes or increase any Taxes reflected in the calculation of Closing Working Capital, the Closing Indebtedness Amount or the Final Pre-Closing Taxes Amount.

5.10. Name Change. As soon as reasonably practicable following the Closing, Limited shall change its legal entity name to a name that does not include, is not similar to, and does not contain any words confusingly similar to, "The Female Health Company" and/or "FHC".

5.11. Use of Inventories of Products

- (a) Sellers hereby grant to Buyer and its Affiliates a royalty-free, fully paid-up, non-exclusive, sublicensable (but solely to service providers, distributors, resellers or channel partners), non-transferable (except as expressly set forth in Section 5.11(b)), irrevocable, and worldwide license to use any Marks in the Excluded Assets (including the "Veru" trade name) or otherwise owned or licensable by any Veru Affiliate in each case that are displayed or included on the Products, packaging for the Products, or other materials included in the Purchased Assets solely to advertise, market, promote and sell the Inventory. The license granted pursuant to this Section 5.11(a) shall automatically expire upon the first to occur of (a) five years from the Closing Date and (b) such time as all Inventory has been sold. For as long as the license granted pursuant to this Section 5.11(a) is in effect, Buyer agrees that it will not ship, transfer or use any Inventory (i) unless Buyer first has all Regulatory Approvals and other licenses or approvals from applicable Governmental Authorities required to so ship, transfer or use any such Products in compliance with applicable Law or (ii) later than the "use-by" or similar expiration date applicable to such Inventory.
- (b) <u>Transfer</u>. The license granted in Section 5.11(a) shall not be transferable; *provided* that such license rights may be assigned upon written notice to Veru without such consent in connection with a Change of Control of Buyer or any of its Affiliates.
- 5.12. <u>Sufficient Capitalization of Business Subsidiaries</u>. Buyer shall take commercially reasonable efforts to provide, that for a period of two (2) years following the Closing Date, the Business Subsidiaries shall hold at least \$1,000,000 of aggregate cash, in the aggregate, across any accounts of the Business Subsidiaries.
- 5.13. Covenant Not To Sue Veru hereby irrevocably covenants that at no time will any party in the Veru Group, directly or indirectly, alone or by, with, or through others, (i) commence, maintain or prosecute, or (ii) induce, authorize, assist, participate, or cooperate in the commencement, maintenance, or prosecution of, in the case of clauses (i) or (ii), any action or proceeding of any kind (including any suit, complaint, grievance, demand, claim, or cause of action in, of, or before any Government Authority) against any party in the Buyer Group or any customer, manufacturer, reseller, channel partner, distributor, marketing partner, licensee, or other transferee of any party in the Buyer Group, involving any allegation or assertion of any direct or indirect infringement of any claim of any Covered Seller Patent by the operation of the Business or by the Exploitation of: (a) any product, service, device or technology in the Business Field; (b) any Purchased Asset; or (c) the Product (including any improvements, modifications, derivatives, variants, or extensions of or to the Product).

- (a) "Business Field" means the field comprised of the development, manufacture, marketing, distribution or sale of any female condom products anywhere in the world.
- (b) "Buyer Group" means Buyer, its Affiliates, any of their respective future Affiliates (including any future Affiliate following any Change of Control), or any assignee or transferee to the transfer or assignment of any asset by Buyer, its Affiliates, or any of their respective future Affiliates (including any future Affiliate following any Change of Control).
 - (c) "Covered Seller Patent" means each Patent that is owned or licensable by Veru or any of its Affiliates as of the Closing Date.
- (d) "Exploitation" means to make (including the right to practice methods, processes and procedures), have made, use, sell, offer for sale, import, and otherwise distribute, dispose of or exploit.
- (e) "Veru Group" means Veru, its Affiliates, any of their respective future Affiliates (including any future Affiliate following any Change of Control), or any assignee or transferee to the transfer or assignment of any asset by Veru, its Affiliates, or any of their respective future Affiliates (including any future Affiliate following any Change of Control).

6. <u>Indemnification</u>.

- 6.01. <u>Indemnification by Veru.</u> Subject to the limitations and expiration dates contained in this <u>Section 6</u>, from and after the Closing, and regardless of any investigation made at any time by or on behalf of Buyer or any information Buyer may have, Veru, on behalf of itself, its Subsidiaries and its successors and assigns, hereby covenants and agrees to indemnify, defend and hold Buyer and its Affiliates, stockholders, directors, officers, employees and agents (collectively, the "<u>Buyer Indemnified Parties</u>") harmless from and against, and reimburse the Buyer Indemnified Parties for all costs of, any demand, claim, damage, Liability, Loss, cost or expense actually incurred, suffered, sustained by, or imposed upon any Buyer Indemnified Party (collectively, "<u>Buyer's Damages</u>") directly or indirectly arising out of, with respect to, resulting from or relating to, or by reason of:
- (a) Any inaccuracy in, or breach of, any representation or warranty of Sellers made in Section 3 of this Agreement (as modified by the Schedules);
- (b) any breach, non-fulfillment or failure of either Seller to duly perform or observe any term, provision, covenant or agreement to be performed or observed by such Seller or any of its Subsidiaries or Affiliates pursuant to this Agreement;
 - (c) any Indebtedness of any Business Entity to the extent not paid in full at or prior to Closing;
- (d) any brokerage or finders' fees or commissions or similar payments based upon any agreement or understanding alleged to have been made by any Person with Sellers (or any Person acting on their behalf) in connection with the transactions contemplated hereby;

- (e) any Excluded Asset or any Excluded Liability;
- (f) any Pre-Closing Taxes (except to the extent that any such Taxes are taken into account in the calculation of Closing Working Capital, the Closing Indebtedness Amount, or the Final Pre-Closing Taxes Amount); or
 - (g) any R&W Exclusion.
- 6.02. Indemnification by Buyer. Subject to the limitations and expiration dates contained in this Section 6, from and after the Closing, and regardless of any investigation made at any time by or on behalf of Sellers or any information Sellers may have, Buyer, on behalf of itself and its successors, hereby covenants and agrees to indemnify, defend and hold Sellers and their respective Affiliates, shareholders, directors, officers, employees and agents (collectively, the "Seller Indemnified Parties") harmless from and against, and shall reimburse the Seller Indemnified Parties for all costs they incur with respect to, any demand, claim, damage, Liability, loss, cost or expense actually suffered or incurred by such Seller Indemnified Party (collectively, "Sellers' Damages") directly or indirectly arising out of, resulting from or relating to:
 - (a) any breach of any representation or warranty of Buyer made in Section 4 of this Agreement;
- (b) any breach or failure of Buyer to duly perform or observe any term, provision, covenant or agreement to be performed or observed by Buyer pursuant to this Agreement; or
- (c) any brokerage or finders' fees or commissions or similar payments based upon any agreement or understanding alleged to have been made by any Person with Buyer (or any Person acting on its behalf) in connection with the transactions contemplated hereby.

6.03. <u>Indemnification Procedure</u>.

(a) Following the discovery of any facts or conditions that could reasonably be expected to give rise to a claim for either Buyer's Damages or Sellers' Damages, the Person seeking indemnification under this Agreement (the "Indemnified Party") shall promptly, but in any event within thirty (30) days thereafter, provide written notice to the Party from whom indemnification is sought (the "Indemnifying Party"), setting forth, to the extent known by the Indemnified Party, the facts and circumstances, in reasonable detail, relating to such Buyer's Damages or Sellers' Damages, as applicable, the amount of such Buyer's Damages or Sellers' Damages (or a non-binding, reasonable estimate thereof if the actual amount is not known or not capable of reasonable calculation) and the Section(s) of this Agreement upon which the Indemnified Party is relying in seeking such indemnification (such written notice being hereinafter referred to as a "Notice of Claim"); provided that no delay in providing such Notice of Claim shall affect an Indemnified Party's rights hereunder, unless (and then only to the extent that) the Indemnifying Party is materially and adversely prejudiced in connection therewith. In the event the Indemnifying Party does not notify the Indemnified Party within thirty (30) days following its receipt of a Notice of Claim that the Indemnifying Party disputes its liability to the Indemnified Party under this Section 6 or the amount thereof, the claim specified by the Indemnified Party in such Notice of Claim shall be conclusively deemed a Liability of the Indemnifying Party under this Section 6 (subject to all of the limitations and other terms of this Section 6), and the Indemnifying Party shall pay the amount of such Liability that it is required to pay under this Section 6 to the Indemnified Party on demand or, in the case of any Notice of Claim in which the amount of the claim (or any portion of the claim) is estimated, on such later date when the amount of such claim (or such portion of such claim) becomes finally determined (subject to all of the limitations and other terms of this Section 6). In the event the Indemnifying Party has timely disputed its Liability with respect to such claim as provided above, such Indemnified Party and the appropriate Indemnifying Party, as promptly as possible, shall establish the merits and amount of such claim (by mutual agreement, litigation, arbitration or otherwise) and, within five (5) Business Days following the final determination of the merits and amount of such claim, the Indemnifying Party shall pay to the Indemnified Party in immediately available funds an amount equal to such claim as determined hereunder (subject to all of the limitations and other terms of this Section 6).

(b) The following provisions shall apply to any claim subject to indemnification under this Agreement that is filed, demanded or instituted by any third party, including any Governmental Authority (a "Third Party Claim"):

(i) Upon receipt of a Notice of Claim relating to a Third Party Claim, the Indemnifying Party shall have the right to assume control of the defense of the Third Party Claim; provided, however, that the Indemnified Party may observe any such Third Party Claim with counsel of its choice and at its expense; provided further, that the Indemnifying Party shall not be entitled to assume or continue control of the defense of any Third Party Claim if [A] the Third Party Claim relates to or arises in connection with any criminal proceeding or [B] the Third Party Claim primarily seeks an injunction or equitable relief against any Indemnified Party. The Indemnifying Party shall keep the Indemnified Party apprised of all material developments, including settlement offers, with respect to the Third Party Claim and permit the Indemnified Party to participate in the defense of the Third Party Claim at the Indemnified Party's expense. If the Indemnifying Party exercises its right to defend under this Section 6.03(b)(i), the Indemnified Party (such consent to the entry of any judgment or enter into any settlement with respect to such Third Party Claim without the prior written consent of the Indemnified Party (such consent not to be unreasonably withheld); provided, however, that [X] the Indemnifying Party may settle any Third Party Claim without the consent of the Indemnified Party from all liabilities with respect to such Third Party Claim and [Y] to the extent the consent of the Indemnified Party is required to settle a Third Party Claim and the Indemnified Party reasonably objects to a settlement offer the Indemnifying Party desires to accept, the Indemnifying Party shall continue to contest or defend such Third Party Claim but the maximum liability of the Indemnifying Party as to such Third Party Claim shall not exceed the amount of such settlement offer, plus other Buyer's Damages or Sellers' Damages, as applicable, paid or incurred by the Indemnified Party and recoverable under this Section 6 up to the point such settlem

(iii) After any indemnification payment is made to any Indemnified Party pursuant to this <u>Section 6</u>, the Indemnifying Party shall, to the extent of such payment, be subrogated to all rights (if any) of such Indemnified Party against any third party in connection with the Buyer's Damages or Sellers' Damages, as applicable, to which such payment relates. Without limiting the generality of the preceding sentence, any Indemnified Party receiving an indemnification payment pursuant to the preceding sentence shall execute, upon the written request of the Indemnifying Party, any instrument reasonably necessary to evidence such subrogation rights.

(iv) The Indemnified Party shall, at the Indemnifying Party's sole cost and expense (without any obligation to reimburse the Indemnified Party for the time spent by its employees on the Third Party Claim), cooperate with and make available to the Indemnifying Party such assistance, personnel, witnesses and materials as the Indemnifying Party may reasonably request with respect to any indemnifiable matter hereunder, including any Third Party Claim.

6.04. Limitations on the Indemnification Obligations of Sellers.

(a) Survival Periods for Representations, Warranties and Covenants.

(i) Subject to <u>clause (iii)</u> below, the representations and warranties set forth in <u>Section 3.01</u> (*Organization*), <u>Section 3.02</u> (*Authorization of Agreement; No Violation*), <u>Section 3.03</u> (*Capitalization of the Business Subsidiaries*), <u>Section 3.04</u> (*Consents and Approvals*), <u>Section 3.14(b)</u> (*Patents*), <u>Section 3.16</u> (*Brokers*), and <u>Section 3.20</u> (*Title and Sufficiency of Assets*) (collectively, the "<u>Fundamental Representations</u>") shall survive the Closing until thirty (30) days after the expiration of the statute of limitations applicable to the underlying breach of representation or warranty, at which time the Fundamental Representations shall terminate. Subject to <u>clause (iii)</u> below, the representations and warranties set forth in <u>Section 3.08</u> (*Taxes*) (the "<u>Tax Representations</u>") survive the Closing until thirty (30) days after the expiration of the statute of limitations applicable to the underlying breach of representation or warranty, at which time the Fundamental Representations shall terminate. Subject to <u>clause (iii)</u> below, the representations and warranties set forth in <u>Section 3.14</u> other than those in <u>Section 3.14(b)</u> (*Intellectual Property*) and <u>Section 3.18</u> (*Legal Compliance*) (collectively, the "<u>Special Representations</u>") shall survive the Closing until the forty-eight (48) month anniversary of the Closing Date.

- (ii) Except for claims of Fraud, and subject to <u>clause (iii)</u> below, all representations and warranties of Sellers set forth in <u>Section 3</u> of this Agreement or in any of the Transaction Documents, other than the Fundamental Representations, the Special Representations and the Tax Representations (which each are subject to <u>clause (ii)</u>), shall survive the Closing until the eighteen (18) month anniversary of the Closing Date.
- (iii) Sellers' obligation to indemnify the Buyer Indemnified Parties with respect to a claim for a breach of any of the representations and warranties of Sellers set forth in Section 3 of this Agreement shall extend (with respect to such claim) beyond the applicable survival period only for claims of Fraud or if the Buyer Indemnified Party asserts such claim in a Notice of Claim duly delivered to Sellers prior to the expiration of such survival period.
- (b) <u>De Minimis Basket</u>. No Buyer Indemnified Party shall be entitled to indemnification from Sellers under<u>Section 6.01(a)</u> or <u>Section 6.01(e)</u> for any individual item or claim, or group or series of related items or claims, where the Buyer's Damages related thereto for which the Buyer Indemnified Party would otherwise be entitled to indemnification are less than \$20,000 (the "<u>De Minimis Basket</u>"); *provided*, *however*, that if the Buyer's Damages related thereto for which the Buyer Indemnified Party would otherwise be entitled to indemnification equal or exceed the De Minimis Basket, then the Buyer Indemnified Party shall be entitled to indemnification in full for all breaches of such representations and/or warranties without regard to the De Minimis Basket (but subject to the other limitations set forth in this <u>Section 7</u>). Notwithstanding the foregoing, the De Minimis Basket shall not apply to any indemnification obligations for Buyer's Damages to the extent resulting from a breach of the Fundamental Representations or the Tax Representations or any Claim based upon Fraud.
- (c) <u>Basket</u>. No Buyer Indemnified Party shall be entitled to indemnification from Sellers under Section 6.01(a) or Section 6.01(e) unless the aggregate amount of the indemnification obligations of Sellers under such Sections (but for this Section 6.04) exceeds a dollar amount equal to the Retention Escrow Amount (the "Basket"), in which event the Buyer Indemnified Parties shall, subject to the other limitations set forth in this Section 6, be entitled to indemnification for only such Buyer's Damages exceeding the Basket. Notwithstanding the foregoing, the Basket shall not apply to any indemnification obligations for Buyer's Damages to the extent resulting from a breach of the Fundamental Representations or the Tax Representations or any claim based upon Fraud.
- (d) <u>Cap.</u> Subject to the limitations contained in <u>Section 6.04(e)</u>, Sellers' aggregate liability under <u>Section 6.01(a)</u> shall not exceed \$2,000,000 (the "<u>Cap</u>"); *provided, however*, that the Cap shall not apply to any claims of Fraud or any breaches of, or inaccuracy in, the Fundamental Representations. Except in the case of Fraud, in no event shall Sellers' aggregate liability under <u>Section 6.01</u> exceed the Purchase Price. For the avoidance of doubt, and without limiting any of the other limitations set forth in this <u>Section 6.04</u>, Sellers' aggregate liability under <u>Section 6.01(a)</u> shall not exceed the Retention Escrow Amount, other than for Fraud or any breaches of, or inaccuracy in, Fundamental Representations, Special Representations or Tax Representations, provided that nothing in this sentence shall affect Buyer's ability to recover from the R&W Policy.

(e) Fu	rther Limitations	Subject to the	ne other	provisions in	n this Section 6.04:
--------	-------------------	----------------	----------	---------------	----------------------

- (i) Any liability for indemnification obligations for Losses under Section 6.01(a) (other than in respect of a breach of a Fundamental Representation, Special Representation, Tax Representation, or in the case of Fraud) shall, in each case, be satisfied (A) first, by recovery from the Retention Escrow Account until the Retention Escrow Account's exhaustion or release and (B) thereafter, by recovery solely from the R&W Policy in accordance with the terms thereof.
- (ii) Except in the case of Fraud, any liability for indemnification obligations for Losses under<u>Section 6.01(a)</u> in respect of a breach of a Fundamental Representation, Special Representation or Tax Representation or any other subsection of <u>Section 6.01</u>, shall, in each case, be satisfied (A) *first*, by recovery from the Retention Escrow Account until the Retention Escrow Account's exhaustion or release, (B) *second* by recovery from the R&W Policy in accordance with the terms thereof, and (C) *finally*, to the extent that recovery is not available under R&W Policy for any reason, by recovery directly from the Sellers.
- (iii) Termination, modification or amendment to the R&W Policy or denial of claims under the R&W Policy will not impact any of the limitations on indemnification set forth in this Agreement except as expressly set forth herein. The R&W Policy shall include: (a) a provision whereby the insurer expressly waives, and agrees not to pursue, directly or indirectly, any subrogation rights against any Seller or any Seller Affiliate with respect to any claim made by any insured thereunder (except in the case of Fraud by the applicable Person), (b) a provision that provides that Buyer is not obligated to bring a claim against any Seller or any Seller Affiliate prior to bringing a claim under the R&W Policy, and (c) a provision designating the Sellers as third-party beneficiaries of the clauses referenced in clauses (a) and (b). Following the Closing, Buyer will not agree to the provisions of the R&W Policy addressing the requirements of the previous sentence being amended or modified in a manner adverse to the Sellers. Notwithstanding the foregoing or anything to the contrary set forth in this Agreement, none of the limitations or exceptions set forth in this Section 6 including any periods of survival with respect to the representations, warranties, covenants and agreements set forth herein, shall in any way limit or modify the ability of the Buyer Indemnified Parties to make claims under or recover against the R&W Policy.
- (iv) The Buyer Indemnified Parties shall not be entitled to indemnification under this Agreement to the extent of any insurance proceeds actually received by the Buyer Indemnified Party in connection with the facts giving rise to such indemnification, as determined pursuant to Section 6.04(f) below.
- (v) The Buyer Indemnified Parties shall not be entitled to indemnification under this Agreement to the extent that the Buyer's Damages subject to a claim for indemnification resulted from failure of Buyer to undertake commercially reasonable efforts to mitigate such Buyer's Damages.

- (vi) The Buyer Indemnified Parties shall not be entitled to indemnification under this Agreement in connection with any Buyer's Damages with respect to which a Buyer Indemnified Party has a claim, right of indemnification or right of set off against any third party, unless the applicable Buyer Indemnified Party assigns such claim, right of indemnification or right of set off against such third party to Sellers.
- (vii) The Buyer Indemnified Parties shall not be entitled to indemnification under this Agreement more than once (i.e., on a dollar for dollar basis to the extent previously indemnified) for the same Buyer's Damages suffered, regardless if such Buyer's Damages may be attributed to more than one indemnity, breach of several paragraphs of the representations or warranties or the breach of or default in connection with several covenants or obligations herein.
- (viii) The Buyer Indemnified Parties shall not be entitled to indemnification under this Agreement to the extent amounts related to the claims for Buyer's Damages of the Buyer Indemnified Parties are taken into account in determining the Purchase Price under <u>Section 2.08</u>.
- (f) Insurance Proceeds. The amount of any indemnity provided in this Agreement shall be computed net of any insurance proceeds actually received by a Buyer Indemnified Party in connection with or as a result of any claim giving rise to an indemnification claim hereunder (including the R&W Policy, provided, that with respect to any amounts recovered under the R&W Policy, such reduction shall not apply with respect to any indemnification up to the Retention Escrow Amount). If the indemnity amount is paid prior to the Buyer Indemnified Party's actual receipt of insurance proceeds related thereto, Buyer shall assign its right to such insurance and allow Sellers to pursue collection of such insurance proceeds. In addition, if a Buyer Indemnified Party subsequently receives such insurance proceeds, then the Buyer Indemnified Party shall promptly pay to Sellers the amount of insurance proceeds subsequently received (net of all related costs, expenses and other Liabilities), but not more, in the aggregate, than the indemnity amount paid by Sellers.
- (g) <u>Materiality</u>. Notwithstanding anything contained herein or elsewhere to the contrary, all "material" and "Business Material Adverse Effect" or similar materiality type qualifications contained in the representations and warranties set forth in this Agreement shall be ignored and not given any effect for purposes of determining whether a breach of a representation or warranty has occurred or for calculating the amount of any Buyer's Damages.
- (h) <u>Certain Damages</u>. Sellers shall not have any liability under any provision of this Agreement for, and the amount of Buyer's Damages shall not include punitive damages (except to the extent such damages are payable in connection with a Third Party Claim).
- 6.05. <u>Multiple Representations and Warranties</u>. For the avoidance of doubt, it is understood that the facts and circumstances alleged to constitute a breach of a representation or warranty may be comprised of elements relating to more than one subject matter and, in such case, each such element will be assessed separately based on the representation or warranty contained herein addressing the subject matter in question to determine whether it amounts to a breach and, to the extent more than one representation or warranty cover the same subject matter, only one and the more specific representation or warranty will govern and no other representation and/or warranty will apply to establish a breach.

- 6.06. Tax Treatment. Any payments made to any Party pursuant to this Section 6 shall constitute an adjustment of the Purchase Price for Tax purposes and shall be treated as such by Buyer and Sellers on their Tax Returns to the greatest extent permitted by Law.
- 6.07. Exclusive Remedy. The parties acknowledge and agree that, after the Closing, the indemnification provisions in this Section 6 shall be the exclusive remedy of the parties with respect to the transactions contemplated by this Agreement, except for (a) claims of Fraud or (b) equitable relief permitted by Section 5.01 and Section 7.12.
- 6.08. Retention Escrow Amount. Within two (2) Business Days after the eighteen (18) month anniversary of the Closing Date, Buyer and the Sellers shall jointly instruct the Escrow Agent to pay to the Seller by wire transfer of immediately available funds to an account designated by the Seller an amount equal to the positive difference, if any, between (i) the then remaining balance of the Retention Escrow Amount, including any earnings accrued thereon, minus (ii) the amount of Buyer's Damages for which any Buyer Indemnified Party has timely made a claim for indemnification pursuant to this Section 6 that has not then been finally resolved. In the event any portion of the Retention Escrow Amount is not released to the Seller as a result of the foregoing sentence, Buyer and the Sellers shall, promptly and in any event within two (2) Business Days following the final determination of any such outstanding claims and, jointly instruct the Escrow Agent to disburse the then-remaining balance of the Retention Escrow Amount (or any portion thereof that relates to the claims that have then been finally determined), including any earnings accrued thereon, to the Sellers.

7. <u>General Provisions</u>.

7.01. Notices. Any notice to be given hereunder shall be deemed given and sufficient if in writing and (a) delivered by hand or by reputable overnight courier, (b) mailed by registered or certified mail or (c) sent by email (.PDF or .TIF attachment). Notices and communications to Buyer or Seller will, unless another address is specified in writing, be sent to the address indicated below:

If to Sellers, to:

Veru Inc. 2916 North Miami Ave, Suite 1000 Miami, FL 33127 Attention: Dr. Mitchell S. Steiner, Chairman, President and CEO Email: msteiner@verupharma.com

with a copy to:

Benjamin G. Lombard, Esq. Reinhart Boerner Van Deuren s.c. 1000 North Water Street, Suite 1700 Milwaukee, WI 53202

Email: blombard@reinhartlaw.com

If to Buyer, to:

Clear Future, Inc. 746 Lake Avenue Greenwich, CT 06830 Attention: Mr. Kyle Tatz

Email: tatz@roaringbrookholdings.com

with a copy to:

Jake D. Gatof Wilson Sonsini Goodrich & Rosati PC One Boston Place, 201 Washington St Suite 2000 Boston, MA 02108 Email: jgatof@wsgr.com

or to such other Person or address as any Party shall have specified by notice in writing to the other parties. If personally delivered, such communication shall be deemed delivered upon actual receipt; if sent by e-mail transmission, such communication shall be deemed delivered the day of the transmission or, if the transmission is not made on a Business Day before 5:00 p.m. (New York time) at the place of receipt, the first Business Day after transmission (and sender shall bear the burden of proof of delivery); if sent by U.S. mail, such communication shall be deemed delivered as of the date of delivery indicated on the receipt issued by the relevant postal service or, if the addressee fails or refuses to accept delivery, as of the date of such failure or refusal; and if sent by internationally recognized overnight delivery service, such communication shall be deemed delivered upon receipt.

- 7.02. Entire Agreement. This Agreement (including the Exhibits and Schedules hereto), together with the other agreements, documents and instruments referred to herein, constitutes the entire agreement of the Parties and supersedes all prior agreements and undertakings (including any indication of interest or any letter of intent entered into between the parties), both written and oral, between the Parties with respect to the subject matter hereof and, except as otherwise expressly provided herein, is not intended to confer upon any other Person any rights or remedies hereunder. There have been and are no representations, warranties or covenants between the Parties other than those set forth or provided for in this Agreement and the other agreements and instruments referred to herein.
- 7.03. <u>Incorporation of Terms</u>. The introductory language and Recitals set forth above, and the Annexes, Exhibits and Schedules identified herein, are incorporated into this Agreement by reference and made a part hereof.
- 7.04. No Other Representations, Warranties, Covenants and Agreements. No Party has made any representations, warranties, covenants or agreements, express or implied, of any nature whatsoever other than those representations, warranties, covenants and agreements expressly set forth in this Agreement or the Transaction Documents.

- 7.05. Waiver. Unless prohibited by Law, the covenants, obligations, agreements or conditions hereunder may be waived by the other Party to whom such compliance is owed by an instrument in writing duly executed by the Party granting such waiver. No waiver of any provision of this Agreement will be deemed, or will constitute, a waiver of any other term or provision, whether or not similar, or any subsequent breach or failure to comply with any such covenant, obligation, agreement or condition, nor will any waiver constitute a continuing waiver.
 - 7.06. Amendment. This Agreement may not be amended or modified except by an instrument in writing signed by or on behalf of each of the Parties.
- 7.07. Assignment; Binding Effect. Neither Party may assign, transfer or otherwise encumber this Agreement or its rights or obligations hereunder, in whole or in part, whether voluntarily or by operation of Law, without the prior written consent of the other Party, and any attempted assignment without such consent shall be void and without legal effect; provided that Buyer may assign or transfer its rights and obligations under this Agreement, in whole or from time to time in part, to any Affiliate of Buyer, provided that such transfer or assignment shall not relieve Buyer of its obligations hereunder or enlarge, alter or change any obligation of any Party hereto. Subject to the preceding sentence, this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and permitted assigns.
- 7.08. Benefit. Subject to the following sentence, but notwithstanding anything contained in this Agreement to the contrary, nothing in this Agreement, express or implied, is intended to confer on any Person other than the Parties hereto or their respective heirs, successors, executors, administrators and assigns any rights, remedies, obligations or Liabilities under or by reason of this Agreement. Each of the Buyer Indemnified Parties and the Seller Indemnified Parties is a third party beneficiary of, and entitled to enforce the provisions of, this Agreement.
- 7.09. Disclosure Schedules. Sellers have prepared the Schedules attached to this Agreement and delivered them to Buyer on the date hereof. The Schedules shall be arranged by number to correspond to the sections of this Agreement, and Seller shall be permitted to include in the Schedules any qualifications, disclosure or exceptions with respect to any subsection of Section 3. Any fact or item disclosed on any Schedule shall be deemed disclosed on all other Schedules to which an appropriate cross reference is made or on all other Schedules where it is reasonably apparent on its face that such disclosure applies to such other Schedules. The Schedules shall qualify the representations and warranties set forth in this Agreement or set forth other information required by this Agreement but shall not otherwise vary, change or alter the language of the representations and warranties contained in this Agreement or be interpreted as additional or independent representations or warranties. The inclusion of any item on any Schedule shall not constitute an admission that a violation, right of termination, default, liability or other obligation of any kind exists with respect to such item, but rather is intended only to qualify certain representations and warranties in this Agreement and/or to set forth other information required by this Agreement.

- 7.10. No Agreement Until Executed. Irrespective of negotiations among the parties or the exchanging of drafts of this Agreement, this Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding among the parties hereto unless and until this Agreement is executed by the parties hereto.
- 7.11. Governing Law. This Agreement shall be governed by the Laws of the State of Delaware without giving effect to any rule or provision thereof which would cause the application of the Law of any other state. EACH OF THE PARTIES HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT. The parties (a) hereby irrevocably and unconditionally submit to the jurisdiction of the state and federal courts located in Delaware for the purpose of any suit, action or other proceeding arising out of or based upon this Agreement, (b) agree not to commence any suit, action or other proceeding arising out of or based upon this Agreement except in the state and federal courts located in Delaware, and (c) hereby waive, and agree not to assert, by way of motion, as a defense, or otherwise, in any such suit, action or proceeding, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that the suit, action or proceeding is brought in an inconvenient forum, that the venue of the suit, action or proceeding is improper or that this Agreement or the subject matter hereof may not be enforced in or by such courts.
- 7.12. Specific Enforcement. The Parties agree that irreparable damage may occur in the event that any of the provisions of this Agreement are not performed in accordance with their specific terms or are otherwise breached and that any breach of this Agreement by a Party could not be adequately compensated in all cases by monetary damages alone. It is accordingly agreed that the Parties shall be entitled to an injunction, specific performance and other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof, in addition to any other remedy to which they are entitled in Law or in equity. Each of the Parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that the other Party has an adequate remedy at Law or that any award of specific performance is not an appropriate remedy for any reason at Law or in equity. Any Party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement shall not be required to provide any bond or other security in connection with such order or injunction.
- 7.13. Negotiation of Agreement. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event of an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement.
- 7.14. <u>Facsimile Signature; Counterparts.</u> This Agreement may be executed and delivered by electronic means (including by e-signature or by e-mail in .PDF or .TIF form) of original signatures and may be executed in counterparts, each of which shall be considered one and the same agreement, and shall become effective when such counterparts have been signed by each Party and delivered to the other Party.

- 7.15. Severability. Any term or provision of this Agreement which is invalid or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms and provisions of this Agreement or otherwise affecting the validity or enforceability of any of the terms or provisions of this Agreement in any other jurisdiction. If any provision of this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.
- 7.16. No Recourse. This Agreement and the other Transaction Documents may only be enforced against, and all claims, obligations, liabilities, or causes of action (whether in contract or in tort, in law or in equity, or granted by statute) that may be based upon, in respect of, arise under, out or by reason of, be connected with, or relate in any manner to this Agreement, or the negotiation, execution, or performance hereof or thereof (including any representation or warranty made in, in connection with, or as an inducement to, this Agreement) may only be brought against the entities that are expressly identified as Parties in the preamble to this Agreement (each, a "Contracting Party") or, with respect to another Transaction Document, the parties thereto, as applicable, and then only with respect to the specific obligations set forth in this Agreement with respect to such Party. No Person who is not a Contracting Party, including any Affiliate of a Party or a Party's or its Affiliates' current, former or future director, officer, employee, incorporator, member, partner, manager, equityholder, Affiliate, agent, attorney, representative or assignee of, and any financial advisor or lender to, any Contracting Party, or any current, former or future director, officer, employee, incorporator, member, partner, manager, equityholder, Affiliate, agent, attorney, representative or assignee of, and any financial advisor or lender to, any of the foregoing (collectively, "Nonparty Affiliates"), shall have any liability (whether in contract or in tort, in law or in equity, or granted by statute) for any claims, causes of action, obligations, or liabilities arising under, out of, in connection with, or related in any manner to this Agreement or based on, in respect of, or by reason of this Agreement or its negotiation, execution, performance, or breach (other than to the extent a specific named party and signatory to another Transaction Document and then only to the extent of the obligations of such Person set forth therein); and, to the maximum extent permitted by Law, each Contracting Party hereby waives and releases all such liabilities, claims, causes of action, and obligations against any such Nonparty Affiliates. Without limiting the foregoing, to the maximum extent permitted by law, (a) with respect to disputes under or arising out of this Agreement, each Contracting Party hereby waives and releases any and all rights, claims, demands, or causes of action that may otherwise be available at law or in equity, or granted by statute, to avoid or disregard the entity form of a Contracting Party or otherwise impose liability of a Contracting Party on any Nonparty Affiliate, whether granted by statute or based on theories of equity, agency, control, instrumentality, alter ego, domination, sham, single business enterprise, piercing the veil, unfairness, undercapitalization, or otherwise; and (b) each Contracting Party disclaims any reliance upon any Nonparty Affiliates with respect to the performance of this Agreement or any representation or warranty made in, in connection with, or as an inducement to enter into, this Agreement. Each of the Nonparty Affiliates is an intended third-party beneficiary of this Section 7.16.

[Remainder of page intentionally left blank.]

IN WITNESS WHEREOF, the parties hereto have caused this St	ock and Asset Purchase Agreement to be signed by their respective officers there	eunto duly
authorized, all as of the date first written above.		

SELLERS:

VERU INC.

BY <u>/s/ Mitchell Steiner</u>
Its: <u>Chairman, Chief Executive Officer and President</u>

THE FEMALE HEALTH COMPANY LIMITED

BY <u>/s/ Mitchell Steiner</u> Its: <u>Director</u>

BUYER:

CLEAR FUTURE, INC.

BY <u>/s/ Kyle Tatz</u> Its: <u>Director</u>



Investor and Media Contact:
Samuel Fisch
Executive Director, Investor Relations and Corporate Communications
Email: veruinvestor@verupharma.com

Veru Announces the Sale of the FC2 Female Condom® (Internal Condom) Business

MIAMI, FL – December 31, 2024 – Veru Inc. (NASDAQ: VERU), a late clinical stage biopharmaceutical company focused on developing innovative medicines for preserving muscle for high quality weight loss, oncology, and viral induced acute respiratory distress syndrome, today announced that it has sold its FC2 Female Condom® (Internal Condom) business to clients managed by Riva Ridge Capital Management LP, a New York City-based investment management firm as well as other co-investors, for \$18 million, subject to adjustment as set forth in the purchase agreement.

"The monetization of the FC2 business allows Veru to be a pure biopharmaceutical company focusing its additional nondilutive resources on the execution and development of its promising late-stage clinical drug pipeline," said Mitchell Steiner, M.D., Chairman, President, and Chief Executive Officer of Veru Inc. "We are excited about the Company's successful strategic evolution to the treatment of cardiometabolic diseases with a fully enrolled Phase 2b QUALITY clinical trial evaluating enobosarm to preserve muscle and augment fat loss for a higher quality weight loss in patients receiving WEGOVY®, a GLP-1 receptor agonist. We are expecting topline clinical results for this study in January 2025."

As a result of the sale of the FC2 business, including the transfer of its UK and Malaysian based subsidiaries, Veru's headcount will be reduced by approximately 90% from 210 to 22. Estimated proceeds to the Company after deducting a change of control premium due SWK Funding LLC pursuant to the Company's Residual Royalty Agreement, dated as of March 5, 2018 (the "Royalty Agreement"), together with other customary fees for transactions of this type, are approximately \$12.5 million subject to certain post-closing adjustment provisions in the purchase agreement. Upon payment of the change of control premium to SWK, the Royalty Agreement terminates in accordance with its terms. The liabilities associated with the Royalty Agreement, which totaled \$9.9 million as of September 30, 2024, will be extinguished. Raymond James acted as a financial advisor to Veru, and Reinhart Boerner Van Deuren s.c. served as legal counsel to the Company.

About the Enobosarm Phase 2b QUALITY clinical trial

The fully enrolled Phase 2b, multicenter, double-blind, placebo-controlled, randomized, dose-finding QUALITY clinical trial is evaluating the safety and efficacy of enobosarm 3mg, enobosarm 6mg, or placebo as a treatment to preserve muscle and augment fat loss in 168 patients with sarcopenic obesity or overweight elderly (>60 years of age) patients receiving semaglutide (Wegovy®). The primary endpoint is total lean body mass, and the key secondary endpoints are total body fat mass and physical function as measured by stair climb test at 16 weeks. Topline clinical results from the trial are expected in January of 2025.

After completing the efficacy dose-finding portion of the Phase 2b QUALITY clinical trial, it is expected that participants will then continue in blinded fashion into a Phase 2b extension clinical trial where all patients will stop receiving a GLP-1 RA, but will continue taking placebo, enobosarm 3mg, or enobosarm 6mg for an additional 12 weeks. The Phase 2b extension clinical trial will evaluate whether enobosarm can maintain muscle and prevent the fat and weight gain that occurs after discontinuing a GLP-1 RA. The topline results of the separate blinded Phase 2b extension clinical study are expected in calendar Q2 2025.

About Sarcopenic Obesity

According to the CDC, 41.5% of older adults have obesity in the United States and could benefit from a weight loss medication. Up to 34.4% of these obese patients over the age of 60 have sarcopenic obesity. This large subpopulation of sarcopenic obese patients is especially at risk for taking GLP-1 drugs for weight loss as they already have critically low amount of muscle due to age-related muscle loss. Further loss of muscle mass when taking a GLP-1 RA medication may lead to muscle weakness leading to poor balance, decreased gait speed, mobility disability, loss of independence, falls, bone fractures and increased mortality which is a condition like age-related frailty. Because of the magnitude and speed of muscle loss while on GLP-1 RA therapy for weight loss, GLP-1 RA drugs may accelerate the development of frailty in older obese or overweight elderly patients.

About Enobosarm

Enobosarm (aka ostarine, MK-2866, GTx-024, and VERU-024), a novel oral daily selective androgen receptor modulator (SARM), has been previously studied in 5 clinical studies involving 968 older normal men and postmenopausal women as well as older patients who have muscle wasting because of advanced cancer. Advanced cancer causes the loss of appetite where there is significant unintentional loss or wasting of both muscle and fat mass which is similar to what is observed with in patients taking GLP-1 RA drugs. We believe the totality of the clinical data from these previous five clinical trials demonstrates that enobosarm treatment leads to dose-dependent increases in muscle mass with improvements in physical function as well as significant dose-dependent reductions in fat mass. The patient data that were generated from these five enobosarm clinical trials in both elderly patients and in patients with a cancer induced appetite suppression provide strong clinical rationale for enobosarm. The expectation is that enobosarm in combination with a GLP-1 RA would potentially augment the fat reduction and total weight loss while preserving muscle mass.

Enobosarm has a large safety database, which includes 27 clinical trials involving 1581 men and women, some of which included patients dosed for up to 3 years. In this large safety database, enobosarm was generally well tolerated with no increases in gastrointestinal side effects. This is important as there are already significant and frequent gastrointestinal side effects with a GLP-1 RA treatment alone.

About Veru Inc.

Veru is a late clinical stage biopharmaceutical company focused on developing novel medicines for the treatment of cardiometabolic diseases, oncology, and ARDS. The Company's drug development program includes two late-stage novel small molecules, enobosarm and sabizabulin.

Enobosarm, a selective androgen receptor modulator (SARM), is being developed for two indications: (i) Phase 2b clinical QUALITY study of enobosarm as a treatment to augment fat loss and to prevent muscle loss in sarcopenic obese or overweight elderly patients receiving a GLP-1 RA who are at-risk for developing muscle atrophy and muscle weakness and (ii) subject to the availability of sufficient funding, Phase 3 ENABLAR-2 clinical trial of enobosarm and abemaciclib for the treatment of androgen receptor positive (AR+), estrogen receptor positive (ER+) and human epidermal growth factor receptor 2 negative (HER2-) metastatic breast cancer in the 2nd line setting.

Sabizabulin, a microtubule disruptor, is being developed as a Phase 3 clinical trial for the treatment of hospitalized patients with viral-induced ARDS. The Company does not intend to undertake further development of sabizabulin for the treatment of viral-induced ARDS until we obtain funding from government grants, pharmaceutical company partnerships, or other similar third-party external sources.

About Riva Ridge Capital Management LP

Riva Ridge, founded in 2003, is a private investment partnership focused on middle market companies going through business transitions and/or operational improvements.

Forward-Looking Statements

This press release contains "forward-looking statements" as that term is defined in the Private Securities Litigation Reform Act of 1995, including, without limitation, express or implied statements related to whether and when the Phase 2b trial of enobosarm discussed above will produce topline data or patients will progress into the extension study, the planned design, number of sites, timing, endpoints, patient population and patient size of such trial and whether such trial will successfully meet any of its endpoints, whether enobosarm will enhance weight loss or preserve muscle in, or meet any unmet need for, obesity patients and whether it will enhance weight loss, whether the Company will be successful in its transformation into a late stage biopharmaceutical company focused on obesity and oncology, and the actual amount of proceeds the Company may receive from the sale of the FC2 business due to the adjustment provisions in the purchase agreement. The words "anticipate," "believe," "could," "expect," "intend," "may," "opportunity," "plan," "predict," "potential," "estimate," "should," "will," "would" and similar expressions are intended to identify forward-looking statements, although not all forward-looking statements contain these identifying words. Any forward-looking statements in this press release are based upon current plans and strategies of the Company and reflect the Company's current assessment of the risks and uncertainties related to its business and are made as of the date of this press release. The Company assumes no obligation to update any forward-looking statements contained in this press release because of new information or future events, developments or circumstances. Such forward-looking statements are subject to known and unknown risks, uncertainties and assumptions, and if any such risks or uncertainties materialize or if any of the assumptions prove incorrect, our actual results could differ materially from those expressed or implied by such statements. Factors that may cause actual results to differ materially from those contemplated by such forward-looking statements include, but are not limited to: the development of the Company's product portfolio and the results of clinical studies possibly being unsuccessful or insufficient to meet applicable regulatory standards or warrant continued development; the ability to enroll sufficient numbers of subjects in clinical studies and the ability to enroll subjects in accordance with planned schedules; the ability to fund planned clinical development as well as other operations of the Company; the timing of any submission to the FDA or any other regulatory authority and any determinations made by the FDA or any other regulatory authority; any products of the Company, if approved, possibly not being commercially successful; the ability of the Company to obtain sufficient financing on acceptable terms when needed to fund development and operations; the Company's failure to timely file certain reports in February 2024 may impair its ability to raise capital under the Company's current effective shelf registration statement on Form S-3 or under a new registration statement; demand for, market acceptance of, and competition against any of the Company's products or product candidates; new or existing competitors with greater resources and capabilities and new competitive product approvals and/or introductions; changes in regulatory practices or policies or government-driven healthcare reform efforts, including pricing pressures and insurance coverage and reimbursement changes; the Company's ability to protect and enforce its intellectual property; costs and other effects of litigation, including product liability claims and securities litigation; the Company's ability to identify, successfully negotiate and complete suitable acquisitions or other strategic initiatives; the Company's ability to successfully integrate acquired businesses, technologies or products; and other risks detailed from time to time in the Company's press releases, shareholder communications and Securities and Exchange Commission filings, including the Company's Form 10-K for the year ended September 30, 2024, and subsequent quarterly reports on Form 10-Q. These documents are available on the "SEC Filings" section of our website at www.verupharma.com/investors.

* Wegovy® is a registered trademark of Novo Nordisk A/S

UNAUDITED PRO FORMA CONSOLIDATED FINANCIAL STATEMENTS

On December 30, 2024, Veru Inc. (the "Company") and The Female Health Company Limited, a wholly owned subsidiary of the Company ("FHC Limited" and, collectively with the Company, the "Sellers") entered into a Stock and Asset Purchase Agreement (the "Purchase Agreement") with Clear Future, Inc. (the "Purchaser"). Pursuant to, and subject to the terms and conditions of, the Purchase Agreement, the Purchaser purchased substantially all of the assets (the "FC2 Business Sale") related to the Company's FC2 female condom business ® (internal condom), including the stock of the Company's U.K. and Malaysian operating subsidiaries (the "FC2 business"). The Purchaser will only assume certain liabilities relating to the FC2 business that are specified in the Purchase Agreement. The transaction closed on December 30, 2024.

The following unaudited pro forma consolidated financial statements have been derived from the Company's historical consolidated financial statements as adjusted to give effect to the FC2 Business Sale. The unaudited pro forma consolidated statements of operations for the years ended September 30, 2024 and 2023 give effect to the FC2 Business Sale as if it had occurred on October 1, 2022 in that they reflect the reclassification of the FC2 business as discontinued operations for all periods presented. The unaudited pro forma consolidated balance sheet as of September 30, 2024 gives effect to the FC2 Business Sale as if it had occurred on September 30, 2024. After the date of the FC2 Business Sale, the historical financial results of the FC2 business will be reflected in our consolidated financial statements as discontinued operations under generally accepted accounting principles in the United States ("U.S. GAAP") for all periods.

The unaudited pro forma consolidated financial statements are for informational purposes only and do not purport to represent what the actual results of operations or financial position would have been had the FC2 Business Sale occurred on the dates assumed, nor are they indicative of future results of operations or financial positions. The pro forma adjustments are based on currently available information and certain estimates and assumptions, which are described in the accompanying notes and which management believes are reasonable. Actual adjustments may differ from the pro forma adjustments.

The unaudited pro forma condensed consolidated financial statements have been prepared in accordance with Regulation S-X Article 11, Pro Forma Financial Information.

VERU INC. UNAUDITED PRO FORMA CONSOLIDATED BALANCE SHEET AS OF SEPTEMBER 30, 2024

Assets Sturent asse		Historical		Discontinued Operations (a)		Transaction Accounting Adjustments		Pro Forma	
Cash and cash equivalents									
Accounts receivable, net									
Accounts receivable, net 1,414,179 1	Cash and cash equivalents	\$	24,916,285	\$	_	\$	(1,462,750) (c)	\$	37,231,925
Inventories, net	Accounts receivable net		3 960 305		(3.960.305)		(4,221,010) (u)		
Prepaid essearch and development costs			, ,				_		_
Prepaid expenses and other current assets			, ,		(4,144,177)		_		419 371
Total current assets					(654 527)		_		
Plant and equipment, net						-	12 315 640	_	
Operating lease right-of-use asset 3,561,773 (311,150) — 3,250,623 Defered income taxes 12,340,237 (12,340,237) — — 6,878,932 Other assets 1,129,952 (140,356) — 989,596 Total assets 5,60,418,772 \$(22,354,036) \$12,315,640 \$50,380,376 Libilities and Stockholders' Equity Current liabilities Accounts payable \$3,036,841 \$(777,173) \$ — \$ 2,259,668 Accrued compensation 4,679,696 (185,418) — \$ 2,259,668 Accrued expenses and other current liabilities 2,060,097 (1,309,412) — 669,685 Accrued expenses and other current portion 1,055,497 (328,527) — 76,6970 Operating lease liability, short-term portion 1,055,497 (328,527) — 76,970 Total current liabilities 2,91,380 (1,012,837) (1,025,837) 8,160,601 Residual royalty agreement, long-term portion 8,850,792 — (8,850,792) 1,295,309 Operating lease liability, long-term portion 2,9			, -,		() / /		-		, ,
Deferred income taxes							_		/
Goodwill 6,878,932 — — 6,878,932 Other assets 1,129,952 (140,356) — 989,506 Total assets 5 60,418,772 \$ (22,354,036) \$ 12,315,640 \$ 50,380,376 Liabilities and Stockholders' Equity Current liabilities Accounts payable \$ 3,036,841 \$ (777,173) \$ — \$ 2,259,668 Accound expenses and other current liabilities 2,060,097 (1,390,412) — 669,885 Acsidual royalty agreement, short-term portion 1,025,837 (328,527) — 736,970 Operating lease liability, short-term portion 1,065,497 (328,527) — 736,970 Total current liabilities 1,1867,968 (5,881,530) (1,025,837) 8,160,601 Residual royalty agreement, long-term portion 8,850,792 — (8,850,792) (d) — Operating lease liability, long-term portion 2,921,380 (16,071) — 2,905,309 Other liabilities 4,461,920 — — 4,461,920			, ,		(/ /		_		
Other assets 1,129,952 (140,356) — 989,596 Total assets 5,0418,772 (22,354,036) 12,315,640 50,380,376 Liabilities Current liabilities Accounts payable \$ 2,259,668 Accrued compensation 4,679,696 (185,418) — 4,494,278 Accrued expenses and other current liabilities 2,060,097 (1,394,412) — 669,685 Residual royalty agreement, short-term portion 1,065,497 (328,577) — 736,970 Operating lease liability, short-term portion 8,850,792 — (8,850,792) — 736,970 Total current liabilities 4,461,920 — (8,850,792) — 2,905,309 Operating lease liability, long-term portion 2,921,380 (16,071) — 2,905,309 Operating lease liabilities 4,461,920 — — 2,905,309 Other liabilities — 2,820,309 — — — — — — — — — <t< td=""><td></td><td></td><td>, ,</td><td></td><td>— (==,= :+,==+,) (g)</td><td></td><td>_</td><td></td><td>6.878.932</td></t<>			, ,		— (==,= :+,==+,) (g)		_		6.878.932
Total assets	Other assets				(140.356)		_		
Liabilities and Stockholders' Equity Samuel Stockholders' Equi		S		\$		S	12.315.640	S	
Current liabilities: S 3,036,841 \$ (777,173) \$ — \$ 2,259,668 Accrued compensation 4,679,696 (185,418) — 4,494,278 Accrued expenses and other current liabilities 2,060,097 (1,390,412) — 669,685 Residual royalty agreement, short-term portion 1,025,837 — (1,025,837) (d) — Operating lease liability, short-term portion 1,065,497 328,527) — 736,970 Total current liabilities 11,867,968 (2,681,530) (1,025,837) 8,160,601 Residual royalty agreement, long-term portion 8,850,792 — (8,850,792) (d) — Operating lease liability, long-term portion 2,921,380 (16,071) — 2,905,309 Other liabilities 28,102,060 (2,697,601) (9,876,629) 15,527,830 Stockholders' equity: Preferred stock, no shares issued and outstanding at September 30, 2020 and 2019, respectively — — — — — Common stock, par value \$0.01 per share; 308,000,000 shares authorized, 148,567,624 shares issued and 146,383,920 shar	Total assets	Ť	**,***,**	Ť	(==,===,,===)	_	,	Ť	,,-
Current liabilities: S 3,036,841 \$ (777,173) \$ — \$ 2,259,668 Accrued compensation 4,679,696 (185,418) — 4,494,278 Accrued expenses and other current liabilities 2,060,097 (1,390,412) — 669,685 Residual royalty agreement, short-term portion 1,025,837 — (1,025,837) (d) — Operating lease liability, short-term portion 1,065,497 328,527) — 736,970 Total current liabilities 11,867,968 (2,681,530) (1,025,837) 8,160,601 Residual royalty agreement, long-term portion 8,850,792 — (8,850,792) (d) — Operating lease liability, long-term portion 2,921,380 (16,071) — 2,905,309 Other liabilities 28,102,060 (2,697,601) (9,876,629) 15,527,830 Stockholders' equity: Preferred stock, no shares issued and outstanding at September 30, 2020 and 2019, respectively — — — — — Common stock, par value \$0.01 per share; 308,000,000 shares authorized, 148,567,624 shares issued and 146,383,920 shar	Liabilities and Stockholders' Fauity								
Accounts payable \$ 3,036,841 \$ (777,173) \$ - \$ 2,259,668									
Accrued compensation		s	3 036 841	\$	(777 173)	\$	_	\$	2 259 668
Accrued expenses and other current liabilities Residual royalty agreement, short-term portion 1,025,837 1,0				Ψ.		Ψ	_	Ψ	, ,
Residual royalty agreement, short-term portion 1,025,837							_		
Operating lease liability, short-term portion 1,065,497 (328,527) — 736,970 Total current liabilities 11,867,968 (2,681,530) (1,025,837) 8,160,601 Residual royalty agreement, long-term portion 8,850,792 — (8,850,792) (d) — Operating lease liability, long-term portion 2,921,380 (16,071) — 2,905,309 Other liabilities 4,461,920 — — 4,461,920 Total liabilities 28,102,060 (2,697,601) (9,876,629) 15,527,830 Stockholders' equity: Preferred stock, no shares issued and outstanding at September 30, 2020 and 2019, respectively — — — — — — — — — — — — — — — — — — —					(1,5)0,112)		(1.025.837) (d)		_
Total current liabilities					(328,527)		_		736,970
Residual royalty agreement, long-term portion 8,850,792 — (8,850,792) (d) — Operating lease liability, long-term portion 2,921,380 (16,071) — 2,905,309 Other liabilities 4,461,920 — — 4,461,920 Total liabilities 28,102,060 (2,697,601) (9,876,629) 15,527,830 Stockholders' equity: Preferred stock, no shares issued and outstanding at September 30, 2020 and 2019, respectively —							(1.025.837)		
Operating lease liability, long-term portion 2,921,380 (16,071) — 2,905,309 Other liabilities 4,461,920 — — 4,461,920 Total liabilities 28,102,060 (2,697,601) (9,876,629) 15,527,830 Stockholders' equity: Preferred stock, no shares issued and outstanding at September 30, 2020 and 2019, respectively — <td></td> <td></td> <td></td> <td></td> <td>(_,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,</td> <td></td> <td></td> <td></td> <td></td>					(_,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,				
Other liabilities 4,461,920 — — 4,461,920 Total liabilities 28,102,060 (2,697,601) (9,876,629) 15,527,830 Stockholders' equity: Preferred stock, no shares issued and outstanding at September 30, 2020 and 2019, respectively —			, ,		(16.071)		(s,st s,,,,=) (s)		2.905.309
Total liabilities 28,102,060 (2,697,601) (9,876,629) 15,527,830 Stockholders' equity: Preferred stock, no shares issued and outstanding at September 30, 2020 and 2019, respectively — <td< td=""><td></td><td></td><td></td><td></td><td>_</td><td></td><td>_</td><td></td><td></td></td<>					_		_		
Stockholders' equity: Preferred stock, no shares issued and outstanding at September 30, 2020 and 2019, respectively — — — — — Common stock, par value \$0.01 per share; 308,000,000 shares authorized, 148,567,624 shares issued and 146,383,920 shares outstanding 1,485,676 — — 1,485,676 Additional paid-in-capital 333,788,795 — — 333,788,795 Accumulated other comprehensive loss (581,519) 581,519 — — Accumulated deficit (294,569,635) (20,237,954) 22,192,269 (e) (292,615,320) Treasury stock, 2,183,704 shares, at cost (7,806,605) — — — (7,806,605) Total stockholders' equity 32,316,712 (19,656,435) 22,192,269 34,852,546					(2.697.601)		(9.876.629)		
Preferred stock, no shares issued and outstanding at September 30, 2020 and 2019, respectively —			,-,-,-,-		(=,00,1,000)		(,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		10,021,000
Preferred stock, no shares issued and outstanding at September 30, 2020 and 2019, respectively —	Stockholders' equity:								
2019, respectively — — — — Common stock, par value \$0.01 per share; 308,000,000 shares authorized, 148,567,624 shares issued and 146,383,920 shares outstanding 1,485,676 — — 1,485,676 Additional paid-in-capital 333,788,795 — — 333,788,795 Accumulated other comprehensive loss (581,519) 581,519 — — Accumulated deficit (294,569,635) (20,237,954) 22,192,269 (e) (292,615,320) Treasury stock, 2,183,704 shares, at cost (7,806,605) — — (7,806,605) Total stockholders' equity 32,316,712 (19,656,435) 22,192,269 34,852,546	Preferred stock, no shares issued and outstanding at September 30, 2020 and								
Common stock, par value \$0.01 per share; 308,000,000 shares authorized, 148,567,624 shares issued and 146,383,920 shares outstanding 1,485,676 — — 1,485,676 Additional paid-in-capital 333,788,795 — — 333,788,795 Accumulated other comprehensive loss (581,519) 581,519 — — Accumulated deficit (294,569,635) (20,237,954) 22,192,269 (e) (292,615,320) Treasury stock, 2,183,704 shares, at cost (7,806,605) — — (7,806,605) Total stockholders' equity 32,316,712 (19,656,435) 22,192,269 34,852,546			_		_		_		_
Additional paid-in-capital 333,788,795 — 333,788,795 Accumulated other comprehensive loss (581,519) 581,519 — — Accumulated deficit (294,569,635) (20,237,954) 22,192,269 (e) (292,615,320) Treasury stock, 2,183,704 shares, at cost (7,806,605) — — (7,806,605) Total stockholders' equity 32,316,712 (19,656,435) 22,192,269 34,852,546									
Accumulated other comprehensive loss (581,519) 581,519 — — Accumulated deficit (294,569,635) (20,237,954) 22,192,269 (e) (292,615,320) Treasury stock, 2,183,704 shares, at cost (7,806,605) — — — (7,806,605) Total stockholders' equity 32,316,712 (19,656,435) 22,192,269 34,852,546	148,567,624 shares issued and 146,383,920 shares outstanding		1,485,676		_		_		1,485,676
Accumulated other comprehensive loss (581,519) 581,519 — — Accumulated deficit (294,569,635) (20,237,954) 22,192,269 (e) (292,615,320) Treasury stock, 2,183,704 shares, at cost (7,806,605) — — — (7,806,605) Total stockholders' equity 32,316,712 (19,656,435) 22,192,269 34,852,546	Additional paid-in-capital		333,788,795		_		_		333,788,795
Treasury stock, 2,183,704 shares, at cost (7,806,605) — — (7,806,605) Total stockholders' equity 32,316,712 (19,656,435) 22,192,269 34,852,546			(581,519)		581,519		_		_
Total stockholders' equity 32,316,712 (19,656,435) 22,192,269 34,852,546	Accumulated deficit		(294,569,635)		(20,237,954)		22,192,269 (e)		(292,615,320)
Total stockholders' equity 32,316,712 (19,656,435) 22,192,269 34,852,546	Treasury stock, 2,183,704 shares, at cost		(7,806,605)						(7,806,605)
0 (0.440,000) 0 (0.040,000)	• • • • • • • • • • • • • • • • • • • •				(19,656,435)		22,192,269		34,852,546
	Total liabilities and stockholders' equity	\$	60,418,772	\$	(22,354,036)	\$	12,315,640	\$	50,380,376

See notes to unaudited pro forma consolidated financial statements. \\

VERU INC. UNAUDITED PRO FORMA CONSOLIDATED STATEMENT OF OPERATIONS FOR THE YEAR ENDED SEPTEMBER 30, 2024

	 Historical	_	Discontinued perations (a)	Acc	nsaction counting ustments	 Pro Forma
Net revenues	\$ 16,886,419	\$	(16,886,419)	\$	_	\$ _
Cost of sales	 11,032,442		(11,032,442)			 _
Gross profit	5,853,977		(5,853,977)		_	_
Operating expenses:						
Research and development	12,808,185		(26,018)		_	12,782,167
Selling, general and administrative	31,184,097		(6,574,421)		_	24,609,676
Total operating expenses	43,992,282		(6,600,439)		_	37,391,843
Gain on sale of ENTADFI® assets	 1,222,908	_				 1,222,908
Operating loss	(36,915,397)		746,462		_	(36,168,935)
Non-operating (expenses) income:						
Interest expense	(607,470)		_		607,470 (f)	_
Change in fair value of derivative liabilities	(239,000)		_		239,000 (f)	_
Change in fair value of equity securities	(176,077)		_		_	(176,077)
Other income, net	 861,619		223,580			1,085,199
Total non-operating expenses	 (160,928)		223,580		846,470	 909,122
Loss before income taxes	(37,076,325)		970,042		846,470	(35,259,813)
Income tax expense	 725,101		(725,101) (h)			 _
Net income (loss)	\$ (37,801,426)	\$	1,695,143	\$	846,470	\$ (35,259,813)
Net loss per basic and diluted common share outstanding	\$ (0.28)					\$ (0.26)
Basic and diluted weighted average common shares outstanding	134,875,016					134,875,016

See notes to unaudited pro forma consolidated financial statements.

VERU INC. UNAUDITED PRO FORMA CONSOLIDATED STATEMENT OF OPERATIONS FOR THE YEAR ENDED SEPTEMBER 30, 2023

	_	Historical	_	viscontinued perations (a)	Transaction Accounting Adjustments		<u>.</u>	Pro Forma	
Net revenues	\$	16,296,958	\$	(16,283,945)	\$	_	\$	13,013	
Cost of sales		8,731,030		(8,689,189)				41,841	
Gross profit		7,565,928		(7,594,756)		_		(28,828)	
Operating expenses:									
Research and development		51,202,219		(90,834)		_		51,111,385	
Selling, general and administrative		48,057,834		(10,911,258)		_		37,146,576	
Provision for credit losses		3,911,714		(3,911,714)		_		_	
Impairment of intangible assets		3,900,000						3,900,000	
Total operating expenses		107,071,767		(14,913,806)		_		92,157,961	
Gain on sale of ENTADFI® assets		5,723,623						5,723,623	
Operating loss		(93,782,216)		7,319,050		_		(86,463,166)	
Non-operating (expenses) income:									
Interest expense		(2,427,041)		_		2,427,041 (f)		_	
Change in fair value of derivative liabilities		2,963,000		_		(2,963,000) (f)		_	
Other income, net		573,771		58,558				632,329	
Total non-operating expenses		1,109,730	_	58,558	_	(535,959)		632,329	
Loss before income taxes		(92,672,486)		7,377,608		(535,959)		(85,830,837)	
Income tax expense (benefit)	_	480,206		(552,258) (h)				(72,052)	
Net income (loss)	\$	(93,152,692)	\$	7,929,866	\$	(535,959)	\$	(85,758,785)	
Net loss per basic and diluted common share outstanding	\$	(1.10)					\$	(1.01)	
Basic and diluted weighted average common shares outstanding		84,973,382						84,973,382	
See notes to unaudited pro forma consolidated financial statements.									

VERU INC. NOTES TO THE UNAUDITED PRO FORMA CONSOLIDATED FINANCIAL STATEMENTS

The unaudited pro forma consolidated financial statements include the following adjustments:

- (a) Reflects the discontinued operations, including associated assets, liabilities, equity, and results of operations attributable to the FC2 business, which were included in the Company's historical financial statements. In accordance with ASC 205-20, Presentation of Financial Statements Discontinued Operations, the amounts exclude intercompany transactions, which are eliminated on consolidation, and general corporate overhead costs allocated to the FC2 business that do not meet the requirements to be presented in discontinued operations, which include the Company's corporate support functions.
- (b) Reflects the estimated cash to be received in connection with the FC2 Business Sale. This amount is subject to adjustment as set forth in the Purchase Agreement, including a customary working capital adjustment based on the amount by which certain working capital items at closing are greater or less than a target set forth in the Purchase Agreement.
- (c) Reflects cash to be paid for estimated non-recurring costs to complete the FC2 Business Sale. These costs primarily relate to investment banker fees, legal fees, and other transaction costs.
- (d) The Company's Residual Royalty Agreement, dated as of March 5, 2018 (the "Royalty Agreement"), with SWK Funding LLC ("SWK") requires upon a sale of the FC2 business a payment (the "Change of Control Payment") equal to the product of 5% of the product revenue from net sales of FC2 for the most recently completed 12-month period multiplied by five, and upon such payment the Royalty Agreement terminates in accordance with its terms. This adjustment reflects the Change of Control Payment to be paid to SWK upon the FC2 Business Sale and the extinguishment of the associated liabilities.
- (e) Reflects the effect on total shareholders' equity of the adjustments described in notes (b) through (d) above.
- (f) The Royalty Agreement terminates upon the Company making the Change of Control Payment in connection with the FC2 Business Sale. The pro forma adjustments reflect the reduction of associated interest expense and the impact on the change in fair value of the derivative liabilities, which will be eliminated upon termination of the Royalty Agreement.
- (g) Represents the removal of deferred tax assets related to the Company's operating U.K. subsidiary, The Female Health Company (UK) plc, which does not have a valuation allowance due to projections of future taxable income.
- (h) Represents the removal of income tax expense associated with the Company's U.K. and Malaysia operating subsidiaries. The Company has U.S. federal and state net operating loss and tax credit carryforwards that it believes would offset any potential tax liability associated with this transaction.